# INVESTMENT FUNDS IN CANADA AND CONSUMER PROTECTION

# STRATEGIES FOR THE MILLENNIUM

A Review by Glorianne Stromberg

Prepared for Office of Consumer Affairs, Industry Canada

October 1998

This publication is also available electronically on the World Wide Web at the following address: http://strategis.ic.gc.ca/oca

This publication is also available in alternative formats on request. Contact Information Distribution Centre at the numbers listed below.

For additional copies of this publication, please contact:

Information Distribution Centre Communications Branch Industry Canada Room 205D, West Tower 235 Queen Street Ottawa ON K1A 0H5

Tel.: (613) 947-7466 Fax: (613) 954-6436

Opinions and statements in this publication do not necessarily reflect the policy of Industry Canada or of the Government of Canada.

©1998 Glorianne Stromberg. All rights reserved. Cat. No. C2-396/1998E ISBN 0-662-27425-3 52487E

Aussi disponible en français sous le titre Les fonds d'investissement et la protection du consommateur au Canada% Stratégies pour le millénaire.



Opinions and statements in this publication do not necessarily reflect the policy of Industry Canada or of the Government of Canada.
©1998 Glorianne Stromberg. All rights reserved.

# **INVESTMENT FUNDS IN CANADA AND CONSUMER PROTECTION**

# STRATEGIES FOR THE MILLENNIUM

# **TABLE OF CONTENTS**

# **EXECUTIVE SUMMARY**

# **GLOSSARY**

1. TERMS OF REFERENCE	
1.1. The Task	1
1.2. The Process	
1.3. The Position of the Consumer/Investor Prior to 1995	
2. THE 1995 REPORT	2
2.1. A Review of the Investment Fund Industry in Canada	2
2.2. Post 1995 Report Activity	Δ
2.3. Areas of Concern Identified in the 1995 Report	
2.4. The Perspective Underlying the 1995 Report Recommendations	7
2.5. Strategic Recommendations Made in the 1995 Report	
2.6. Relationship of the Strategic Recommendations to Each Other	10
3. CHANGES SINCE THE 1995 REPORT THAT SHOULD BENEFIT THE CONSUM	ER/INVESTOR11
0.4 D .44005 D .401	4.4
3.1. Post 1995 Report Changes in the Investment Fund Industry	11
3.2. Heightened Awareness of Issues	
3.4. Sales Practices Rule	
3.5. Mutual Fund Prospectus Disclosure	
3.6. Code of Ethics on Personal Investing	
3.7. Personal Rate of Return Calculation Standard	
3.8. Standard for Categorizing Mutual Funds	
3.9. Mutual Fund Dealers Association of Canada	14
3.10. Mutual Reliance Proposals	
3.11. Cooperation between Insurance and Securities Regulators	16
3.12. Coordinating Mechanism for Regulation	
3.13. CICA Audit of Investment Funds Research Project	17
4. CHANGES RESULTING FROM GOVERNMENTAL POLICY INITIATIVES	47
4. CHANGES RESULTING FROM GOVERNMENTAL POLICY INITIATIVES	
4.1. Federal Government Policy Initiatives	17
4.2. Provincial Government Policy Initiatives	17
4.3. Self-Funding	
5. CHANGES IN THE STRATEGIC ENVIRONMENT	18
5.1 Strategic Forces	18

5.2. The Changing Face of the Investor in the New Financial Landscape	
5.3. Industry Consolidation	20
5.4. The Blurring of Product and Function and its Consequences	21
6. POST 1995 REPORT EFFECT OF CHANGES ON CONSUMER/INVESTORS	24
6.1. Industry Perspectives on Whether Consumer/Investors Are Better Off Today	
6.2. My Evaluation	26
7. IMPEDIMENTS TO CHANGE	26
7.1. Industry Perspectives on Impediments to Change	26
7.2. Anomalies Driving the Investment Fund Industry	
8. INVESTMENT FUNDS AS CONSUMER GOODS OR SERVICES	28
8.1. Viewing Investment Funds as Consumer Goods or Services	28
8.2. Consumer/Investor Goals and Objectives	31
9. MAKING A DIFFERENCE FOR CONSUMER/INVESTORS	33
9.1. Recognition of Fundamental Needs	33
9.2. Reducing the Knowledge Gap	33
9.3. Recognition of the Consumer/Investor's Integrated Needs	34
10. REGULATORY FRAMEWORK	35
10.1. Regulatory Coordination	35
10.2. Additional Initiatives	37
10.3. Leadership and Vision	37
10.4. Provision of Sufficient Resources	37
11. IMPLICATIONS OF THE CURRENT LEVEL OF KNOWLEDGE AND AWARENESS	38
11.1. Consumer/Investors Current State of Knowledge and Awareness	
11.2. Implications for the Regulatory and Supervisory Framework	40
11.3. Adequacy of the Current Regulatory and Supervisory Framework	40
12. SUGGESTED APPROACHES TO REGULATORY AND SUPERVISORY STRATEGIES	41
12.1. Suggested Approaches	41
12.2. Regulatory Structure	
12.3. International Regulatory Cooperation	44
13. SYSTEMIC CHANGES IN THE REGULATORY SYSTEM	45
13.1. Systemic Changes	45
13.2. Consideration of an Issuer-Based Integrated Disclosure Model	
13.3. Timely Disclosure - Level Playing Field	47
14. KNOWLEDGE AND AWARENESS OF CONSUMER/INVESTORS	48
14.1. What is Needed to Make a Difference	
14.2. Filling the Information Gap - Youth	
14.3. Filling the Information Gap - Adults	
I/I/I FIGURATE AT 3 SUCCESSIII AAUIT I ABTAINA PRAAFAM	h /

14.5. Specific Learning Tools for Consumer/Investors	57
14.6. Consumer/Investor Education Strategy	58
15. FINANCIAL PLANNING AND ADVICE	58
15.1. Education and Proficiency Standards for Industry Participants	58
15.2. Beyond Product Sales	
15.3. Growing Awareness of Need for Change	60
15.4. The Reasonable Expectations of Consumer/Investors	
15.6. Confusion of Issues	
15.7. Additional Confusion	
15.8. Establishing a Financial Planning Standard	
15.9. Observations Relating to Education and Proficiency Programs for Industry Participants	
15.10. Need for a Non-Partisan Standards Council	
15.11. The FPSCC and the CFP	
16. THE RIGHT TO DEAL WITH THE PUBLIC	60
16.1. Registration and Licensing Requirements	69
16.2. Simplification of Regulatory Requirements for Registration	
16.3. Registration Framework	
16.5. Problems with the Current Registration System	
16.6. Specific Concerns	
16.7. Industry Standards for Intermediaries	
16.8. Compliance with Sales Practices Rule	
16.9. Penalty Provisions	
16.10. Other Related Issues	
17. MEETING CONSUMER/INVESTOR INFORMATION NEEDS	87
17.1. Disclosure as a Cornerstone Strategy	87
17.2. Effectiveness of Disclosure as a Consumer Protection Strategy	87
17.3. The Elements and Dimensions of Disclosure	
17.4. Problems with the Prospectus Disclosure System	
17.5. Premises Underlying 1995 Report Recommendations for Changes in the Disclosure System	91
17.6. The CSA Proposals for Mutual Fund Prospectus Disclosure	
17.7. Enhancement of the CSA Proposals for Mutual Fund Prospectus Disclosure	
17.8. Confirmation Forms	97
17.9. Consumer/Investor Checklist	
17.10. Account Statements	
17.11. Consolidated Reporting and Account Statements	
17.12. Know-Your-Client/Suitability Procedures	104
18. IMPROVING DISCLOSURE	106
18.1. Better Disclosure of Fees and Charges	
18.2. Timely Disclosure of Portfolio Holdings	108
18.3. Disclosure of Ownership Positions in Portfolio Holdings	
18.4. Voting Rights	
18.5. Annual and Interim Reports	
18.6. Expanding Public Access to the Internet	
18.7. Expanding Use of the Internet to Disseminate Information	
10.0. SLDAN	11∠

19. FUND GOVER	NANCE	112
10.1 Fund Go	overnance	112
	ablishment and Structure of Investment Funds	
	s, Directors and Officers of Investment Funds	
	cy of Internal Controls, Systems and Procedures	
	on of Costs and Expenses	
19.5. Allocatio	n or Costs and Expenses	115
20. INVESTMENT	ALTERNATIVES	116
20.1. Investme	ent Alternatives	116
20.2. Substan	tive Concerns	117
20.3. Recomn	nendations to Address Concerns	121
21. ONGOING RE	LATIONSHIPS BETWEEN INTERMEDIARIES AND CONSUMER/INVESTORS	123
21.1. Switchin	g and Churning	123
	nendations to Address Switching and Churning	
	tion of Securities and Powers of Attorney	
	nendations - Registration of Securities and Powers of Attorney	
	6	
	nendations Respecting Leverage	
	· of Accounts	
	nendations Relating to the Transfer of Accounts	
22. TAX MATTERS	S	141
22.1. Tax Asp	ects of Investment Funds	141
23. PROPERTY LA	AWS	146
23.1. Moderni	zing and Harmonizing Property Laws	146
24. CONSUMER R	REDRESS MECHANISMS	149
24.1 Consum	er Redress Mechanisms	1/0
	f Consumer Redress Mechanisms	
	es in Consumer Redress Mechanisms	
24.4. Recomm 24.5. Ombuds	nendationssmen	
05 0THED INVES	THENT FUND MATTERS	450
25. OTHER INVES	TMENT FUND MATTERS	152
25.1. Other In	vestment Fund Matters	152
26. CONCLUSION		157
APPENDIX A	IFIC Statistics for the 10 Year Period Ended August 31, 1998	
APPENDIX B	Supplementary Information Relating to Section 14 - Knowledge and Awareness of Consumer/Investors	

**APPENDIX C** 

Supplementary Information Relating to Subsection 15.9 - Observations Relating to Education and Proficiency Programs for Industry

**Participants** 

**APPENDIX D** 

Supplementary Information Relating to Subsection 17.12 - Industry Observations Relating to Know-Your-Client/Suitability Procedures

### INVESTMENT FUNDS IN CANADA AND CONSUMER PROTECTION

# STRATEGIES FOR THE MILLENNIUM

### **EXECUTIVE SUMMARY**

# Why the Review Was Undertaken

This Review was undertaken at the request of Industry Canada's Office of Consumer Affairs. It was prompted by the realization that Canadian consumers are increasingly turning to investment funds as a source of their retirement income.

The investment fund industry has been good for Canada and for Canadians. The increased popularity of investment funds, combined with strong market performance, has fuelled the growth of the investment fund industry. Most of this growth has occurred within the last five years or so. There are many new players, many different types of investment funds, and many new investors in investment funds.

However, what was once a simple means of accessing professional investment advisory services by pooling money with other like-minded investors has become very complex. Three main factors contribute to this complexity. They are:

- the different distribution structures and the costs related thereto:
- the lack of a common regulatory regime and of common standards respecting money management; and
- the structuring of investment funds and their alternatives so as to:
  - (i) avoid the consequences of certain tax laws, securities and other laws, or
  - (ii) take advantage of them.

The Office of Consumer Affairs therefore decided that it was appropriate to consider whether there are reasonable measures in place to protect consumer/investors given the increased tendency by the industry and consumers alike to regard investment funds as "consumer goods."

### What I Was Asked to Do

### I was asked to:

- assess the consumer's position prior to the 1995 Report, the changes that have been made since then that benefited consumers and what remains to be done;
- identify impediments to changes that would benefit consumers;
- recommend what facilitating role governments could play to increase consumer sovereignty and control over their assets;

- identify mechanisms that would better enable consumers to identify their goals, how to achieve them and how to assess risk; and
- review consumer redress and compensation mechanisms with a view to identifying how consumers receive greater or less protection depending on where the purchase is made.

# Structure of the Review

The Review is structured to deal with these subjects as follows:

Section 1	This Section sets out the terms of reference and describes the review process.
Section 2	This Section outlines the purpose of the 1995 Report, the areas of concern it identified, the key recommendations that were made and the perspectives underlying the recommendations.
Section 3	This Section refers to changes that have occurred since the 1995 Report that should benefit the consumer/investor.
Section 4	This Section outlines changes that have resulted from governmental policy initiatives.
Section 5	This Section outlines changes in the strategic environment, with particular emphasis on the retailization of the marketplace, industry consolidation, and the consequences of the blurring and fusion of product, function and advice.
Section 6	This Section outlines various perspectives on whether the changes described in Sections 3, 4 and 5 have benefited consumer/investors.
Section 7	This Section outlines various perspectives on the impediments to change.
Section 8	This Section discusses the validity of the perspective of viewing investment funds as consumer goods or services. It also discusses the goals and objectives of consumer/investors and their expectations.
Section 9	This Section identifies factors that will make a difference for consumer/investors. These include measures to reduce the knowledge gap and to create a regulatory and supervisory structure that recognizes that consumer/investors have integrated financial needs.
Section 10	This Section outlines the need for regulatory coordination to reflect the changing marketplace and to meet the needs of all participants in the marketplace. The Section includes proposals for simplifying, streamlining and coordinating processes and for making sure that they are complementary.
Section 11	This Section assesses the current level of knowledge and awareness of consumer/investors and the implications for the regulatory and supervisory framework.
Section 12	This Section recommends new approaches to regulatory and supervisory
L	

	strategies to reflect:
	Strategies to reflect.
	the need to meet the integrated needs of the consumer/investor;
	<ul> <li>the expanded business activities and the diversified geographical operations of today's participants in the financial services sector;</li> </ul>
	the need to de-fragment the regulatory structure;
	the need for a common regime for money management;
	the need to re-evaluate the effectiveness of the self-regulatory component of the regulatory regime; and
	the need for international regulatory cooperation.
	It recommends changes to the regulatory structure to implement these approaches, including the establishment, pursuant to intergovernmental agreement, of an integrated regulatory body (the "Regulator") to reflect the fusion of product, function and advice which has occurred since the deregulation of the financial services industry that began in 1987.
Section 13	This Section recommends systemic changes to securities legislation including:
	replacing the "closed system" of securities regulation with a simple issuer-based, integrated disclosure model; and
	<ul> <li>leveling the playing field with respect to the dissemination of information.</li> </ul>
Section 14	This Section outlines the need for increasing the knowledge and awareness of consumer/investors. It recommends ways to do this through youth and adult learning programs aimed at filling information gaps and at enabling people to use information to make effective decisions to meet their personal goals and objectives.
Section 15	This Section outlines the need for a new approach to education and proficiency requirements for participants in the financial services industry to reflect the industry shift to advice-giving (with transactions to implement the advice being secondary).
	It recommends:
	a process for establishing a financial planning and investment advisory education and proficiency standard; and
	creating an independent, Non-Partisan Standards Council to identify this education and proficiency standard and to monitor and maintain the same.
	These recommendations are complementary to those in Section 12 and reflect the fusion of product, function and advice.
Section 16	This Section recommends a simplified approach to registration and licensing requirements for participants in the financial services industry. It recommends

replacing all registration and licensing requirements with: a single requirement for registration with the Regulator (referred to in Sections 10 and 12): and a *single* membership in the Single SRO. The Single SRO (which would operate on a national basis) is described in subsection 16.3. It would be the single self-regulatory organization that would be recognized by the Regulator for the purpose of the required membership in a self-regulatory organization. This is a companion recommendation to the recommendations in Section 12 for the establishment (pursuant to intergovernmental agreement) of an integrated regulatory body (the "Regulator") and to the recommendations in Section 15 for an integrated education and proficiency standard to reflect the fusion of product, function and advice. Section 17 This Section discusses: what is needed for an effective disclosure system that will meet the information needs of consumer/investors; • the need to look beyond the prospectus to meet such needs; and know-your-client and of the suitability requirements. confirmation/point-of-sale statements, account statements and annual and interim reports in meeting these needs. It contains recommendations for enhancing the usefulness of these requirements and documents to meet the needs of consumer/investors. Section 18 This Section makes specific recommendations for changes aimed at improving disclosure including recommendations related to: the disclosure of fees and charges and their impact on total return over a period of years; timely disclosure of portfolio holdings and of ownership and control positions; timely disclosure of ownership and control positions in the securities of issuers whose securities are included in the investment portfolio of the managed account; the exercise of voting rights attaching to securities of issuers whose securities are included in the investment portfolio of the managed account: timely annual and interim reporting of information relating to the operations of an investment fund; using the Internet to bring about effective disclosure; and enhancing the usefulness of SEDAR (the CSA's System for Electronic

	Document Analysis Retrieval).
Section 19	This Section discusses:
	the critical role that fund governance plays in consumer protection;
	the need for a legislative structure in Canada for "business trusts" comparable to the constating legislative structure that exists for corporations; and
	the need for prudential oversight mechanisms and for establishing prudent operating procedure standards.
	It confirms the recommendations made in the 1995 Report relating to these matters.
Section 20	This Section discusses investment fund alternatives. It discusses the substantive differences that give rise to substantive concerns that relate primarily to:
	the suitability of the particular investment alternative to meet the specific needs of individual consumer/investors; and
	the level of prudential and regulatory supervision and oversight over the structuring of these products and the offering of the same to consumer/investors; and
	the inability of consumer/investors to appreciate the differences in suitability and in prudential and regulatory supervision and oversight.
	The Section notes that the substantive concerns would be substantially addressed by the implementation of the proposals made in Sections 10 - 18 for:
	an integrated regulatory structure;
	a common regime for money management;
	increasing the knowledge and awareness levels of consumer/ investors and their advisers;
	effective disclosure;
	and by the additional proposals outlined in this Section for:
	prudential enhancements; and
	<ul> <li>the removal of certain restrictions and preferences that add costs, distort investment decisions and provide some sectors of the financial services industry with a competitive advantage over other sectors of the financial services industry.</li> </ul>

Section 21	This Section discusses some of the areas relating to the advice-giving/distribution activities of financial service providers that cause particular problems for consumer/investors. It contains recommendations to address these problems.
Section 22	This Section discusses some tax aspects relating to investment funds that are not well understood by consumer/investors and includes recommendations relating to these matters.
Section 23	This Section discusses the need for coordinated, cooperative and concerted action by all levels of government to simplify, modernize and harmonize basic laws relating to the ownership of property so that people's basic property rights are not dependent on where they live in Canada. This action would greatly facilitate the ability of consumer/investors to meet their goals and objectives.  The recommendations deal specifically with the laws relating to the eligibility of investment funds as investments for trustees, the designation of beneficiaries, probate requirements, and its property and its property and its property and its property.
	probate requirements, creditor-proofing, powers of attorney, in-trust accounts, and standard execution requirements.
Section 24	This Section discusses consumer redress mechanisms and includes some recommendations relating to them, including the need for further work to be done in this area.
Section 25	This Section discusses various investment fund maters relating to the need for additional prudential oversight and operating standards and procedures.
Section 26	This Section contains concluding remarks.

# Key Perspectives Underlying Review Recommendations

**Integration, Simplification and Rationalization:** There is a need to integrate, simplify, streamline and rationalize regulatory requirements to serve the needs of the marketplace and of consumer/investors better. This theme underlies all of the recommendations in the Review.

The continued maintenance of a segmented regulatory structure based either on:

- (i) the fragmented "four pillar" structure of the financial services industry that existed prior to the 1987 deregulation, or
- (ii) the fragmented "inter" or "intra" provincial (and national) boundaries;

is becoming increasingly unrealistic.

Retailization of the Marketplace: The convergence of strategic forces has resulted in:

- (i) the consumer/investor (the "retail investor") assuming more market risk; and
- (ii) providers of financial services (the "institutional investor") structuring their activities and products to reduce or eliminate their market risk

This in turn has led to:

- an increased blurring of the lines between the "retail investor" and the "institutional investor";
   and
- (ii) a corresponding erosion of the basis for distinguishing regulatory strategies on whether the investor is a "retail investor" or an "institutional investor".

In this environment, the regulatory and supervisory structure should not favor the "institutional investor" over the "retail investor". This is particularly so when the institutional investor may also be a purveyor of advice, services and products to the individual consumer/investor.

**Fusion of Product, Function and Advice:** The activities of industry participants have outpaced the regulatory structure.

The focus of all participants in the financial services industry is on asset gathering, asset allocation services and asset management. This has led to substantial changes in the distribution side of the financial services industry as it shifts from a transactions-based business to a fee-based, relationship-driven, financial services business. It has also led to substantial changes in the advisory side of the financial services industry as it shifts to gain access to distribution.

The need to gain market share and to protect income streams has resulted in the blurring (and in some instances the fusion) of product, function and advice. With this fusion has come a corresponding shift in emphasis from the success of the transaction to the success of the portfolio. The traditional product sales-oriented representative has evolved into, or is being replaced with, a relationship-oriented advisory-based manager whose role is focused on acting as a conduit for internal or external money managers.

Banks, insurance companies, trust companies, credit unions, independent mutual fund management organizations, independent mutual fund dealers and full service investment dealers now offer, directly and indirectly, a full range of investment advisory services and products. Many of them, through equity interests in other segments or sectors of the financial services industry or through strategic alliances, have multiple entries into both advisory and distribution channels.

The activities carried on, directly or indirectly, by the various sectors of the financial services industry are virtually indistinguishable to consumer/investors yet they are often regulated differently. Regulation (or the lack thereof) varies according to the type of investment product, the institution issuing the investment product and the intermediary involved. The ramifications of these differences are not readily apparent.

It is difficult for consumer/investors to identify:

- (a) when they are being provided with independent advice; and
- (b) when they are simply being sold a product which may be:
  - (i) advice packaged as a product,
  - (ii) a proprietary product, or
  - (iii) a product in which the seller/adviser has a substantial financial interest.

**Common Regime for Money Management:** Consumer/investors are seldom able to appreciate the legal and regulatory differences that flow from the different types of investment participations or products. This highlights the need for there to be a uniform regime for all aspects of money management and to bring all aspects of money management for others under a common regulatory structure.

Integrated Needs of the Consumer/Investor: The needs of consumer/investors are integrated.

Consumer/investors have recognized the need to adopt sound personal financial management practices. A large number have recognized that because of their lack of personal financial expertise (or their lack of confidence in their expertise) they will need to turn to someone for advice. When they turn to an adviser for advice they expect (and should be entitled to expect) that:

- (i) the advice they get will be based on a full client-needs assessment;
- (ii) the person providing the advice and the related financial planning/investment advisory services will have the capability to make that assessment;
- (iii) the person providing them with financial planning/investment advisory services will be able to (and will) provide them with advice that is in their best interests;
- (iv) this advice (and the plan to implement such advice) will be based upon a full range of money management services and products being available to meet their goals and objectives; and
- (v) the ability to carry out transactions to implement the plan and to provide ongoing monitoring and reporting services is and will be ancillary to the foregoing.

In order to meet these reasonable expectations, a fresh approach to regulatory structure, educational courses and proficiency requirements, and to registration requirements is required. The changes should be enabling (as opposed to maintaining or creating barriers) provided that minimum standards based on competence and safety are met.

In addition, systemic changes in the regulatory system and in basic laws relating to the ownership of property, securities laws, corporate laws, personal property security laws, bankruptcy and insolvency laws, and tax laws are also needed to make these laws work together as effectively as they should to promote economic well-being.

Reducing the Knowledge Gap: The single most important thing that could be done to benefit the consumer/investor would be to reduce the knowledge gap - i.e. the gap between those who know and those who do not. This knowledge gap, which economists refer to as "informational asymmetry", usually operates to the disadvantage of the consumer/investor and results in consumer/investors receiving too little value at too high a cost. This is what happens when "sellers" know a lot more about the nature of the product and what the services are than the buyers generally do. This is what happens when the economic interests of the "sellers" and the "buyers" are not aligned. Alignment of economic interests and knowledge and awareness usually go hand-in-hand.

Measures to enhance the knowledge and awareness of consumer/investors should reduce the knowledge gap. Enhanced knowledge and awareness should equip consumer/investors with the basic life skills that they need for informed decision-making, including the ability to understand information that is communicated to them. It should empower consumer/investors to take charge of their own economic well-being.

Information is a critical tool and it is a major equalizing force. It is a valuable commodity. Access to it in an open and timely manner is a crucial ingredient to leveling the playing field. However, if consumer/investors do not have the fundamental ability to understand and make use of the information that is communicated to them, none of the other consumer protection remedies (particularly those based on disclosure) will work effectively.

Knowledge and awareness are the keys to economic well-being. According to the World Bank's 1998 *World Development Report*, knowledge has become so critical to economic and human development that countries that do not make it their top priority will find themselves "mired in poverty" in the next century. The World Bank notes that both developing and developed countries need to make the creation and flow of knowledge their primary concern.

The need to enhance knowledge and awareness and to build key competencies for consumer/investors, industry participants and regulators alike to enable them respectively to make use of the information that is communicated to them is a theme that underlies all of the recommendations in the Review.

# **Blueprint for Change**

The proposals in this Review create a blueprint for change in the regulatory and supervisory structure to reflect the changes that have occurred in the marketplace, and to address the needs of all participants in the marketplace better, including those of consumer/investors. The proposals recommend new approaches to regulatory and supervisory structures and strategies. They propose systemic changes to securities legislation and they highlight the need for changes in basic laws relating to the ownership of property, securities laws, corporate laws, personal property security laws, bankruptcy and insolvency laws, and tax laws to make these laws work together as effectively as they should. The proposals focus on ways to build on existing regulatory structures and strategies and to simplify, streamline and coordinate processes to make them work more effectively together. The proposals envisage (and are dependent upon) the involvement, cooperation and agreement of all levels of government.

It is important to recognize that the proposals are conceptual in nature. Filling in the details will involve a coordinated, concerted effort on the part of governments, industry and consumer/investors. It must be a community effort fuelled by leadership, a common vision and unity of purpose. Collectively, we need to find a way to give meaning to the often overworked words of "coordination, cooperation and harmonization". No one sector or regulatory group or government acting alone has the ability or power to do this.

Although we need to act together, the reality is that nothing will happen unless it is made to happen by a leadership role being assumed by those at the highest levels of the federal and provincial governments and industry.

We need to remember that it is people who make things happen and that it is people who keep things from happening. There is a positive role for everyone to play. The well-being of Canadians depends upon it.

### INVESTMENT FUNDS IN CANADA AND CONSUMER PROTECTION

# STRATEGIES FOR THE MILLENNIUM

### **GLOSSARY**

The following terms used in this Review have the following meanings:

**1995 Report** refers to the Report prepared by Glorianne Stromberg for the Canadian Securities Administrators that was issued in January 1995 entitled *Regulatory Strategies for the Mid-*'90s - *Recommendations for Regulating Investment Funds in Canada.* 

**Canadian Securities Administrators** or **CSA** refers collectively to the securities regulatory authority in each of the provinces or territories of Canada.

**CDS** refers to The Canadian Depository for Securities.

**CDIC** refers to the Canada Deposit Insurance Corporation.

CIPF refers to the Canadian Investor Protection Fund.

**CICA** refers to the Canadian Institute of Chartered Accountants.

FPSCC refers to the Financial Planners Standards Council of Canada.

**IALS Report** refers to the Canadian Report on the International Adult Literacy Survey entitled Reading the Future: A Portrait of Literacy in Canada which was released in September 1996. The IALS Report followed the December 1995 release by Statistics Canada and the Organization for Economic Cooperation and Development of the International Adult Literacy Survey entitled Literacy, Economy and Society.

IDA refers to the Investment Dealers Association of Canada.

IFIC refers to The Investment Funds Institute of Canada.

**IOSCO** refers to the International Organization of Securities Commissions.

**MFDA** refers to the Mutual Fund Dealers Association of Canada referred to in subsection 3.9 of this Review.

MRRS refers to the mutual reliance review system described in subsection 3.10 of this Review.

**Non-Partisan Standards Council** refers to the non-partisan standards council described in subsection 15.10 of this Review

**OECD Paper** refers to the Background Paper by Glorianne Stromberg entitled *Regulation and Supervision of Investment Funds in the New Financial Landscape - A Canadian Perspective* dated June 12, 1997, which was presented as part of the background material for the third session of the Expert Meeting on Institutional Investors organized by the OECD Committee on Financial Markets in July 1997. (1998 OECD Publications, *Institutional Investors In The New Financial Landscape*, page 449).

**OSFI** refers to the Office of the Superintendent of Financial Institutions.

**OSC** refers to the Ontario Securities Commission.

**Regulator** refers to the integrated regulator contemplated by the proposals contained in Sections 10 and 12 of this Review that would be created pursuant to intergovernmental agreement.

**SEDAR** refers to the CSA's System for Electronic Document Analysis Retrieval.

**Single SRO** refers to the single self-regulatory organization described in subsection 16.3 of this Review that would be recognized by the Regulator for the purpose of mandatory membership.

**SRO** refers to the strong, independent, effective self-regulatory organization described in Section 9.02 of the 1995 Report that would operate on a national basis in which membership would be mandatory for everyone (regardless of their status) who is selling securities, including investment fund securities, to the public.

**TSE** refers to The Toronto Stock Exchange.

### INVESTMENT FUNDS IN CANADA AND CONSUMER PROTECTION

# STRATEGIES FOR THE MILLENNIUM

### 1. TERMS OF REFERENCE

### 1.1. The Task

In February of 1998, I was asked by Industry Canada's Office of Consumer Affairs to review the requirements for the reasonable protection of the retail consumer/investor who is increasingly turning to investment funds as a source of retirement income.

The review was prompted by the growing realization that investment funds have become "consumer goods" and that it was appropriate to consider whether there are reasonable protection measures in place to protect the consumer/investor.

### I was asked to:

- assess where the retail consumer/investor was prior to the 1995 Report<sup>1</sup>, the changes that were made after the 1995 Report that have benefited retail consumer/investors and what still needs to be done;
- identify impediments to changes that would benefit retail consumer/investors;
- recommend what facilitating role governments could play to bring about changes that would increase individual consumer/investor sovereignty and control over their assets, with specific recommendations for initiatives that could be undertaken by the Committee of Consumer-Related Measures and Standards<sup>2</sup>;
- identify mechanisms that would enable retail consumer/investors to better identify their goals, the means of achieving them and to assess risk; and
- review consumer redress and compensation mechanisms with a view to identifying how the retail consumer/investor receives greater protection or less depending on where the purchase is made.

### 1.2. The Process

As I did when I prepared the 1995 Report, I turned to the investment fund industry for assistance. This assistance was unstintingly given and I am most grateful for it.

<sup>&</sup>lt;sup>1</sup> See Regulatory Strategies for the Mid-'90s - Recommendations For Regulating Investment Funds in Canada prepared by Glorianne Stromberg for the Canadian Securities Administrators, January 1995.

<sup>&</sup>lt;sup>2</sup> The Committee of Consumer-Related Measures and Standards is composed of federal, provincial and territorial officials whose mandate is to monitor, implement and administer specific commitments made under Chapter Eight of the Agreement on Internal Trade by the federal, provincial and territorial governments relating to consumer-related measures and standards.

The process has involved extensive consultation with industry participants, their advisers, and industry organizations. I also sought input from industry regulators. The consultation process primarily involved personal interviews and meetings. This was supplemented by letters, telephone, fax and e-mail communications from people across the country expressing their views. Some industry organizations, in response to my invitation to do so, invited their members to contact me directly to express their views. This contact was extremely helpful.

The terms of reference did not contemplate that any independent economic analysis would be conducted or that any focus groups would be held or that any surveys or polls would be undertaken. Accordingly, I did not undertake any such work.

Industry participants did, however, make available the results of some of the empirical research that they had carried out on issues that are also relevant to this Review. To the extent that I refer to such research in this Review, I have identified the sources.

My comments are based on information that has been provided to me by knowledgeable industry participants and on my assessment of this information. There is a high level of consistency in the identification of the issues and in the identification of the solutions. As might be expected, people differed on the details and on the timing of initiatives to address the solutions.

It is a tribute to the industry that so many people with such competing interests in today's increasingly competitive marketplace made such a substantial commitment in time and resources to participate in the Review. I want to acknowledge publicly and to express my thanks for the unqualified help and support that was provided to me by so many during the course of the Review.

It is important to note that the comments, views and recommendations that I express in this Report are my own. They are not necessarily shared by the Ontario Securities Commission or by any of the other Canadian Securities Administrators or anyone involved with any of the Canadian Securities Administrators.

### 1.3. The Position of the Consumer/Investor Prior to 1995

To assist in understanding the position of the consumer/investor prior to the release of the 1995 Report, the Review begins with an outline of:

- (i) what the purpose of the 1995 Report was;
- (ii) what the areas of concern were;
- (iii) what the key recommendations were; and
- (iv) what the perspectives underlying the recommendations were.

This outline is contained in Section 2 of the Review.

### 2. THE 1995 REPORT

### 2.1. A Review of the Investment Fund Industry in Canada

In 1994 the Ontario Securities Commission commissioned a review of the investment fund industry in Canada for the benefit of all of the securities regulators in Canada. This Review was prompted by the

rapid growth of the mutual fund industry<sup>3</sup> and the fact that it had been at least 25 years since the last major review of this industry had been undertaken.<sup>4</sup>

The main reason for the 1994 review was to be sure that regulation was keeping pace with changes in the marketplace, to identify any changes that were needed to align industry interests with those of investors, and to respond to industry vulnerabilities. In particular, recommendations were requested on the role of self-regulatory organizations and on how the systems and procedures of the Canadian Securities Administrators<sup>5</sup> ("CSA") could be structured for more effective and efficient regulation of investment funds.

To this end, the 1994 review included:

- (i) an identification of the significant trends in the industry and their impact on regulation;
- (ii) an assessment, from a regulatory point of view, of the problems affecting the investment fund industry and investors and how they might be remedied; and
- (iii) an identification of the emerging issues in the investment fund field that should be addressed by the CSA and their impact on the resources that are required by the CSA.

In view of the importance of the investment fund industry to Canada and the investing public, industry leaders, as well as regulators, recognized that it was an appropriate time to review practices and refocus on objectives. Early in the process, the industry recognized that the results of the review would be largely dependent upon the quality of the industry's input into it. As a result, people made a real commitment to participate in the review. The review served as a catalyst for industry participants to focus on what was happening in the industry, what the concerns were and how they could be meaningfully addressed.

According to information provided by Investor Economics Inc., investment funds' penetration of the Household Balance Sheet (Canada) at the end of 1997 is approximately 20.2% (1996 - 15.7%).

Information provided by CIBC Wood Gundy Inc. indicates that at the end of 1997 the approximate amount per household that Canadian households have invested in mutual funds is C\$24,452 (1996 - C\$18,557). 37.4% of United States households owned mutual funds at the end of 1997 (1996 - 37.2%) with the amount of money per household being US\$44,915 (1996 - US\$35,777). CIBC Wood Gundy Inc. advises that accurate figures relating to the number of households in Canada that own mutual funds are not available. Anecdotally the estimated number ranges from 35% to 40%.

Ernst & Young's analysis from its 1998 report on *The Canadian Investment Funds Market* indicates that in May 1998 household penetration for Canada was 38% (United States - 37.5%) and that the average Mutual Fund Assets per Fundowning household is C\$73,300 (United States - C\$156,000; US\$104,600).

<sup>&</sup>lt;sup>3</sup> In 1993 assets invested in mutual funds grew from C\$67.3 billion to C\$114.6 billion. The extraordinary growth has continued. According to statistics issued by IFIC, at August 31, 1998, total assets under management by IFIC Members were about C\$286.2 billion (July 31, 1998 - C\$320.1 billion). These assets are held in approximately 39.1 million unitholder accounts maintained by IFIC Members. Appendix A sets out statistical information issued by IFIC for the ten year period ended August 31, 1998.

<sup>&</sup>lt;sup>4</sup> See the Report of the Canadian Committee on Mutual Funds and Investment Contracts - A Provincial and Federal Study published in 1969 by the Queen's Printer for Canada.

<sup>&</sup>lt;sup>5</sup> In Canada, jurisdiction over securities matters is exercised by its ten Provinces and two Territories. The term "Canadian Securities Administrators" or "CSA" refers collectively to the securities regulatory authority in each of the Provinces and Territories of Canada.

Industry participants were, and continue to be, proud of the fact that the industry has grown to the size it has without there being any major problems. They recognized, and continue to recognize, the importance of ensuring that the public's expectations are met. Industry participants stressed, and continue to stress, the importance of there being the right regulatory environment designed to support the integrity and trust upon which the industry is founded but freed of impediments such as interprovincial or national barriers that make it difficult and costly for the industry to develop and function on a national basis or globally.

Industry participants were, and continue to be, mindful of the potential benefits that could flow to Canada by encouraging the development of a Canadian-based investment fund service industry that has access to and is supported by the technological infrastructure necessary to provide services not only to Canadian-based operations but also to global ones. To encourage and sustain this development in Canada, they advised, and continue to advise, that there is a need for Canada to provide highly-trained people and a competitive tax and regulatory environment.

The 1994 review culminated in a report (the "1995 Report") issued January 31, 1995 entitled *Regulatory Strategies for the Mid-90s - Recommendations for Regulating Investment Funds in Canada.* The 1995 Report discusses:

- (i) the major strategic forces affecting the investment funds industry in Canada;
- (ii) what is happening in the industry as a result of pressures brought to bear on investment fund managers, changing consumer attitudes, and the competition for consumer savings;
- (iii) the major factors contributing to the widely-held view that changes in the securities regulatory system are needed to serve the needs of the marketplace better;
- (iv) the major areas that present regulatory challenges; and
- (v) the proposals that were made for a new regulatory framework.

### 2.2. Post 1995 Report Activity

<a href="http://www.osc.gov.on.ca">http://www.osc.gov.on.ca</a>.

The 1995 Report generated a lot of discussion and a lot of activity. Virtually every firm and organization involved in the financial services industry formed its own committees and/or participated in industry committees to consider the recommendations.

### **Investment Funds Steering Group**

A key group was the Investment Funds Steering Group, a joint industry-regulatory group appointed under the auspices of the CSA, that in turn established several working committees to consider the recommendations. In November, 1996, the Investment Funds Steering Group issued its report<sup>7</sup> which in effect validated the findings contained in the 1995 Report, endorsed the core strategic recommendations that were made for dealing with the identified challenges and made

<sup>6</sup> A copy of the 1995 Report and of the Briefing Notes that were issued in connection with the 1995 Report may be obtained from the Ontario Securities Commission, 8<sup>th</sup> Floor, P. O. Box 55, 20 Queen Street West, Toronto ON M5H 3S8 - Telephone: 416-593-8117; Fax: 416-593-8122. The Report and Briefing Notes are also available on the Ontario Securities Commission's website at

<sup>&</sup>lt;sup>7</sup> A copy of the Investment Funds Steering Group Report, *The Stromberg Report: An Industry Perspective*, may be obtained from the OSC as noted in Footnote 6.

recommendations for actions to implement the strategic recommendations. In many cases the recommendations coincided with those contained in the 1995 Report; in other cases they did not.

# **CSA Implementation Committee**

Following the release of the Investment Funds Steering Group Report, the CSA established a CSA Implementation Committee to consider the 1995 Report and the Investment Funds Steering Group's comments thereon and to recommend a course of action for the CSA. The CSA Implementation Committee's proposals for action were approved by the CSA in April of 1997.8

### Discussion and Debate

The Investment Funds Steering Group's review was only part of the process for digesting the 1995 Report and its implications. The 1995 Report stimulated intense public discussion and debate both before and after its release. It caused many organizations to look inward to their own processes and to make some fundamental changes. It sparked renewed effort on the part of IFIC to contribute to the regulatory and self-regulatory policy process. It brought together people who were not accustomed to talking with each other and provided them with a forum for discussion that contributed to improved knowledge and understanding of each other's needs and expectations. In doing so it served as a means for identifying areas where common goals were shared.

### **Quebec Consultative Committee**

The 1995 Report also prompted the Quebec Securities Commission to establish its own Consultative Committee. The Report of the Quebec Consultative Committee was issued at the end of January 1997. One of the members of this Committee was also a member of the Investment Funds Steering Group. The responses in the Quebec Consultative Committee Report and in the Investment Fund Steering Group Report are substantially the same.

# **CICA Study Group**

In addition, The Canadian Institute of Chartered Accountants ("CICA") established a Study Group as part of its continuing research program to review the current practices of Canadian investment funds with a view to providing guidance to the industry and users on accounting and financial reporting matters. The CICA Research Report was issued in May of 1997. It is anticipated that as a minimum the CICA Research Report will be used by all types of investment funds as supplementary guidance to the recommendations in the CICA Handbook as to what constitutes "generally accepted accounting principles". The CICA Research Report is supportive of the recommendations made in the 1995 Report about accounting and financial reporting matters.

### International Attention

The 1995 Report has been broadly distributed throughout the world and is receiving considerable attention in various countries where comparable regulatory challenges exist.

<sup>&</sup>lt;sup>8</sup> See Report of The CSA Investment Fund Implementation Group Endorsed by the Canadian Securities Administrators dated May 15, 1997 published in the OSC Bulletin at (1997) 20 OSCB 2512.

<sup>&</sup>lt;sup>9</sup> Financial Reporting by Investment Funds, CICA Research Report, May 1997 published by the CICA.

### Consumer/Investor Attention

Most importantly, the 1995 Report has caught the attention of individual Canadians and has made them aware of issues that impact their well-being and what they should expect from the investment fund industry and its regulators. The impact of this increasing knowledge and awareness of individual consumer/investors is being felt by fund managers and by fund distributors alike and should ultimately have an impact on how the industry is regulated.

### 2.3. Areas of Concern Identified in the 1995 Report

The major areas of concern identified in the 1995 Report relate to:

- (i) how investment funds are structured;
- (ii) how they are sold;
- (iii) the inherent conflicts of interest with respect to both the structuring of investment funds and the distribution of their securities that can result in the interests of the consumer/investor not being placed first;
- (iv) the adequacy of the proficiency and training of some of the people who sell investment funds and of some of the people who manage funds;
- (v) the adequacy of the resources (human, technical and financial) of some industry participants to carry on the activities that they carry on;
- (vi) conflicts of interest with respect to the management of the investment portfolio;
- (vii) the fact that many investors seem to have an imperfect understanding of investment issues;
- (viii) the fact that the disclosure system in place in Canada is ineffective in actually informing investors of material information they need in order to make a reasoned investment decision and is perceived by many in the industry (as well as by investors) as being irrelevant to them;
- (ix) the fact that there is a lack of comparability of performance information about investment funds both within the same product category and with their functional equivalents;<sup>10</sup> and
- (x) the fact that the securities regulatory regime in Canada is cumbersome and costly to comply with and lacks the resources to monitor compliance with its requirements or to respond in a timely manner to the changing marketplace.

-

<sup>&</sup>lt;sup>10</sup> Examples of investment funds that are the functional equivalent of mutual funds but which are not subject to comparable regulation even though they are made available to the retail investor include certain pension and other retirement plans, variable investment contracts and annuities offered by insurance companies, and certain types of pooled funds and wrap accounts offered by various dealers in securities, investment advisers and financial institutions. These investment fund "look-alikes" present a problem both from the perspective of the investor and from the perspective of the mutual fund manager. The problem for the investor lies in the lack of prospectus-level disclosure about the investment fund, the lack of comparability of performance and other information about the investment fund and the lack of required adherence by these investment fund "look-alikes" to merit regulation standards. The problem from the perspective of the mutual fund manager lies in the lack of a level playing field while competing for the same business. Further problems relate to the differences in proficiency and education and other licensing standards for the intermediaries through whom these products are made available.

### 2.4. The Perspective Underlying the 1995 Report Recommendations

The recommendations made in the 1995 Report reflect the following perspectives:

- what is good for the investor is good for the industry and will foster efficient and effective capital markets;
- the industry includes all aspects of money management;
- the foundations of the industry are integrity and trust;
- because there is frequently a lack of knowledge and a lack of bargaining power on the part of many investors, the concept of "buyer beware" cannot by itself be permitted to govern the activities that result in individuals investing their money in investment funds; there is a need, in this situation, to have regard for the interests of investors by ensuring that the system operates fairly and openly;
- there is a need to increase the knowledge and awareness of investors and to ensure that investors have timely, meaningful and relevant information to assist them in making their investment decisions and in monitoring the same;
- the line between selling and advising has substantially disappeared and there is a need to increase the proficiency and training of the persons dealing with the public to enable them better to serve the needs of investors:
- there is a need for **better** regulation rather than for **more** regulation; and
- within this environment, the regulatory strategy should be to allow competitive forces to operate with minimal intervention.

It is this perspective that has generated the attention of investors in Canada and has struck responsive chords globally.

# 2.5. Strategic Recommendations Made in the 1995 Report

Given the nature of the concerns identified in the 1995 Report in connection with the Canadian investment fund industry, it is not surprising that the recommendations made in the 1995 Report to address the problems center around two core concepts - "fairness and integrity" and "information and knowledge". The recommendations contemplate a strong self-regulatory role for the industry that is supported by a strong regulatory structure that is founded on these core concepts. The recommendations are aimed at encouraging and assisting the industry and its regulators in addressing the areas of concern proactively and, in doing so, to continue to meet the public's expectations and to merit the public's confidence.

The framework for the recommendations is based upon proposals for:

- 1. Centralized, coordinated, streamlined, functional regulation; this would include provisions for:
  - (1) simplifying the regulatory structure by combining provincial resources to create a single centralized unit to regulate investment funds:

- (2) establishing a Joint Regulatory Coordination Group consisting of representatives of the Canadian banking, trust, loan, insurance, pension and securities regulators (including the self-regulatory organizations and the customer protection organizations) to provide a coordinating mechanism among the member regulators to ensure that there are no regulatory gaps arising from: (i) deregulation of the financial services industry, (ii) the entry of new industry participants into the marketplace, and (iii) the multiplicity of regulators having jurisdiction over various aspects of the business of the various participants in the industry; another function of the Joint Regulatory Coordination Group would be to provide a similar coordinating mechanism with their international counterparts;
- treating all types of arrangements whereby money is managed on a collective basis, directly or indirectly, for individuals as "investment funds" and bringing all such arrangements under a common regulatory structure; what is contemplated by this recommendation is a regime that would build specific requirements for specific types of investment funds on a uniform base of core principles (e.g. good faith requirements) that would be applicable to all investment funds regardless of what kind they were or who was offering or sponsoring them;<sup>11</sup>
- 2. strong, effective self-regulation by the industry in respect of the management and distribution of investment funds based on: (i) high ethical standards, (ii) fair practice and business conduct rules, (iii) effective and efficient internal and external systems, controls and procedures, and (iv) pro-active and timely monitoring; in this respect:
  - (1) the framework contemplates that a single, strong, effective self-regulatory organization would be established that would operate on a national basis and in which membership would be mandatory for everyone who sells securities (including mutual fund securities);<sup>12</sup>

<sup>11</sup> The perspective underlying this recommendation is that regulatory requirements in respect of investment funds should apply equally to all industry participants and investors should be assured of certain common standards regardless of whether they are dealing with an independent investment fund organization, a bank, a trust company, a life insurance company or other financial institution, an investment counsel or an investment dealer, a mutual fund dealer or some other category of dealer or a financial planner.

<sup>12</sup>This self-regulatory organization ("SRO") would be required to adopt rules of fair practice and business conduct that would, among other things, address industry sales practices. It would also set standards for education, proficiency and training, advertising of performance and other investment information, capital requirements, internal systems, controls and procedures for SRO members, insurance and bonding requirements and contingency fund participation. In addition, the SRO would be responsible for monitoring compliance with its rules and for conducting supervisory audits.

The importance of this recommendation cannot be underestimated. Securities regulation in Canada is founded on the principles of: (i) disclosure of information, and (ii) the registration of persons who deal with the public or provide investment advice. The problem with the current regime on the registration side relates to the fact that not all persons who sell investment fund securities to the public are required to be a member of a recognized self-regulatory organization and to adhere to the higher standards of the self-regulatory organization with respect to rules of fair practice and business conduct, education and training, capital, bonding and insurance requirements and contingency fund participation. Most importantly, those persons who are not members of a recognized self-regulatory organization are not subject to the compliance monitoring procedures of the self-regulatory organization. Investors are often not aware of the different standards or of the impact that the differences in these standards can have. These differences result in potential problems for investors. These differences also present an economic problem for dealers who are members of a self-regulatory organization who are concerned about the lack of a level playing field when they are required to adhere to higher standards than their non-member counterparts while competing with them for the same business.

- the framework also contemplates that managers of investment funds would continue to be regulated by the Securities Regulator directly through registration with the securities regulator, with conditions of registration being attached designed to provide for the abovementioned high ethical standards, rules, systems, controls and procedures and for monitoring compliance therewith through a system of self-assessment review procedures comparable to those presently required of financial institutions in respect of deposits covered by deposit insurance;<sup>13</sup>
- 3. improved corporate governance provisions in respect of investment funds including:
  - (1) the enactment of an investment funds statute to provide a common statutory framework for the constitution and governance of investment funds, the need for which under Canadian law is discussed in Section 18 of the 1995 Report;
  - (2) provision for independent boards at the management level and at the fund level;
  - improved oversight provisions through enhancing human, technical and financial resources to ensure the adequacy thereof for the functions carried out;
  - (4) codes of ethics and business conduct;
  - (5) proposals for dealing with conflicts of interest including those arising from the desire to:
    - (a) engage in principal trading with a related party;
    - (b) deal in securities of or guaranteed by a related party including securitizations and mortgages sourced by a related party;
    - (c) participate in related party underwritings;
    - (d) execute portfolio transactions through an affiliate of the investment fund manager or portfolio adviser;
    - (e) execute portfolio transactions through persons (or their related parties) who distribute securities of the investment fund (i.e. reciprocal trading);
    - (f) engage in inter-fund trading;
    - (g) create certain fund-of-fund structures; and
    - (h) engage in certain soft dollar transactions:<sup>14</sup>

<sup>&</sup>lt;sup>13</sup>The 1995 Report recommends that the Securities Regulator retain direct responsibility for the registration of investment fund managers by reason of the fact that the functions of a manager of an investment fund are so closely linked with the functions of its sponsored investment funds. As a result, it would be very difficult and inefficient to attempt to separate the regulation of the two. It was therefore recommended that the establishment of investment funds and the disclosure documents relating thereto should remain the direct responsibility of the Securities Regulator and should not be delegated to a self-regulatory organization and that the registration function with respect to managers of investment funds should also remain the direct responsibility of the Securities Regulator. See Section 14 of the 1995 Report for recommendations as to the standards required to be met in order to act as an investment fund manager.

<sup>&</sup>lt;sup>14</sup> See Sections 19, 20, 21 and 22 of the 1995 Report.

- 4. increased emphasis on educational and proficiency requirements for industry participants aimed at providing industry participants with:
  - (1) better training and proficiency skills;
  - (2) better awareness of ethical standards, fair practice and business conduct rules; and
  - (3) better ability to meet client needs and expectations;
- 5. increased emphasis on the importance of investor education aimed at improving the ability of investors to:
  - (1) identify, request, review and understand the information needed to assess investment recommendations made to them;
  - (2) apply the information to their own situation in making investment decisions; and
  - identify, request and review in a meaningful manner the information needed to monitor their investments on a continuing basis and assess whether adjustments are needed;
- 6. realignment of the elements of the disclosure system aimed at:
  - (1) integrating primary and secondary disclosure requirements;
  - (2) improving disclosure requirements to ensure that the information is timely, meaningful and relevant; and
  - (3) ensuring that disclosure is integrated and continuing;
- 7. establishment of a basis for achieving comparability of performance information about different investment fund products and between investment fund products of the same type; this proposal contemplates:
  - (1) the development of uniform rules respecting the calculation and use of performance information that would be required to be universally used by all industry participants with respect to performance information that such industry participants may choose to give to, or in respect of, all types of managed accounts; and
  - (2) that it would be desirable (and probably essential) that international standards be developed in this respect:<sup>15</sup>

this need for comparability is an important component of the disclosure proposals contained in the 1995 Report.

### 2.6. Relationship of the Strategic Recommendations to Each Other

The last four recommendations are integral to each other and to the efficacy of the continued reliance by governments and regulators on the adequacy of disclosure as an appropriate strategy on which to base the future well-being of consumer/investors. To continue to rely on the strategy of disclosure

\_

<sup>&</sup>lt;sup>15</sup> See Section 24 of the 1995 Report.

necessitates finding a better way to make sure that information that is timely, meaningful, and relevant to consumer/investors is communicated in plain language to them in a way that a reasonable person acting reasonably should be able to understand. It is submitted that information that has neither been communicated to, nor understood by, the consumer/investor cannot be considered to have been disclosed to the consumer/investor.

This strategy of disclosure is based upon the assumption that the consumer/investor has the knowledge and awareness to understand the significance of the information provided (or not provided) to the consumer/investor and to apply it to the consumer/investor's own circumstances. It also presupposes that industry participants and regulators have a comparable level of knowledge and awareness to understand the significance of the information that is provided (or not provided) in order for them respectively to carry out their responsibilities to consumer/investors adequately. Accordingly, increasing the knowledge and awareness on the part of industry participants, consumer/investors and regulators alike is an integral part of the process of achieving effective disclosure.

The need to increase knowledge and awareness goes beyond the fact of asymmetrical information flows in the marketplace. It goes to the fundamental ability of the consumer/investor to assume responsibility for his or her own well-being, whether acting directly or through an intermediary. Effective access to information and the ability to use it levels the playing field and enables the adoption of a regulatory and supervisory structure that allows competitive market forces to operate with minimal intervention.

The balance of the recommended strategies are structural in nature and/or go to the fundamental integrity of the investment fund product. All of the strategies are based on investing in solutions rather than paying for problems. They are based on "getting it right" in the first place. This is a well-recognized and well-proven strategy for enhancing one's competitive position in the marketplace.

# 3. CHANGES SINCE THE 1995 REPORT THAT SHOULD BENEFIT THE CONSUMER/INVESTOR

### 3.1. Post 1995 Report Changes in the Investment Fund Industry

There have been significant changes in all aspects of life during the last four years. One of the areas where the impact of these changes is most apparent is in the investment fund industry. The changes, for discussion purposes, are divided into three main categories

The first category relates to changes that have been made since the 1995 Report that should benefit the consumer/investor. These changes are discussed in this Section 3.

The second category relates to federal and provincial governmental policy initiatives that have been undertaken since the 1995 Report that will affect the consumer/investor. These changes are discussed in Section 4 of this Review.

The third category relates to changes in the strategic environment that have an effect on the investment funds industry. These changes are discussed in Section 5 of this Review.

### 3.2. Heightened Awareness of Issues

Probably the most important result of the discussion and activity generated by the 1995 Report has been people's heightened awareness of issues. This has happened at all levels regardless of whether one is a regulator, an investor, an industry participant or a combination of the three.

A major contributing force to this heightened awareness of issues is the interest that the media has taken in investment fund issues. The media has played an important role both in identifying and keeping these issues in the public light and in helping people to understand their significance.

Another factor is the increase in the number of books, newsletters, websites and other publications that deal with mutual funds and other investment products and with financial and retirement planning and investing generally. This ready availability of information has been a major contributing force to people's knowledge and awareness.

### 3.3. Industry-Generated Initiatives

The 1995 Report prompted a lot of organizations to implement initiatives without waiting for industry or regulatory endorsement. Industry participants tell me that changes have been made in the way business is done and that a lot of organizations are using the recommendations in the 1995 Report as benchmarks or standards against which to measure the changes.

In the case of fund managers that are public companies, industry participants have noted that their boards of directors have taken a heightened interest in their managed funds and that the audit committees of the fund managers increasingly are reviewing the financial statements of the managed funds, including the expense allocations between the managers and their respective managed funds. I also am told that more attention is being paid to the adequacy of controls and to code of conduct issues.

Another area where change is occurring is seen in the fact that more fund management groups have established, or are in the process of establishing, independent fund advisory boards.

### 3.4. Sales Practices Rule

On May 1, 1998, the CSA's National Sales Code Rule No. 81-105 together with the related Companion Policy No. 81-105 CP became effective. 16 This Rule is the result of three years of industry and regulatory efforts to establish rules of fair practice and business conduct relating to the distribution of securities of a mutual fund. The Rule and Companion Policy reflect the industry and regulatory responses to the concerns about sales practices and business conduct raised in the 1995 Report.

### 3.5. Mutual Fund Prospectus Disclosure

The CSA's proposals for mutual fund prospectus disclosure are contained in the proposed National Instrument 81-101, Forms 81-101Fl and 81-101F2 and Companion Policy 81-101CP. These proposals have been published for comment in the OSC's Bulletin dated July 31, 1998. The proposals are posted on the OSC's website<sup>17</sup> and on IFIC's website.<sup>18</sup> The proposals are discussed in Section 17 of this Review.

<sup>&</sup>lt;sup>16</sup> National Instrument 81-105 and Companion Policy 81-105CP together with the related Notice of Rule and Policy were published in the OSC Bulletin in February of 1998 at (1998) 21 OSCB 747. Earlier versions were published for comment in the OSC Bulletin in August of 1996 at (1996) OSCB 4727 and in July of 1997 at (1997) 20 OSCB 3879.

<sup>&</sup>lt;sup>17</sup> See Footnote 6.

<sup>&</sup>lt;sup>18</sup> See Footnote 19.

### 3.6. Code of Ethics on Personal Investing

In May of 1998, IFIC released the Final Report of its Code of Ethics Committee on Personal Investing. This Report is IFIC's response to the concerns about personal trading raised in the 1995 Report and to the increased public attention that is being focused on personal investing by portfolio managers and other persons having access to a mutual fund's trading practices.

Each Manager member of IFIC is required to adopt, and to confirm that it has adopted, a personal investing code of ethics containing the basic principles that IFIC has determined that all Manager members of IFIC must include in their personal trading codes. This must be done by December 31, 1998. IFIC has published a Model Code that incorporates its required basic principles in order to provide guidance to its Manager members who wish to write their own codes.

### 3.7. Personal Rate of Return Calculation Standard

Another initiative that will be helpful to consumer/investors in monitoring the performance of their investments is the adoption by IFIC of a standard formula for calculating personal rates of return.

In a press release issued June 4, 1998, IFIC announced that its Members who choose to provide personal rates of return as a value added service have agreed to use the methodology and to adhere to standards of presentation determined by IFIC and outlined in IFIC Bulletin (number twenty-one). The methodology that is required to be used is the Modified Dietz Method with monthly linking. IFIC's announcement states that this methodology will provide investors with a percentage return figure indicating how their investments have performed rather than simply how the fund performed.

IFIC Members have the option of deciding: (i) whether to prepare rates of return for investors, and (ii) which clients, if any, will receive the information. The industry standard (Modified Dietz Method with monthly linking) is required to be identified as such when providing investors with personal rates of return. Members may, in addition, provide rates of return using another formula but all relevant differences from the IFIC standard must be identified in the documents in which the personal rate of return appears at the first time of presentation.

In announcing the adoption of its standard, IFIC stated that to the best of its knowledge, the Canadian mutual fund industry is the only mutual fund industry group in the world to tackle the issue of personal rates of return and to come up with a consensus.<sup>20</sup>

### 3.8. Standard for Categorizing Mutual Funds

Over the past several months, suppliers of fund data have produced what they describe as a set of rigorous criteria to determine how to categorize mutual funds. Their proposals have been developed by a committee composed of representatives from Portfolio Analytics Limited, BellCharts, The Financial Post, Southam News, FundData, The Fund Library, The Globe and Mail and Fund Masters. IFIC has participated as an observer. The committee has agreed to:

<sup>19</sup> Final Report of the Code of Ethics Committee on Personal Investing, May 1998 published by IFIC. A copy of the Report is posted on IFIC's website at <a href="http://www.ific.ca/regulation">http://www.ific.ca/regulation</a>.

<sup>&</sup>lt;sup>20</sup> See IFIC's News Release dated June 4, 1998 - Mutual Fund Industry Standardizes Personal Rates of Return.

<sup>&</sup>lt;sup>21</sup> See Rudy Luukko, Fund Watch, February 1998 - *The Drive for Clearer Fund Categories. Also see* Jean Murphy, Investment Executive, July 1998 - *New mutual fund categories in use later this year* and September 1998 - *Categories, criteria to be released this month.* 

- establish clear criteria and apply the criteria using known holdings data as well as investment objectives established by the fund companies;
- publish all ranking and "quartile" information based on these fund categories so that each provider of mutual fund information is consistent in terms of ranking individual funds with others;
- make public, via the Internet and other means, the final document that details the criteria, philosophy and membership of each fund in each category so that the process is completely transparent to the investing public; and
- review the category membership of each fund annually, while maintaining a monthly watch on new funds with the result being possible reclassification.

In September 1998, the committee plans to distribute its proposed criteria and new fund categories to industry participants (including fund companies and analysts) and the relevant media for a three week review and comment period. The committee plans to release the final criteria with the resulting fund categorization in October 1998.

There is a special significance to the agreement that has been reached on common categorization standards. The significance of this agreement is that mutual funds will always be found in the same category regardless of which newspaper, newsletter, software program or other medium is providing the information. This has been an area of great discrepancy because prior to the categorization agreement, the different suppliers of data used different sets of criteria to categorize funds. This discrepancy has enabled fund companies to cherry-pick its rankings and to advertise the supplier's ranking that was most favorable to it.

The categorization of investment funds is very important for consumers and financial advisers. Shortcomings in fund categorization criteria and the resulting rankings have been a concern for some time. The need to develop standards in this area was discussed in the 1995 Report.<sup>22</sup> The end result of the new categorization criteria will be a new tool for consumers and financial advisers.

### 3.9. Mutual Fund Dealers Association of Canada

In 1997, IFIC and the IDA were asked by the OSC to come forward with proposals for a self-regulatory organization for mutual fund dealers. This request was endorsed by the CSA. The proposals have resulted in the recent incorporation of the MFDA. It is intended that the MFDA will ultimately be recognized throughout Canada as a self-regulatory organization and that it will be mandatory for each mutual fund dealer that is not a member of the IDA or of a stock exchange that is a self-regulatory organization, to join the MFDA.<sup>23</sup>

Both IFIC and the IDA will play a key role in the MFDA. The objective is to take advantage of IFIC's knowledge and experience in the mutual fund industry and of the IDA's experience and credibility as a self-regulatory organization to build a new self-regulatory organization from the ground up in a short time frame.

\_

<sup>&</sup>lt;sup>22</sup> See the 1995 Report - Categorization of Investment Funds at page 209 and at page 221.

<sup>&</sup>lt;sup>23</sup> The Request for Comments on Proposed Rule 31-506 - SRO Membership - Mutual Fund Dealers was published in the OSC Bulletin of June 19, 1998 at (1998) 21 OSCB 3875.

To this end, governance of the new self-regulatory organization will be shared by IFIC and the IDA. The Chair of the MFDA will be the IFIC Chairman. The President and Chief Executive Officer of the MFDA will be the IDA President and Chief Executive Officer. IDA supervisory staff will provide management, direction and administrative support for the MFDA. The chief operating officer of the MFDA will be appointed by the board of directors of the MFDA. The MFDA board of directors will consist of 21 individuals with equal representation from fund distributors, the IDA and the public. It has been agreed that an independent mutual fund dealer, who is not a member of IFIC or of the IDA, will be a member of the board of directors and that the IDA bank-related board nominees will be limited to three.

Five committees have been established to address key issues integral to the development of the MFDA's self-regulatory by-laws and regulations. These committees will provide distributors with the opportunity to shape and inform policy and make recommendations to the board. The committees include IDA members and CSA observers and, in most cases, one of the public directors. Fund managers will be invited to provide their perspective. The five committees will deal with (i) capital requirements and an investor protection fund, (ii) proficiency and continuing education, (iii) sales practices, reporting and advertising, (iv) administration, and (v) franchising.

### 3.10. Mutual Reliance Proposals

While the goal of creating a single securities regulatory body in Canada remains elusive, there is a spirit of renewed cooperation among the ten provincial and two territorial commissions. The end result of this initiative, if successful, will be to create a "virtual" single securities regulatory body for people to deal with. At the heart of this cooperative effort are proposals for mutual reliance by the respective CSA members on work that will be substantially performed only by the "designated jurisdiction" for the issuer or registrant. The mutual reliance program is proposed to extend to the filing of prospectuses, applications for exemptive relief and the registration of dealers and advisers. The program is premised on reaching agreement on: (i) common standards for selective review criteria, (ii) a common approach to how matters are reviewed, and (iii) uniform or harmonized regulatory requirements. Each jurisdiction will retain the ability to opt out.

The CSA have published for comment the memorandum of understanding that sets out the working arrangement among the CSA.<sup>24</sup> The Mutual Reliance Review System ("MRRS") is expected to be effective in early 1999. The MRRS will apply initially to the (i) filing of prospectuses (including mutual fund prospectuses) and initial annual information forms, (ii) registration applications for advisers and dealers who are members of a self-regulatory organization, and (iii) applications for discretionary relief. It is anticipated that the MRRS may be extended to other categories of filings such as continuous disclosure documents and rights offering circulars and related documents.

The CSA in its Request for Comments<sup>25</sup> state that they agree that in addition to simplifying and facilitating the use of the regulatory system for filers, an objective of the MRRS is also to effect a unified approach to the many aspects of securities regulation in Canada. The CSA state that in order to achieve this objective, the Memorandum of Understanding provides for:

1. Participation in staff development and training programs organized by CSA committees to standardize the review processes and to develop common approaches to issues.

\_

<sup>&</sup>lt;sup>24</sup> The Request for Comments on the Mutual Reliance Review System Memorandum of Understanding is published in the OSC Bulletin of June 19, 1998 at (1998) 21 OSCB 3882.

<sup>&</sup>lt;sup>25</sup> See Footnote 24.

- 2. Consultation to promote consistency and communication in the review of material filed pursuant to the MRRS and to coordinate changes to the MRRS.
- 3. The provision of adequate funding by the participating CSA to support the training and consultation components of the MRRS.
- 4. The possible coordination of the utilization of personnel among themselves under which the staff of one securities regulatory authority may be utilized by another in fulfilling its obligations as principal regulator.

### 3.11. Cooperation between Insurance and Securities Regulators

In June of 1998, provincial insurance regulators and securities regulators agreed to work together in determining how segregated funds should be regulated. The federal Office of the Superintendent of Financial Institutions ("OSFI") is also participating in this initiative. It is expected that the working group will present its recommendations in the fall of 1998.

### 3.12. Coordinating Mechanism for Regulation

On June 22, 1998, the Ontario Insurance Commission issued a discussion paper on the regulation of insurance distribution and the coordination of financial services regulation in Ontario. <sup>26</sup> The discussion paper proposes that a body to be known as the Ontario Council of Financial Regulators ("Council") would be established. The Council would be composed of the Chief Executive Officers of the Financial Services Commission of Ontario, the OSC and the proposed Insurance Distribution Regulatory Board. The discussion paper notes that:

"The Council would have a mandate to enhance the quality and efficiency of the regulation of financial services in Ontario by:

- identifying ways to coordinate activities, reduce duplication and avoid unintended gaps or conflicts in the regulatory net,
- making recommendations to their constituent bodies on effecting complementary regulatory initiatives,
- coordinating activities with their counterpart financial services regulators in other jurisdictions within and outside Canada.

The proposals in the discussion paper are the first public indication that the recommendations made in the 1995 Report for a cross-sector, coordinating mechanism for regulation<sup>27</sup> is receiving some attention in Canada. The proposals also draw on the recommendations relating to regulatory structure made in the OECD Paper.

<sup>&</sup>lt;sup>26</sup> Discussion paper on Regulation of Insurance Distribution and Coordination of Financial Services Regulation in Ontario, dated June 1998 and issued by the Ontario Insurance Commission (Financial Services Commission of Ontario). A copy is posted on its website at <a href="http://www.ontarioinsurance.com">http://www.ontarioinsurance.com</a>. Paper copies in English can be obtained by calling 1-800-263-7965. Copies in French can be obtained by calling 1-800-668-5821.

<sup>&</sup>lt;sup>27</sup> See Sections 5, 7 and 8 of the 1995 Report.

### 3.13. CICA Audit of Investment Funds Research Project

In the Spring of 1998 the CICA initiated a research project to identify and address audit issues that relate to the recommendations and conclusions set out in the CICA Research Report *Financial Reporting by Investment Funds.*<sup>28</sup> It is anticipated that this project will:

- (i) express the Study Group's views on what action should be taken by the CICA's Auditing Standards Board with respect to the audit of investment funds;
- (ii) provide guidance to auditors and other interested parties on various issues that pertain to investment funds; and
- (iii) offer suggestions to regulatory authorities on what changes should be made to existing requirements.

### 4. CHANGES RESULTING FROM GOVERNMENTAL POLICY INITIATIVES

### 4.1. Federal Government Policy Initiatives

In response to the current economic and social environment and the impact of this environment on the marketplace several initiatives are underway. These include proposals for social program and pension reform, a review of the foreign property tax rules for pension and other retirement plans, the establishment of a task force on financial institution reform and an examination of the role of the institutional investor by the Standing Senate Committee on Banking, Trade & Commerce.

### 4.2. Provincial Government Policy Initiatives

In Ontario, pension, trust and insurance regulation have been brought under the umbrella of the Financial Services Commission of Ontario.

In Quebec, new initiatives are underway aimed at regulating the distribution of financial products and services by creating a single financial services bureau to be known as the *Bureau des services financiers*. The Bureau's mission will be to protect the public. It is proposed that the powers to regulate representatives will be shared between the Bureau and the Quebec Securities Commission based on whether the representatives pursue activities in the insurance or financial planning sectors or the securities sector.

### 4.3. Self-Funding

The chronic under-funding problems of the various CSA - or of at least the four largest jurisdictions - have been addressed. Self-funding is now in place for the securities regulatory authorities in the Provinces of Alberta, British Columbia, Ontario and Quebec and provision has been made to enable them to operate in an autonomous, but nevertheless accountable, manner.

-

<sup>&</sup>lt;sup>28</sup> See Section 2.2 of this Review under the sub-heading, CICA Study Group.

### 5. CHANGES IN THE STRATEGIC ENVIRONMENT

# 5.1. Strategic Forces

In 1994, the key strategic forces driving the growth of the investment fund industry centered around:

- (i) the economic and demographic factors relating to an aging population;
- (ii) there being fewer people supporting such population;
- (iii) the lack of the likelihood of lifetime employment;
- (iv) the likelihood of substantial underemployment; and
- (v) the recognition of the inadequacy of public and private pension plans to provide for retirement needs.<sup>29</sup>

These strategic forces still continue to affect the investment fund industry. However, in 1998, these forces are subsumed to even broader strategic forces that have emerged. The predominant strategic forces in 1998 include:

- (i) the impact of globalization;
- (ii) the reality of financial services deregulation;
- (iii) the impact of information technology;
- (iv) the impact of advanced communications facilities; and
- (v) the strengthening of the free market economic system.

These are all enabling forces and they have converged to empower individuals as well as institutions. This convergence has in itself resulted in the emergence of the consumer/investor (a term whose bottom-line meaning ultimately describes us all) as a strategic force in his or her own right.

The impact of the emergence of the consumer/investor as a strategic force in the investment fund industry is placing demands and expectations not only on the marketplace but also on the regulatory and supervisory framework. Our current regulatory and supervisory framework is simply not structured to meet these demands and expectations.

### 5.2. The Changing Face of the Investor in the New Financial Landscape

One of the major changes that has occurred is in respect of the marketplace itself and in the "face" of the institutional investor. The financial marketplace has traditionally been viewed as being composed of "institutional" investors and "retail" investors. Traditionally, the institutional investor has been exempt from the application, and the benefit, of various provisions of securities legislation on the theory that the institutional investor is able to look out for its own interests.

<sup>&</sup>lt;sup>29</sup> These forces are discussed in Sections 2 and 3 of the 1995 Report.

This approach was reflective of an era when the institutional investor generally was a financial institution (such as a bank, trust company, loan company, insurance company) or a defined benefit pension fund or endowment fund investing money for its own account. Regulation and supervision of such institutional investors was primarily focused on solvency issues and prudential regulation aimed at ensuring that such institutions would be able to meet their respective obligations to depositors, policyholders, plan participants and beneficiaries as the case may be. Investor protection was not part of the regulatory or supervisory focus for such institutions.

In the current Canadian financial landscape, the face of the institutional investor has changed. This is partly a result of the deregulation of the financial services industry that began in Canada in 1987 when restrictions on the nature of the activities in which regulated financial institutions could engage were substantially eliminated. This enabled such financial institutions to expand their money management and other business activities substantially.

While institutional investors still manage money for their own account, an increasing amount of the money that is managed by institutional investors is being managed in a representational capacity rather than for the institutional investors' own accounts. The fact that institutional investors in managing money in a representational capacity have discretionary authority (whether limited or unlimited) does not change the fact that these institutional investors are managing **other people's money** and that their authorization to continue to manage such money may be terminated on relatively short notice or, in some cases, no notice.

The increasing representational role of the institutional investor is seen not only in the case of the traditional "financial institutions" but also in the case of pension and retirement plans, which are increasingly being structured as defined contribution plans (instead of defined benefit plans) or as registered retirement savings plans. In the case of pension plans, one of the implications that flows from the shift from the use of defined benefit plans to the use of defined contribution plans to provide retirement benefits is that the money held in defined contribution plans or registered retirement savings plans is being managed in a representational capacity for the direct benefit of the plan participants whose rights to the accumulated assets held for their respective accounts are fully vested in them.

This is in contrast with the ownership interest that the plan sponsor of a defined benefit plan has in the assets held in the defined benefit plan where the investment performance of such assets directly impacts on the plan sponsor's funding obligations and its liability for any benefits' shortfall.

Although mutual funds (which are separate legal entities formed at the instance of a mutual fund management organization as a vehicle to enable individual investors to pool their money for investment purposes) are regarded as institutional investors and are sometimes described as being "financial institutions", it should be kept in mind that:

- (i) in Canada there is no constating statutory basis for their constitution and governance as a "financial institution"; and
- (ii) the assets of a mutual fund are invested in a representational capacity for investors who are entitled, usually on a daily basis, to demand repayment of their interest in the mutual fund at the net asset value thereof.

## Retailization of the Marketplace

The purpose of focusing on the distinction between the management of money by an institutional investor for its own account and the management of other people's money is to highlight the increasing "retailization" of the institutional investor and the resulting "retailization" of the marketplace when

considering what regulatory and supervisory policies are appropriate for a marketplace that is increasingly made up of "instividual" investors.<sup>30</sup>

This "retailization" is occurring not just as a result of the expanded powers of financial institutions but also as a result of the convergence of several factors including:

- (i) the increased awareness of individual investors of the need to move beyond deposit-type savings and life insurance to provide for their well-being,
- (ii) the ease with which individual investors are able to invest not only in Canada but throughout the world as a result of new and improved technology and communication facilities, and
- (iii) the increasingly global and free market perspectives underlying governmental policies.

The characterization of the marketplace as a "retailized" or "instividual" marketplace contrasts with what is perhaps the more common characterization of today's marketplace as one that is "increasingly institutionalized".

I do not think that the mindset reflected in this latter characterization is supported in today's reality. What is becoming evident is that the lines between the "retail investor" and the "institutional investor" are increasingly blurring and the basis for distinguishing regulatory strategies on whether the investor is a "retail investor" or is an "institutional investor" is eroding.

From the regulatory and supervisory perspective, the fact of whether an individual holds his, her or its investment interest in underlying securities directly or through some sort of collective investment vehicle should make little difference. This is particularly important when viewed in conjunction with the secular shift of household assets from deposit-type savings and life insurance products to investments (with investment funds being the investment vehicle of choice) that has occurred and is continuing.

The individual consumer/investor now has a direct ownership interest in his or her investments that fluctuates according to the market value thereof as opposed to simply having a contractual right to receive an agreed-upon payment or payments at a stipulated time or times.

The individual consumer/investor who holds his or her investments either directly or through some sort of collective investment vehicle is in direct competition with the institutional investor that invests for its own account. In this environment, the need for equality of regulatory treatment takes on a greater dimension. It is important that the regulatory and supervisory structure not favor the institutional investor over the individual consumer/investor, who traditionally is referred to as the "retail investor". This is particularly so when the institutional investor may also be a purveyor of advice, services and products to the individual consumer/investor.

# 5.3. Industry Consolidation

\_

The period since 1994 has been marked by a lot of consolidation in the financial services industry. This consolidation is taking place in every aspect of the industry and is a major factor that is shaping the development of the investment fund industry in Canada.

<sup>&</sup>lt;sup>30</sup> One of the fundamental questions that needs to be asked in the light of this change in the marketplace is whether the so-called institutional investor should continue to be exempt from the application, and from the benefit, of various provisions of securities legislation on the theory that the institutional investor is able to look out for its own interests.

I simply mention industry consolidation as a strategic force<sup>31</sup> without going into detail. The implications of the industry consolidation that is taking place could well be the subject of a separate review.

However, the implications of industry consolidation will have an important bearing on the development of regulatory and supervisory policy which will need to be adapted to the reality of the industry structure that is resulting from the industry consolidation that is occurring.

The recommendations that I make in this Review regarding regulatory and supervisory structure are focused on addressing the reality of the emerging industry structure. The blurring of product and function that is discussed in the next two subsections is both (i) an example of the changes in industry structure that will require a change in the regulatory and supervisory approach, and (ii) a force in its own right.

# 5.4. The Blurring of Product and Function and its Consequences

Since 1994 another key factor that is having a significant impact on the marketplace is the increased blurring of product and function. This has occurred as the different segments of the marketplace continue to compete for the hearts and minds of the consumer/investor.<sup>32</sup> The increased blurring of product and function results from the need to gain market share and to protect income streams.

For example, the need to gain market share and to protect income streams has led to:

- (i) life insurance companies increasing their focus on investment products such as segregated investment funds, universal life policies that are indexed to the performance of a pool of investment securities, and guaranteed investment funds;
- (ii) banks focusing on how they can combine guaranteed deposit obligations with a market-linked return; the result has been to create guaranteed investment certificates where the return is indexed to the performance of a market-linked index and to create debt obligations that are linked to the performance of designated mutual funds or other identifiable pools of assets or commodities;
- (iii) mutual fund management organizations exploring ways to offer mutual funds with a guaranteed return; some mutual fund organizations have chosen to enter into arrangements with an insurance company to offer their mutual funds with an insurance wrapper thereby converting the offering of their mutual funds into a segregated fund that is not subject to regulation under securities legislation; at least one mutual fund organization is offering its mutual funds with a guarantee that is issued by an insurance company;
- (iv) independent mutual fund distributors establishing their own proprietary investment funds which are sold to their clients; by doing so, these distributors not only receive the distribution fees but they also receive the investment advisory and business management fees that their clients pay for the privilege of investing in their sponsored funds;<sup>33</sup>

\_

<sup>&</sup>lt;sup>31</sup> One could well debate whether the consolidation that is taking place in the financial services industry is a strategic force in its own right or is simply the result of other strategic forces that are at work.

<sup>&</sup>lt;sup>32</sup> See Section 3 of the 1995 Report.

<sup>&</sup>lt;sup>33</sup> This capturing of the management fee revenue stream is seen as the distributors' response to the growing recognition that distribution fees will come under pressure from an increasingly knowledgeable consumer/investor. The creation of proprietary funds by the independent mutual fund distributors is referred to in the industry as "backward integration".

- (v) mutual fund management companies increasingly focusing on sales and marketing activities as opposed to money management activities; and
- (vi) banks, insurance companies, trust companies, credit unions, independent mutual fund management organizations, independent mutual fund dealers and full service investment dealers offering, directly and indirectly, a full range of investment advisory services and products.

It should be noted that many of the entities referred to in clause (vi) above, through equity interests in other segments of the industry or through strategic alliances, have multiple entries into the manufacturing and distribution channels. The packaging of these services and products is virtually indistinguishable to the consumer/investor. Unfortunately, it is all too common that the consumer/investor does not know what he or she has invested in. All too often the consumer/investor does not know whether it is a mutual fund, a segregated fund, an equity-linked GIC, a wrap account or whatever and is unable to describe the nature of the underlying investments.

# Sales-Focused Fund Management

One of the fundamental challenges that the investment fund industry faces today is how to ensure that the exigencies of marketing pressures do not prevail over the fiduciary obligations of the investment advisory activities.

The increased focus by mutual fund management companies on sales and marketing activities as opposed to money management activities reflects the pressure that results from the mutual fund management companies having competing securityholders' interests<sup>34</sup> that make it necessary for the mutual fund management companies to focus on increasing their sales in order to increase assets under administration to generate the revenues needed to service the various interests. This pressure to manage for sales has become such a predominant force in the investment advisory industry that the line between managing money and simply selling product is no longer clear.

Whole departments have been established to deal with marketing strategies, to provide marketing support to distributors, and to service elaborate dealer compensation payment structures relating to the payment of trailing commissions and to keeping track of deferred sales commission obligations. Where once portfolio management skills were the most sought after skills for mutual fund management companies, marketing skills are now perceived as an even more essential asset.<sup>35</sup> As one fund manager expressed it to me, the fund management industry is becoming a home for marketing personnel who have left the consumer products industry.

<sup>&</sup>lt;sup>34</sup> An increasing number of mutual fund management companies have issued shares to the public. There are potential conflicts between the interests of the shareholders of the mutual fund management companies and the securityholders of the sponsored investment funds. Many mutual fund management companies have also financed the payment of sales commissions in respect of mutual fund securities that are sold on a deferred sales commission basis by raising money from the public in a variety of ways including limited partnership offerings, income trust offerings, securitized offerings and debt offerings. The interests of the various securityholders of these offerings may in some instances conflict with the interests of the securityholders of the sponsored investment funds. See Section 3.01 of the 1995 Report for a discussion of the various pressures on investment fund managers.

<sup>&</sup>lt;sup>35</sup> The ever-increasing costs of distribution that are borne by the fund management companies are reflected in the comments made by a senior officer of a mutual fund organization at its 1998 annual meeting. He explained the substantial increase in the company's cash position as being due to "soft" sales in its funds because the company did not have to pay out as much in up-front commissions to brokers and financial planners in respect of back-end load sales made by such brokers and financial planners. (See Shirley Won, The Globe and Mail, June 25, 1998 - *Trimark unveils segregated funds*.)

Skilled as these marketing personnel may be in the marketing of consumer products, it will be important for fund management organizations to ensure that this expertise is sufficiently combined with a deep-rooted understanding of the fundamental fiduciary obligations that an entity that manages other people's money is bound by.

In other words, it will be important to the consumer/investor that fund management organizations ensure that their marketing personnel understand the inherent essence and obligations of the money management service/product. This need also applies to the proprietary funds that have been and are being created by mutual fund distributors.

# Consequences of the Blurring of Product and Function

There are several consequences that result from the blurring of function and product. These include:

- (i) the difficulty of delineating when and in what circumstances a service becomes a product and vice versa; and
- (ii) the difficulty of identifying the differences among products that are offered as equivalents, with or without added features.

It is difficult for consumer/investors to identify when they are being provided with independent advice and when they are simply being sold a product<sup>36</sup> which may be a proprietary product or a product in which the seller/adviser has a substantial financial interest.

It is difficult for consumer/investors to identify the risks that are present in these products, particularly when the products are marketed as being "guaranteed" or as "protected". It is difficult to identify just what is in fact guaranteed.

It is virtually impossible for consumer/investors to identify the cost of:

- (i) the advice;
- (ii) the service;
- (iii) the product; and
- (iv) the guarantee.

There is no transparency. Without the ability to identify these differences and to understand their associated costs, it is an impossible task for consumer/investors to identify whether their financial planning objectives are being met or to understand the level of risk that they have undertaken.

In this environment, it is no wonder that the consumer/investor is so often left in a state of dependency that is not adequately addressed either by the financial services industry or by today's fragmented regulatory and supervisory structure.

It is particularly unfortunate that this state of affairs is occurring at a time when most people have realized that they have to do something more - both individually and collectively - to assure their future well-being. The reality is that most people are poorly equipped to do so.

<sup>&</sup>lt;sup>36</sup> It should be noted that "products" may include advice packaged as a product.

Lured by the double digit returns offered by mutual funds and stock market indices, people have turned away from deposit instruments to mutual funds and, we are told, have become "investors" rather than "savers". However, many people have done so without comprehending the difference between being an "investor" and being a "creditor" of a regulated financial institution that is backed by governmental safety nets. Many people do not have any comprehension of risk and do not have any idea of how to manage risk. Many people do not appreciate that there is risk inherent in being too conservative in managing their affairs.

Something needs to be done about this. The efforts that are made to remedy this situation need to be supported by the education system, by employment and labor laws, by tax laws, by pension laws, by securities laws and by the basic laws relating to property rights. This support needs to be coordinated and to be complementary across the legal and regulatory system. The recommendations contained in this Review are designed to provide a blueprint for building the underlying framework

People's rights and opportunities should not be permitted to depend on which province or territory that they live in. Their rights and opportunities should not be permitted to be lost in regulatory gaps or to be tied up with needless and costly bureaucracies or institutional power struggles for the right to provide for their needs.

# 6. POST 1995 REPORT EFFECT OF CHANGES ON CONSUMER/INVESTORS

# 6.1. Industry Perspectives on Whether Consumer/Investors Are Better Off Today

This Section of the Review discusses the question of whether the changes that have occurred since the 1995 Report have resulted in consumer/investors being better off today. I asked this question of virtually everyone with whom I spoke during the course of this Review. The answers varied.

On the positive side, a number of industry participants said that:

- the industry has gone a long way in trying to adapt to the recommendations in the 1995 Report and is better than it was in 1994;
- there is no evidence of there being really big problems in the industry;
- there has been lots of growth and new entrants as well as some failures but no one has yet lost any money; however, lots of fine-tuning is needed;
- more attention is being paid to compliance;

• the initiatives aimed at bringing all mutual fund dealers under the umbrella of a self-regulatory organization, the implementation of the sales code and business practices rule, the issuance of guidelines on personal trading by portfolio managers and other access persons, and the enhanced oversight by audit committees and independent boards are all indicators of very positive improvements for the industry and for consumer/investors.

24

\_

<sup>&</sup>lt;sup>37</sup> Given the increased popularity of "guaranteed" products such as segregated funds, variable market-linked universal life policies and index-linked GICs, there is reason to question whether Canadians have really become "investors" rather than "savers".

On a less positive note, a number of industry participants said that there is not a whole lot of difference observing that:

- "shills" are still very much in evidence;
- trips are coming back;
- inappropriate product sales are still being made;
- there are some fundamental structural problems with the investment fund industry that center
  on the fact that consumer/investors think they are dealing with an independent professional
  investment adviser when in fact they often are simply dealing with a salesperson;
- the foregoing situation is aggravated by the fact that there is insufficient control over the use of descriptive terms such as "financial planner", "financial adviser", "financial consultant", "investment adviser", and other terms which do not clearly differentiate between the qualifications of the intermediary as a professional investment adviser able to exercise discretionary authority and the qualifications of a sales representative who is not permitted to exercise such discretion;
- there is still poor disclosure of what the consumer/investor is paying for and what service this includes.

Other industry participants expressed their responses to the question by commenting that the consumer/investor is not worse off and is probably better off. They observed that:

- the market has provided "fabulous returns" and people have enjoyed good returns on their investments;
- the profile of the investment fund industry is higher than it was in 1994 with the result being that more information is readily available;
- the availability of information has increased people's knowledge and awareness of issues affecting their investments; it has heightened their awareness that they have more to learn; and it has made it relatively easy for people who want to learn, to do so;
- there has been a voluntary increase in initiatives to make fund prospectuses more readable, user-friendly and meaningful;
- interim reports are more informative and readable;
- individual rates of return are being provided to consumer/investors along with comparisons to benchmarks.

However, they noted that the improvements in the quality and relevancy of the information are not standard across the board. They also observed that a lot of the information continues to be aimed at the investment adviser rather than at the client who is the consumer/investor.

Some industry participants complained that there is too much reporting and that the reporting is not timely. They complained about the amount of "paper" that is involved. They observed that the popularity of mutual funds has resulted in an explosion in the number of new mutual funds that have been created

with the number that are now offered to the public being almost double the number that were offered four years ago. This development (according to them) has not benefited the consumer/investor.

They expressed concern about the popularity of mutual funds having resulted in the proliferation of products that bear a close resemblance to publicly offered mutual funds but have substantial differences that are not readily apparent to the consumer/investor, including the fact that the products are not regulated.

They observed that one reason why these differences are often masked to the consumer/investor is that the increased visibility and popularity of mutual funds has created a false sense of confidence of "knowing all that there is to know about mutual funds". However, consumer/investors often fail to realize that there are substantial differences between mutual funds and the products that they are being advised to buy and that these differences substantially change the nature of their investment and/or the relationship.

# 6.2. My Evaluation

I believe that all of the observations referred to in subsection 6.1 are accurate. Accordingly, how one evaluates these observations in relation to the question of whether the changes that have occurred since the 1995 Report have resulted in consumer/investors being better off than they were in 1994 depends on whether one sees the universe in terms of the glass being half-full or the glass being half-empty. I prefer the former view.

Having said this, I do not think that there is any room for complacency. Although a lot has been done, a lot more remains to be done.

# 7. IMPEDIMENTS TO CHANGE

### 7.1. Industry Perspectives on Impediments to Change

The second question I asked the people with whom I spoke was what did they regard as being impediments to change. Again, people's responses differed.

Many industry participants said that the biggest impediment to change was the continuation of the bull market. They said that the strength of the market returns that were being enjoyed by consumer/investors masked fundamental structural and operational weaknesses in the industry.

One senior executive of a fund management organization answered the question by saying, "big players; big money - why would anyone change what's successful?"

Another senior executive referred to the high management expense ratios as being a major impediment to change, observing that "it's a pretty profitable world today for the producer and the distributor. They don't want to cut their cloth". Similar views were expressed by other industry players.

Other responses by industry participants to the question included:

• the observation that the industry is made up of different groups with different interests;

- the underlying "politics" of the shifting balance of power among industry participants;<sup>38</sup>
- the regulators; in this respect concern was expressed about:
  - (i) the adequacy of regulatory staffing both in terms of expertise and in terms of the sufficiency of resources directed to investment fund matters;
  - (ii) the length of time it takes for regulatory initiatives to be undertaken and implemented;
  - (iii) the multiplicity of regulators and the lack of awareness and understanding on the part of some regulators;

these factors are seen as serious impediments to change and as contributing to the complexity of actually bringing about change;

- the high cost of communicating fundamental changes to investors because of the high cost of the regulatory fees that must be paid in conjunction with filing amendments to prospectuses;
- the cost of "compliance" i.e. the cost of putting in place, maintaining or enhancing the adequacy of, good internal, controls, systems and procedures - is seen as an impediment to change;<sup>39</sup>
- the lack of understanding by consumer/investors of:
- (i) the fee structure;

(ii) the impact of the fee structure on their total return;

- (iii) fundamental concepts such as the time value of money and the effect of compounding;
- (iv) performance numbers in general;
- (v) the differences between mutual funds, segregated funds (including the different types thereof), pooled funds, wrap accounts, equity-linked GICs, investment contracts, income interests in trusts, and other similar products;
- (vi) the fact that despite the appearance of being provided with independent advice tailored to their specific needs, consumer/investors are often only being sold products;
- the difficulty consumer/investors have in assessing the significance of the information with which they are provided and in knowing how they should use it to determine what they should or should not do.

<sup>38</sup> In this respect, the June 1998 withdrawal by Dynamic Mutual Funds from IFIC is viewed by some industry participants as an impediment to change and not in the best interests of the industry and investors because it fragments both the industry and the industry association in the eyes of investors and regulators.

<sup>&</sup>lt;sup>39</sup> The observation was made that industry producers and distributors would prefer to spend their money on marketing and sales rather than on the enhancement of internal controls, systems and procedures.

# 7.2. Anomalies Driving the Investment Fund Industry

Again, I think there is validity in the observations referred to in subsection 7.1 of this Review. The observations highlight the fact that there are a lot of anomalies that are driving the investment fund industry. The key ones are:

- professionalism vs. marketing (i.e. the provision of professional investment advisory services vs. the sale of product);
- advocating reliance on the principles of integrity and trust while operating on the basis of the principles of opportunism and buyer beware;
- the analysis/paralysis syndrome;
- recognition of the need to change combined with a reluctance to be the first to change because of concerns about the loss of competitive advantage;
- a regulatory system that despite its purpose to protect investors and to foster efficient and fair capital markets:
  - (i) gives rise to opportunities to move between the gaps;
  - (ii) relies on self-regulation but falls short on adequate oversight;
  - (iii) has, in seeking to be responsive to the needs of the marketplace, become too reliant on ensuring that there is unanimous consent to the regulatory provisions and thereby falls short of addressing the needs and expectations of the marketplace; and
  - (iv) has become very technical and legalistic, the result of which has been the emergence of technical schemes that steer their way between the regulations, thereby achieving technical compliance but falling short on substantive compliance.

# 8. INVESTMENT FUNDS AS CONSUMER GOODS OR SERVICES

## 8.1. Viewing Investment Funds as Consumer Goods or Services

This Section of the Review discusses the soundness of the assumption that investment funds are consumer goods or products. The discussion focuses on this from the perspective of:

- (i) whether investment funds are consumer goods or services;
- (ii) whether it makes a difference, from the perspective of the type of problems that consumer/investors who own investment funds encounter, if investment funds are considered to be consumer goods or services; and
- (iii) whether it makes a difference, from the perspective of the remedies that are available to address these problems, if investment funds are considered to be consumer goods or services

While I have always viewed investment funds from the perspective of the fairness to the consumer/investors who invest in them, I have not, in thinking about investment funds, categorized them as being "consumer goods or services".

The traditional thinking about investment funds is that they are a simple means:

- (i) to pool your money with other like-minded investors;
- (ii) to have your money professionally managed by independent investment advisers at a reasonable cost;
- (iii) to diversify your investments;
- (iv) to have the administrative work in connection with owning investments performed in a costeffective manner by persons who are trained to provide such services; and
- (v) to have your investments held in safekeeping by a professional custodian at a lower cost than you could arrange on an individual basis.

To the extent that investment funds are regulated, they are regulated under applicable securities legislation.

This traditional investment fund perspective places the emphasis on investing, with the benefit of independent professional advice, rather than on just purchasing a product (or more accurately, being sold a product).

I have continued to look at investment funds from this traditional perspective despite the increasing tendency on the part of both fund management organizations and distribution organizations over the past several years to view and position investment funds simply as another commodity that is sold.

However, I found when talking with industry participants about whether investment funds were consumer goods or services, a ready acceptance of the perspective that investment funds are "consumer goods or services". This is the reality in which industry participants work and therefore explicit recognition of this reality is welcome.

Several industry participants made comments to the effect that the "securities model" is no longer working and that we need to think in terms of building a "consumer model". They observed that people who are buying investment funds (and in particular mutual funds) have the mindset of a consumer and not the mindset of an investor. As consumers, they are not "investing". They are simply buying a recognized brand name. The particular fund does not matter. What matters is that what they are buying is a recognized brand.

While I have concerns about simply viewing investment funds as products or services that are sold like laundry soap, blue jeans or sports shoes, the reality is that this is how they are being positioned in the public eye and how they are being viewed by many of the key participants in the investment fund industry. Given this reality, it is useful to step back from the specifics of the investment fund industry and to look at investment funds as if they are just a type of consumer product or service. My starting point was to look at the main areas that traditionally have posed problems for consumers.

#### Traditional Problem Areas for Consumers

The traditional problems encountered by consumers with respect to consumer goods or services can be divided into six main areas:

Problems Related to Price Is the price reasonable? Uncertainty over whether a price is
reasonable increases with the complexity of the product or service and with the infrequency of
the purchase of the product or service.

- **Problems Related to Quality or Suitability** Does the product or service meet the requirements for which it is purchased?
- Problems Related to Availability How much choice is there in the marketplace? Is the product or service universally available or are there certain people (e.g. people with low incomes or in regional markets) who have their choices restricted? Are similar products or services being promoted as being different? Are different products or services being promoted as being the same or as equivalent?
- **Problems Related to Safety** Is the product or service as secure as it is professed to be? Is it fit for the purpose that it is professed to be?
- Problems Related to Information Is sufficient information about the product or service available? Problems arise when the available data about the product or service is insufficient, inaccurate or untimely.
- Problems Related to Redress When a product or service is found to be defective or deficient or unfit for the purpose, is access to redress available on a timely basis and at a reasonable cost?

#### Traditional Consumer Protection Measures Used to Address Problems

I then reviewed the traditional consumer protection measures that are used to address the traditional problems encountered by consumers with respect to consumer goods or services. These measures fall into seven main categories.

- Disclosure of the material facts relating to the product or service.
- Registration or licensing of the intermediary and/or of the product or service, with proficiency requirements, capital requirements and insurance and bonding requirements that vary according to the nature of the product or service.
- Prohibition of transactions where there are conflicts of interest.
- Prohibition of unfair trade or business practices.
- Prohibition of misleading advertising.
- Fitness for purpose or product suitability requirements, contractual formalities and withdrawal rights.
- Redress mechanisms if problems occur as a result of the consumer acquiring an unsuitable or defective product or service including compensation funds.

Overlaying all of these measures are efforts to improve the knowledge and awareness of consumers.

# **Comparisons**

I then compared the areas where problems commonly occur in relation to investment funds with the areas where problems commonly occur with products and services that are more generally thought of

as "consumer goods and services". I also compared the remedies used to deal with the respective problem areas.

It is interesting that in both comparisons, there is a perfect correlation. What this shows is that it does not matter whether you characterize investment funds as consumer goods or services. The types of problems that consumer/investors have and the types of remedies for their problems are the same.

If one looks at the securities legislation of the respective Canadian jurisdictions from this perspective, one recognizes in this legislation the use of the same traditional consumer protection measures that are used in respect of other consumer products and services.

It is just that in the effort to deal with the complexity of the current securities legislation and the minutiae of the detail necessary to achieve "certainty," the tendency during the last three or so decades has been not to think about securities legislation in basic consumer protection terms.

If one looks at the 1995 Report, one recognizes that the 1995 Report reflects an approach to investment funds and to the investment fund industry that:

- (i) centers on the traditional areas that have caused problems for consumers; and
- (ii) reflects in its recommendations the application of traditional consumer protection remedies, adapted to reflect the nature of the products and services provided by the investment fund industry.

In other words, we do have a "consumer model" of regulation in place. The challenge is to make sure that this model continues to protect consumer/investors.

It should be helpful to industry participants (and to regulators) to keep this "consumer-model" perspective in mind as they deal with their clients - the public - rather than to think of the provisions of securities legislation as being unique to the financial services industry and as perhaps being unwarranted interference with the way they do business.

It is important to remember that the provisions in securities legislation (and the recommendations that have been made to enhance them) are basic consumer protection measures applied to specific types of consumer goods and services. The refocusing by industry participants and their regulators on the primacy of the consumer protection purpose of securities legislation should be of benefit the public, the consumer/investor.

# 8.2. Consumer/Investor Goals and Objectives

When one views investment funds as consumer goods or services, it is important to focus on some key questions. These include:

- (i) Why does the consumer/investor want to acquire the goods or services?
- (ii) What does the consumer/investor expect to achieve as a result of the acquisition?

An Investors Group/Gallup Canada Survey conducted in May of 1998 indicates that most Canadians who save and invest do so because they want to be financially independent when they retire.<sup>40</sup> The

\_

<sup>&</sup>lt;sup>40</sup> 9<sup>th</sup> Annual Investors Group/Gallup Canada Survey, 1998 Feature Report.

survey indicates that the desire for financial self-sufficiency is what is driving most people to invest. Fear of not having enough money in their old age and of not wanting to be a burden on their family are other motivating factors.

When one focuses on the meaning of "financial self-sufficiency" and on the action verb, "to invest", it is a challenge to reconcile how these concepts can be fulfilled if the consumer/investor is simply **being sold** products. "Financial self-sufficiency" and "investing" are reflective of long-term planning objectives. Buying - or more accurately, being sold - products is more reflective of acquiring something with a view to satisfying an immediate or short-term need that will be consumed within the immediate or short-term time frame.

The resulting mismatch (or potential for mismatch) raises the issue of the suitability of the products for their intended purpose of achieving long-term financial self-sufficiency or, in "consumer terms", it raises the question of the fitness of the product for its purpose and the corresponding issue of product warranty obligations that flow from this.

The issue of suitability of the products or goods for their intended purpose is magnified by the fact that the intermediaries all offer assurances of their integrity and of the products they offer together with assurances that the consumer/investor is justified in placing reliance and trust in both the intermediaries and their products.

When one thinks about the need to adapt the regulatory regime and the way the industry operates, it is important to keep this consumer perspective in mind.

Apart from the need to provide an investment and regulatory climate that is aligned with the needs of consumer/investors to achieve their goals of financial self-sufficiency, there is also a fundamental economic need for the industry to protect itself from liability for claims based on wrongful advice and from product liability claims. One could well envision a scenario where shortfalls in retirement obligations could become the obligation of an industry that may not have the requisite resources to cover the obligations.

There is also the need from the perspective of governmental social and retirement planning policy to ensure the investment and regulatory climate fosters the ability of consumer/investors to meet their goals.

Taking the abovementioned needs into account and having regard to the blurring of functions and to the anomalies driving the investment fund industry<sup>41</sup> - and indeed the investment industry generally - it would benefit us all if some fundamental thinking were to be done about the regulatory regime and how it needs to be adapted to an environment, that although it purports to be an advisory one built on relationships, in reality is still one that is transaction or product-driven and focused on "sales".

We would be well-served if some fundamental thinking were to be done about aligning the compensation system to remove inherent conflicts of interest between the intermediaries (including the

Also, the 1997 Annual SIA Investor Survey: *Investors' Attitudes Towards the Securities Industry* which was prepared for the Securities Industry Association by Yankelovich Partners Inc. in November 1997 indicates that planning for retirement continues to be the most important goal for US investors with 49% of investors saying retirement is their most important goal (down from 55% in 1996) and one-fifth of investors (21%) saying that being prepared for the future is their most important goal (up from 16% in 1996).

<sup>&</sup>lt;sup>41</sup> See subsections 5.4 and 7.2 of this Review.

investment funds) and those who are availing themselves of the intermediaries' services whether as investors, clients or plan participants.

Inherent in this review is the need to examine the appropriateness from the perspective of the consumer/investor of the continuance of the commission-based compensation structure. Complementary to this review is the need to ensure that the professional advisory component of the investment fund industry is not compromised by sales product biases.

I make this recommendation fully recognizing that once something becomes an industry practice, it is unlikely that there will be much enthusiasm for re-examining the practice or for changing it. This reality simply presents an additional challenge for those who think that fairness, effective decision-making. and efficiency makes changes in industry practices necessary.

Alan Greenspan, the Chairman of the United States Federal Reserve Board, expressed this thought recently by saying that change often comes slowly and is viewed as threatening by many. He went on to sav:

"... it is frequently difficult to reform the rules of the game, as it were, because change requires easing rules and opening options for some while increasing competition for others, redrawing lines that create new limits, and applying some pre-existing regulatory structures to new institutions. However, ... our financial system has clearly reached the stage where pressures from the market will force dramatic changes regardless of existing statutory and regulatory limits. The ability of financial managers to innovate and find loopholes seems endless. Recognizing this reality, the congressional landscape appears to have made the decision to attempt to fashion a new set of rules that are both comprehensive and perceived as equitable to all participants."42

Mr. Greenspan's observations about effecting change in the American financial system are equally applicable to describing the challenges faced in Canada in effecting change in the structure, regulation and supervision of the Canadian financial services industry.

#### 9. MAKING A DIFFERENCE FOR CONSUMER/INVESTORS

#### 9.1. **Recognition of Fundamental Needs**

There are innumerable things that would make a difference to consumer/investors. However, before addressing what would make a difference to consumer/investors, there are some fundamental needs that must first be recognized in order to make a difference to consumer/investors. These fundamental needs are discussed in this Section 9.

#### 9.2. Reducing the Knowledge Gap

The single most important thing that could be done to benefit the consumer/investor would be to reduce the knowledge gap - i. e. the gap between those who know and those who do not. This knowledge gap, which economists refer to as "informational asymmetry", usually operates to the disadvantage of the consumer/investor. The need for reducing the knowledge gap was highlighted in the observations of

<sup>&</sup>lt;sup>42</sup> Remarks of Alan Greenspan, Chairman of the United States Federal Reserve Board, at the Charlotte Chamber of Commerce, Charlotte, North Carolina, July 10, 1998.

Keith P. Ambachtsheer<sup>43</sup> during his testimony on November 18, 1997 at the hearings to examine the state of the financial system in Canada (Institutional Investors) held by the Standing Senate Committee on Banking, Trade and Commerce. Mr. Ambachtsheer said:

"The key thing I would look for in hiring a third party outside investment counsellor would be the alignment of economic interests. ... Once you go to a third party, whether it is investment counselling or most of the mutual funds, there is an interesting dichotomy of economic interest in the sense that these businesses are owned by other people who are trying to maximize their own bottom lines. They are in the business of running that business, and they are looking at revenues, less costs, equaling their profits. There is a question as to whether there is a clarity of economic interest between serving the customers well and how they enhance their own bottom lines.

In theory, the market should work these things out, but unfortunately, this is an area of great informational asymmetry where the sellers know a lot more about the nature of the product and what the services are than the buyers generally. Economics tell us that when you have informational asymmetry, generally the outcome is low quality products at too high a price. That is one of the challenges - namely, how do you deal with great informational asymmetry typically between the buyers of investment management services and the sellers of investment? I do not have an obvious answer to that other than raising the general level of clarity about what the product is producing and a level of understanding of capital market efficiency."<sup>44</sup> (Emphasis added)

Until we address this knowledge gap, the consumer/investor will remain seriously disadvantaged in being able to participate in the marketplace as a fully-informed person able to assess on a continuing basis his or her needs and the best means of meeting them.

Until the knowledge gap is closed other regulatory strategies aimed at consumer protection will not be fully effective. This has serious implications for governmental social, economic and retirement policies. It has serious implications for the efficacy (from both a national and global perspective) of the operation of the desired free market economic system and for the desire to provide the least restrictive regulatory alternative necessary for investor protection.

# 9.3. Recognition of the Consumer/Investor's Integrated Needs

The four-pillar structure of the financial services industry that existed before the deregulation of the industry that began in 1987 fragmented client finances among the four categories of players (namely, banks, trust companies, insurance companies and brokerages) with there being a separate regulatory structure for each category of player and with each category of player vying for a larger portion of the consumer/investor's business.

<sup>43</sup> Keith P. Ambachtsheer is an economist and President of KPA Advisory Services Inc., a pension fund advisory firm in Toronto, Canada.

<sup>&</sup>lt;sup>44</sup> See page 23 of the transcript of the evidence for November 18, 1997 presented to the Standing Senate Committee on Banking, Trade and Commerce during its Hearings to Examine the State of the Financial System in Canada (Institutional Investors).

This fragmented structure resulted in both a *product-category* bias<sup>45</sup> and a *firm-product* bias<sup>46</sup> in favor of the firm and created the product-driven business model that continues to this day notwithstanding the 1987 deregulation of the provision of financial services. Apart from any other inadequacy of a product-driven business model, what such a model ignores is the reality that consumer/investors have integrated financial needs.

When one focuses on the integrated financial needs of the consumer/investor, including the need for the consumer/investor to become less reliant on governmental social, economic and retirement plans and the corresponding need to take on the risk (and the opportunity) of participating in corporate and private retirement plans, it becomes clear that a governmental, business or regulatory model that is not driven by the integrated financial needs of the consumer/investor will fall short of both public and private expectations and needs.

When one focuses on the integrated financial needs of the consumer/investor, one recognizes that there is an added challenge and complexity to meeting these needs. This challenge and complexity is created both by the division of powers under the Canadian constitution and by the structural division of the way the powers within each of the respective federal, provincial and territorial jurisdictions are exercised.

One cannot readily change the reality of the constitutional and intra-governmental structures but one can strive to make the structures work better to serve the needs of the Canadian consumer/investor regardless of where he or she lives in Canada. Making the structures work better is the place where the work needs to start. Section 10 proposes a framework for doing this.

#### 10. REGULATORY FRAMEWORK

# 10.1. Regulatory Coordination

One of the most important things that could be done to benefit the consumer/investor would be to review our financial services regulatory regime. We need to look for ways to simplify, streamline and coordinate our processes and to make sure that they are complementary.

In the course of this review, we need to examine the roles of the various regulatory authorities having jurisdiction over investment, insurance, pension and retirement matters including tax, pension, securities, insurance, employment and labor matters.

We need to update, clarify, align and harmonize the applicable laws.

We need to eliminate laws that are no longer necessary or serving the purpose for which they were intended. This needs to happen not only on a cross-regulatory basis within each Canadian jurisdiction

\_

<sup>&</sup>lt;sup>45</sup> A financial adviser representing a specific type of product/service and behaving rationally can be expected to offer that product/service to the client over another thereby introducing a "product-category bias" in relationship with the client (e.g. investment funds).

A financial adviser representing a single product and behaving rationally can be expected to sell that product/service to the client thereby introducing a "product bias" in the relationship with the client (e.g. Brand X investment funds). The single product bias may also be brought about by the fee, commission and other reward structures offered by the suppliers.

<sup>&</sup>lt;sup>46</sup> A financial adviser representing a firm and behaving rationally can be expected to represent the firm's products and services to the client thereby introducing a "firm-product bias" in the relationship with the client (e.g. ABC Bank's products).

but also on a cross-regulatory basis among the thirteen (and soon to be fourteen) governments that are involved 47

Ideally, this review would result in some sort of centralized, coordinated, streamlined and functional regulation or at least in a mutual recognition system that truly is non-duplicative. This could be achieved pursuant to intergovernmental agreements.

The cost of continued, multi-jurisdictional, multi-layered, fragmented regulation is too great a burden on Canadians and is not meeting the integrated financial planning and investment advisory needs of the consumer/investor.

Fortunately, there are indications that these needs are being recognized and that work is under way. Subsections 3.10, 3.11 and 3.12 of this Review deal with important initiatives that are being proposed.<sup>48</sup> It is important that these initiatives be encouraged, harmonized and proceeded with on a timely basis. We will all be disadvantaged if the status quo is permitted to continue.

It is encouraging that one of the areas that was looked at during the hearings of the Standing Senate Committee on Banking, Trade and Commerce<sup>49</sup> was the merit of an integrated regulatory and supervisory system.<sup>50</sup> Other countries including the United Kingdom, New Zealand and Australia are in various stages of moving towards an integrated regulatory and supervisory structure.

Proceeding with the development of an integrated regulatory and supervisory system for Canada would be one of the most important things that could be done to benefit consumer/investors. It is beyond the scope of this Review to try to work out the details of such a system. However, what I have in mind is an integrated regulatory and supervisory system that would focus on:

- (i) prudential and solvency concerns that relate to various activities (such as, for example, banking, insurance and liability trading) of the respective institutions, and
- (ii) the advice-giving/distribution activities of the intermediaries with specific product knowledge being subsumed under the framework of the advice-giving/distribution activities.<sup>51</sup>

Another example of the recognition of the need to streamline and coordinate our processes and to make sure that they are complementary is seen in the submission<sup>52</sup> of Senator Pitfield on the proposals in Bill C-2 with respect to revisions to the Canada Pension Plan and the establishment of the Canada

<sup>52</sup> Senate Debates, December 16, 1997, at pages 817 - 819.

<sup>&</sup>lt;sup>47</sup> One diversified financial services company advises that it has at least 85 regulatory authorities in Canada to which it reports. Much of the information that is required is duplicative in nature. Filing fees and other costs associated with dealing with the various regulatory authorities are very high.

<sup>&</sup>lt;sup>48</sup> Subsection 3.10 deals with the MRRS proposed by the CSA. Subsection 3.11 deals with the agreement between securities and insurance regulators to cooperate. Subsection 3.12 deals with the proposal made by the Ontario Insurance Commission to establish a Council of Financial Regulators.

<sup>&</sup>lt;sup>49</sup> These hearings were held in November 1997 and in May and June 1998 to examine the state of the financial system in Canada (Institutional Investors).

<sup>&</sup>lt;sup>50</sup> One of the recommendations made in the OECD Paper and in subsection 12.2 of this Review is for an integrated regulatory and supervisory system.

<sup>&</sup>lt;sup>51</sup> Also see Sections 15 and 16 of this Review.

Pension Plan Investment Board. In his remarks, Senator Pitfield emphasizes the necessity of streamlining and coordinating social and retirement planning with tax and investment initiatives and appropriate accountability mechanisms.

The Agreement on Internal Trade is yet another important example of the recognition by governments of the need to review "measures". The Agreement on Internal Trade was entered into by the federal and the provincial and territorial governments of Canada. It bears the formal execution date of July 18, 1994 and became effective on July 1, 1995.

While the Agreement on Internal Trade, with certain exceptions, does not for the most part extend to "measures" that are related to "financial services" or "financial institutions"<sup>53</sup>, it does provide a structure for a broad review if the respective governments should wish to use it to extend the initiatives referred to in subsections 3.10, 3.11 and 3.12.

### 10.2. Additional Initiatives

Hopefully, we will see more initiatives that will reflect a fundamental rationalization of how powers are exercised with a corresponding reduction in the burden of regulation and heightened value for consumer/investors.

# 10.3. Leadership and Vision

The single most important thing that is needed at the moment is the leadership and vision to address the identified problems as a whole. At present, there is no single, easily identifiable and accountable body that has the mandate to do this. Accordingly, problems are dealt with on a piecemeal basis and the results are perceived as being less than satisfactory.

Perhaps if we created the regulatory structure (as suggested in subsection 12.2 of this Review), the leadership and vision that is necessary to fulfill its mandate would emerge. The starting point is clarity of vision followed by achieving unity of purpose and gaining the momentum to implement this vision.

Unless we know what we want to do, there is no way that we will ever be able to do anything other than to react to yesterday's problems in an effort not to lose further ground. We therefore need to work together in a coordinated and cooperative way to find a means for allowing competing interests to thrive under the umbrella of some basic common structures.

## 10.4. Provision of Sufficient Resources

There is also the need to dedicate the necessary human, capital and technical resources to implement the regulatory changes that are needed and to address the concerns about the securities regulatory system that are outlined in the 1995 Report and in this Review.

-

<sup>&</sup>lt;sup>53</sup> See Article 1806 of the Agreement on Internal Trade. The terms "financial service" and "financial institution" are defined at some length in Article 200 of the Agreement on Internal Trade. The term "measure" is defined as including any legislation, regulation, directive, requirement, guideline, program, policy, administrative practice or other procedure. The term "harmonization" is defined as meaning "making identical or minimizing the differences between standards or related measures of similar scope". The term "mutual recognition" is defined as meaning "the acceptance by a Party of a person, good, service or investment that conforms with an equivalent standard or standards-related measure of another Party without modification, testing, certification, re-naming or undergoing any other duplicative conformity assessment procedure". These definitions are contained in Article 200 of the Agreement on Internal Trade.

There is a need to apply some project planning skills to the implementation process. The specific recommendations should not be looked at as "stand-alone" items to be prioritized or cherry-picked as to what will be done or not done. Instead, there is a need to stand back and recognize how the recommendations fit together as part of an integrated whole and how they are implementing of each other. There is a need to prepare a project plan for an "integrated whole" with critical paths identified as to what needs to be done in what sequence and what can be done on parallel tracks. Someone needs to be put in charge and to be made responsible for making things happen.

I have identified these needs before<sup>54</sup> and I identify them again. Acting effectively on this identification would go a long way to making a big difference for consumer/investors.

#### 11. IMPLICATIONS OF THE CURRENT LEVEL OF KNOWLEDGE AND AWARENESS

# 11.1. Consumer/Investors Current State of Knowledge and Awareness

This Section of the Review discusses the current level of knowledge and awareness of consumer/investors and what the implications of this are for the regulatory and supervisory framework.

Many people today are inadequately equipped with basic life skills that enable them to identify their needs, to identify the means to achieve them and to evaluate whether the choices they have made are meeting their needs. Many people are unable to read beyond an elementary level and their numerical skills lag their literacy skills.<sup>55</sup> People are vulnerable to being sold a "get rich quick" scheme without truly understanding what they have bought, its suitability for their needs, or the "fine print" that negatively impacts on them. People tend to ignore what they do not understand - particularly if it is intangible. People often spend more time comparison shopping for a television set or planning their vacations than they do planning and managing their affairs to meet their current and future financial needs or choosing a consultant to help them do so.

According to information posted by ABC Canada on its website, <a href="http://www.abc-cananda.org">http://www.abc-cananda.org</a> the reading skills of 16% of Canadian adults are too limited to deal with the majority of the written material they encounter in everyday life. Reading and numeracy skills are closely related. 62% of Canadians have numeracy skills sufficient to handle the numerical tasks normally encountered in everyday life. 24% of Canadians do not possess the necessary skills to meet most everyday numeracy requirements but can deal with familiar documents that require simple math skills such as addition and subtraction. The remaining 14% of Canadians cannot perform numerical operations consistently but can recognize numbers in isolation or in a short text. ABC Canada states that these statistics are based on a 1990 survey, Adult Literacy In Canada: Results of a National Survey, Ottawa: Statistics Canada, 1990.

In September 1996, the Government of Canada issued a report entitled, *Reading the Future: A Portrait of Literacy in Canada*, which is the Canadian Report on the International Adult Literacy Survey (IALS) that was conducted by Statistics Canada and the Organization for Economic Cooperation and Development entitled, *Literacy, Economy and Society*. At the broadest level, the IALS findings are consistent with those that resulted from the survey on which the ABC Canada statistics are based. However, the IALS approached literacy from a different perspective defining it as "the ability to understand and employ printed information in daily activities at home, at work and in the community, to achieve one's goals and to develop one's knowledge and potential".

<sup>&</sup>lt;sup>54</sup> For example, see *Remarks of Glorianne Stromberg Respecting the Response to Stromberg and to the Steering Group Report* at the 1997 Annual Mutual Funds Symposium of The Canadian Institute, Toronto, Ontario - April 14, 1997.

<sup>&</sup>lt;sup>55</sup> "Literacy" according to the definition used by ABC Canada, a Canadian literacy organization, means the information processing skills necessary to use the printed material commonly encountered at work, at home, and in the community. Literacy is the ability to read, write, calculate, speak, and understand as well as sign (for the Deaf) and communicate in other forms of language, according to need. It is a continuum of these skills necessary for everyday life in the home, at work, in education and in the community.

The need to do something about this rather bleak situation is beginning to be recognized by people in their individual capacities as consumer/investors as well as by others who are responsible for the strategic development of business plans, governmental policy and educational and training programs.

The identification of the need of the consumer/investor to adopt sound personal financial management practices and the recognition that most consumer/investors because of their lack of personal financial expertise will need to turn to someone for advice, has created vast business opportunities for those in the financial services sector to convert personal financial management know-how into an economic good that adds value to the customer and for which there is a market.

This conversion of personal financial management know-how into a marketable commodity has highlighted the importance of the need for increasing the knowledge and awareness of industry participants who seek to provide this know-how. In fact, the survival of these industry participants as intermediaries is directly dependent on their ability to do so. I say this because if industry participants fail to use every opportunity to increase their knowledge and awareness and to add value to the services that they provide to their clients they risk losing their clients to their competitors.

Here, the competition is not just from other intermediaries. The competition is also coming from the incredibly powerful tools that technology and communications facilities have made readily available to the consumer/investor at nominal cost. Today, many people in their home environment have access to better state-of-the-art services than sometimes are available in their workplace. Recent Statistics Canada studies indicate that one in three households in Canada now use computers for communication purposes or for surfing the Internet and that 60 per cent of the households that have computers use them at least seven times a week.<sup>56</sup>

The impact of technology on personal financial planning cannot be overestimated. For the first time in the history of the investment funds industry, the consumer/investor has access to information and knowledge that was once the preserve of only the most expert investment professionals. These tools and facilities empower the consumer/investor not only to identify his or her own needs but to compare alternatives and, in the case of investments, to customize his or her own portfolio, to monitor its performance, and to bring it into line when necessary. The usefulness and versatility of, and the access to, these tools increases almost daily.

As Alan Greenspan, the Chairman of the United States Federal Reserve Board, noted in his recent remarks,<sup>57</sup> the rapid growth of computer and telecommunications technology has lowered the cost and broadened the scope of financial services. These developments have made it increasingly possible for the consumer/investor to transact business directly without the need for an intermediary and for a wide variety of financial products to be tailored for very specific purposes. As a result, competitive pressures in the financial services industry are greater than ever before. Technological innovation has accelerated financial globalization and the combination of technological innovation and financial globalization has resulted in expanded cross-border asset holdings, trading and credit flows, with many organizations having increased their cross-border operations. Mr. Greenspan observes that deregulation has been as much a reaction to technological change and globalization as an independent factor. I agree. I also think he is right when he observes that:

"the sharply enhanced market signals emanating from the vast set of technology-driven new products have undermined much regulation which rested on the ability to maintain

<sup>68</sup> Wallace Immen, The Globe and Mail, June 18, 1998, Families with young at home with computers.

<sup>&</sup>lt;sup>57</sup>See Footnote 42.

market segregation ... the continuing evolution of markets suggest that it will be increasingly difficult to support some of the remaining rules and regulations established for a different economic environment."

# 11.2. Implications for the Regulatory and Supervisory Framework

As noted in subsection 5.1, globalization, deregulation, information technology, communications facilities, and the strengthening of the free market economic system - all of which are enabling forces - have converged to empower not only institutions but also individuals. This has significant implications for the regulatory and supervisory framework not the least of which is how to provide for investor protection and market integrity in a marketplace that for all practical purposes has become global.

Investors are becoming increasingly intolerant of what they perceive to be regulatory barriers, maintenance of monopolies, unequal treatment and regulatory inefficiency. At the same time, investors want a regulatory structure that assures fairness, equality of opportunity, integrity of the people and firms who are registered to provide advice and investment services to others, integrity of the product and service provided, effective oversight and monitoring of activities, and speedy action to deal with problems and to keep them from re-occurring.

Two factors that present major challenges to the structure of the marketplace and to the regulatory response thereto relate to conflicts of interest and to what is described as "market fragmentation". These factors converge into the fundamental issue of whether an intermediary can be on both the buy side and the sell side of the same matter. An example where this occurs is the situation where an institution underwrites, either directly or through affiliates, the issue of securities and then exercises discretionary authority to place such securities in its managed accounts. Another example where this question occurs relates to situations where an institution, either directly or through affiliates, engages in principal trading with its managed accounts either directly or through the order flow. The resolution of these issues will have more fundamental implications for consumer/investors and for the functioning and fairness of the marketplace than debating the number of branches of financial institutions that are needed in a community.

Another challenge to the structure of the marketplace and to the regulatory response thereto relates to the need to decipher when the voice of the institutional investor is the voice of hundreds and thousands of individual investors calling for needed market and regulatory reform and when it is the voice of institutional self-interest that is not aligned with the interests of individual consumer/investors, some of whom may be their clients.

### 11.3. Adequacy of the Current Regulatory and Supervisory Framework

When one considers the foregoing in conjunction with the strategic forces outlined in Section 5 of this Review and the consumer/investor goals and objectives outlined in subsection 8.2 of this Review, it

-

<sup>&</sup>lt;sup>58</sup> The term "market fragmentation" is used to describe order routing from the floor of a recognized stock exchange such as The Toronto Stock Exchange or the New York Stock Exchange to a private upstairs trading system (often referred to as a "PUP") operated by a dealer or to a proprietary (private) electronic trading system (often referred to as a "PET") such as Instinet. Another draw of orders from the floor of a recognized stock exchange comes from the "pay-for-order" firms such as Bernard L. Madoff Investment Services.

<sup>&</sup>lt;sup>59</sup> Some financial institutions argue that because the power conferred on them under the legislation deregulating the financial services industry to participate in both the business of underwriting and in the business of providing investment management and advisory services to others allows them to do this, the normal fiduciary obligations (some of which are currently expressed as statutory prohibitions on certain conduct) should not apply.

becomes clear that the current regulatory and supervisory framework may not be serving the consumer/investor as effectively as it should be for two reasons:

- (a) the continued differentiation between retail investors and institutional investors sometimes operates at cross-purposes given the increased retailization of the institutional investor; and
- (b) the fact that notwithstanding the deregulation of the financial services industry that began in 1987, the current regulatory framework has continued to segment the financial services sector along the product lines that were conventionally provided by each of the so-called "four pillars", with there being a separate regulatory structure for each pillar regardless of the fact that each pillar is now able to deal, directly or indirectly, in all products and provide all services.<sup>60 61</sup>

#### 12. SUGGESTED APPROACHES TO REGULATORY AND SUPERVISORY STRATEGIES

# 12.1. Suggested Approaches

The passage of time has confirmed the appropriateness of the directional strategies identified in the 1995 Report to address the issues we are faced with today. These directional strategies are outlined in subsection 2.5 of this Review.

The needs of consumer/investors, and indeed of all market participants, would be effectively met on both a national and on an international basis, by the implementation of these directional strategies. Included in these strategies and the recommendations for their implementation is the need to harmonize national and international regulation and supervision.

There are, however, some additional observations and suggestions<sup>62</sup> that I would make to take into account the clearer trends that have emerged in the intervening four years by reason of the increased global focus of the marketplace, the improvements that have been made in communications facilities, the informational and other technological advances that have been made, and the increasing knowledge and awareness of participants in the marketplace. I also pose some questions with a view to stimulating consideration of alternative structures and strategies.

# 12.2. Regulatory Structure

# Integrated Regulatory Structure

Given the expanded business activities and the diversified geographical operations of today's participants in the financial services sector, there seems to be an even greater need to find a way to bring about centralized, coordinated, cooperative, streamlined and functional regulation.

The continued maintenance of a segmented regulatory structure based either on:

<sup>61</sup> In the context of the integration of financial services provided by institutions, it is interesting to note that mutual fund organizations, (which are sometimes referred to as the "fifth pillar"), through their acquisition of trust companies, have expanded the products and services that they are able to provide and have directly accessed the ability to participate in the Canadian Payments System.

<sup>&</sup>lt;sup>60</sup> See subsection 5.4 - The Blurring of Product and Function and its Consequences.

<sup>&</sup>lt;sup>62</sup> These observations and suggestions are outlined in Section 12 and subsequent Sections of this Review.

- (i) the fragmented "four pillar" structure of the financial services industry that existed prior to the 1987 deregulation, or
- (ii) the fragmented "inter" or "intra" provincial (and national) boundaries;

is becoming increasingly unrealistic.

Section 10 of this Review outlines a suggested regulatory framework that could be brought about by agreement among all levels of government.

Detailed regulation needs to give way to systemic changes that can and will operate on a universal basis in each jurisdiction.

In identifying these systemic changes, there are some threshold questions that need to be addressed such as the following:

- 1. How can regulation be made relevant to today's marketplace?
- 2. Have the factors of globalization, technology and instantaneous communication on a worldwide basis resulted in regulators becoming largely irrelevant to the process?
- 3. Is the process really controlled by the financial institutions and other large transnational corporations who have the ability to operate directly or indirectly and through the flow of capital across borders?
- 4. Is the process really controlled by the software companies and the providers of the communications facilities that make the instantaneous flow of information and capital possible?
- 5. Will the role of regulators be focused increasingly on ensuring that reciprocal enforcement actions are speedily taken in their respective jurisdictions for "fraud on the market" type activities when harmonized standards have not been met so that there is "no place to hide"?

### Common Regime for Money Management

Consistent with the need to de-fragment the regulatory structure, is the need to give high priority to bringing all aspects of money management for others under a common regulatory structure. This arguably goes beyond the recommendation in the 1995 Report which only referred to arrangements whereby money is managed on a collective basis, directly or indirectly, for individuals. However, technological advances have now made it economically feasible to permit customized investment portfolios to be maintained on an individual basis for relatively small amounts that until a short time ago would not have been economically feasible to manage except in some sort of collective investment vehicle. There is a growing recognition that mass marketing can now be reduced to a mass audience of one. These factors make it desirable that there be a uniform regime for all aspects of money management as the consumer/investor is seldom able to appreciate the legal and regulatory differences that flow from the different types of investment participation.

# Effectiveness of the Self-Regulatory System

It is time to take a fresh look at the effectiveness of the self-regulatory component of the securities regulatory regime. Questions that should be asked in the context of this review include:

1. Is there too much "self" in "self-regulation"?

- 2. Would the regulatory system work more effectively if the focus were to change from "self-regulation" to "self-management" of regulatory requirements?
- 3. Have regulators abdicated too many of their functions to self-regulatory organizations?
- 4. How effective is regulatory oversight of the activities of self-regulatory organizations?
- 5. What is the purpose of self-regulatory organizations?
  - (i) Is their purpose to protect the public?
  - (ii) Are they structured appropriately to do so?
  - (iii) Are their functions focused on protecting the public?
  - (iv) How effectively is the public interest addressed and protected?
- 6. How effective is the governance structure of self-regulatory organizations?
- 7. How pro-active are self-regulatory organizations in making timely changes that would benefit the marketplace?
- 8. Do competitive interests interfere with or delay the rule-making process of the self-regulatory organizations?
- 9. Would we be better off to privatize, through outsourcing or otherwise, the performance of certain regulatory activities?
- 10. Does the self-funding status of securities regulators remove or reduce the necessity for regulation being delegated to self-regulatory organizations?
- 11. With governmental securities regulation now being vested in free-standing, self-funded agencies, would it be possible (assuming appropriate restructuring of **what** work is done and the **way** it is done) to eliminate the intervening layer of regulation that the self-regulatory organizations represent?
- 12. Are consumer/investors well-served by the continuation of this two-tier structure of securities regulation?
- 13. How adequately, timely and effectively are the complaints, problems and concerns of consumer/investors dealt with in this two-tier structure?
- 14. With respect to career advancement, how dependent are staff of the self-regulatory organizations on the goodwill of the members?
- 15. Are we continuing to design our processes and procedures as if the significant changes resulting from the advances in technology and communications facilities and self-funding have not taken place?
- 16. If we are, how is this serving the needs of consumer/investors?

- 17. Are some self-regulatory organizations such as stock exchanges with their electronic trading systems or their proposals for such systems now in direct competition with the market participants that they regulate?
- 18. Should there be a separation of member regulation activities, market regulation activities and industry trade association or advocacy activities?
- 19. Is the public interest served by having a multitude of self-regulatory organizations for each "pillar" of the industry and/or for each product sold by members of that pillar?

These are all questions that affect consumer/investors regardless of whether they invest directly or through some sort of collective investment vehicle. There no doubt are more questions. We need to address the questions. We need to address the underlying issues.

# 12.3. International Regulatory Cooperation

Today's technology has effectively eliminated time and distance to create a borderless world in terms of virtual reality. This has made it even more important to coordinate the cross-border regulatory activities in what people still think of being the "real world". Common, practical regulatory approaches to the use of facilities such as the Internet need to be developed. These approaches need to reflect the reality of the marketplace.

Efforts to improve and simplify international regulatory cooperation need to be encouraged at every level. The work of the International Organization of Securities Commissions ("IOSCO") has been an important catalyst in this respect.<sup>63</sup>

IOSCO's work in the investment fund area which resulted in the adoption of principles of regulation for collective investment vehicles as a preliminary step to creating an "IOSCO passport" is of particular interest in the present context. Perhaps it is time to renew the efforts to create an effective IOSCO passport and to define the parameters of its use. For instance, should nationals of one country be able to invest in investment funds domiciled in another country if the investment fund meets the IOSCO criteria? Free trade agreements, tax treaties, on-line interactive communications facilities and other kinds of technology make ideas such as these more feasible to pursue.

If this type of international investment were to be permitted, it would be necessary to consider what changes (in addition to changes in securities laws) would be necessary in other applicable laws to provide adequate remedies in the event of the need to pursue such remedies.

While the idea of a supra-national securities regulatory regime has appeal, it is probably premature to concentrate resources on trying to establish such a regulator. Instead, it might be more productive and effective to concentrate efforts on extending international regulatory cooperative efforts beyond compliance and enforcement activities to include developing protocols for mutual reliance by regulators in one country on work done by regulators in another country.

<sup>&</sup>lt;sup>63</sup> An important step forward was taken by IOSCO at its 23<sup>rd</sup> Annual Conference held September 12 - 18, 1998. At this Conference, IOSCO adopted its "Objectives and Principles of Securities Regulation", Disclosure Standards to Facilitate Cross-Border Offerings and Listings of Multinational Issuers, and authorized the publication of a report on securities activities on the Internet. The IOSCO Press Communiqué about these "milestones" is posted on the IOSCO website at <a href="http://www.iosco.org">http://www.iosco.org</a>.

The development of international mutual reliance, compliance and enforcement protocols, combined with establishing joint regulatory coordination groups as advocated by the Tri-Partite Group of securities, banking and insurance regulators (of which IOSCO is a member), and supporting the work of IOSCO should go a long way to creating international standards and to meeting the needs of the marketplace.

Complementary activities would include entering into reciprocity agreements to take regulatory action or enforce judgments in the "home office" jurisdiction for breaches of applicable laws in other jurisdictions. This is an extremely complex area that various IOSCO Working Groups have been working on for some time.

It is often easier to agree on the adoption of uniform standards than it is to agree on ceding jurisdiction. Therefore, common standards of universal application should go a long way towards facilitating cross-border activities and minimizing the need for regulatory constraints. The importance of creating international standards cannot be overestimated.<sup>64</sup>

#### 13. SYSTEMIC CHANGES IN THE REGULATORY SYSTEM

# 13.1. Systemic Changes

This Section discusses the need for systemic changes in the securities regulatory system. Two fundamental changes are discussed. These changes are aimed at putting all investors on an equal footing and improving the quality and timeliness of the information that is in the marketplace.

# 13.2. Consideration of an Issuer-Based Integrated Disclosure Model

The most fundamental issue that needs to be addressed as a result of the changes that have occurred in the financial services industry is the need to consider whether there is a more effective system for securities regulation than the current "closed system" that is in place in various countries including Canada and the United States. <sup>65</sup>

When one considers the strategic forces discussed in Section 5 in combination with the increasing knowledge and awareness of individuals and the resulting trend towards disintermediation, the time seems to have arrived to consider replacing the "closed system" of securities regulation<sup>66</sup> with an issuer-based disclosure model. This model would focus on ensuring that at the time of entry into the marketplace, full, true and plain disclosure of the material facts respecting an issuer and its securities is made and that this information is kept current through continuous disclosure requirements designed for this purpose. This model would result in an integrated disclosure system, the feasibility of which has now been made possible by the technology and communications facilities that exist today.

.

<sup>&</sup>lt;sup>64</sup> See Footnote 63.

<sup>&</sup>lt;sup>65</sup> The closed system focuses on the primary issue of specifically identified securities and it distinguishes its requirements on the basis of who the purchaser of the securities is and, in some cases, who the issuer is by creating classes of exempt purchasers and of exempt securities. It is a very legalistic, technical structure that is fraught with compliance pitfalls and does not sufficiently recognize the fact that securities of the same class or series of a class are fungible and that most trading activity takes place in the secondary market.

<sup>&</sup>lt;sup>66</sup> Securities regulation is based on two fundamental requirements. These are: (i) requirements to disclose information about the securities and the issuers thereof, and (ii) requirements that the persons who deal in securities or offer advice about securities be registered.

It is important to note that the emphasis in this proposal is on disclosure about the "issuer" rather than on disclosure about the issuance of specific securities. The information about the issuance of specific securities by the issuer would be included as part of the timely disclosure information in the marketplace about the issuer.

For example, new issues would simply be preceded by a timely disclosure notice with an appropriate delay period to allow dissemination of the information into the marketplace and an opportunity for securities regulators to require further disclosure (or whatever) if there is concern that the integrated disclosure forming the information record of the issuer is deficient.<sup>67</sup>

This shift in emphasis should simplify the ability to raise capital at a reasonable cost while enhancing investor protection. In the case of investment funds, where basic information about the nature and structure of the investment funds changes infrequently, it should eliminate the need to file a prospectus each year once the initial base disclosure document has been filed. This would represent an enormous cost saving for consumer/investors. Section 18 of this Review includes complementary suggestions for enhancing disclosure about investment funds for the benefit of consumer/investors.

The shift in emphasis contemplated by the proposal for an issuer-based integrated disclosure system parallels the shift in the nature of the relationship between intermediaries<sup>69</sup> and their clients. This relationship is now focused on advice-giving (with the transactions to implement the advice being incidental) as opposed to being transaction-based (with the advice-giving being incidental to the transaction). Section 16 of this Review (which deals with registration matters) discusses some of the aspects of this shift in the nature of the relationship between intermediaries and their clients and includes some specific recommendations that should be of benefit to consumer/investors.

The shift to an issuer-based integrated disclosure system should allow securities regulators to devote more attention to the *quality* of the information that is in the marketplace on an ongoing basis. This is consistent with the selective review procedures of the CSA which are premised on less time being spent on reviewing primary distribution issues and more time being spent on the review of continuous disclosure documentation. The shift should also free up the substantial time and resources of securities regulators that are devoted to dealing with applications for exemptive relief from the prospectus and registration requirements of applicable securities legislation as there should be less need for such exemptions resulting from a less technical system of securities regulation.

The shift should also enable securities regulators to focus more attention on registration-related issues such as the suitability for registration and for continued registration of those who are authorized to deal with the public. Section 16 of this Review (which deals with registration matters) includes suggestions for initiatives that should be of benefit to consumer/investors.

<sup>&</sup>lt;sup>67</sup> The delay period that is contemplated in this example would not be longer than the current delay period in, for example, the prompt offering system.

<sup>&</sup>lt;sup>68</sup> Changes in operations and activities would continue to be reported in the periodic reports. Material change reports would continue to be required to be filed or delivered to investors and regulators. The maintenance of the current prospectus disclosure system for mutual funds (including the proposals of the CSA referred to in subsection 3.5 to revise the disclosure system) is increasingly difficult to justify because of: (i) its costs, (ii) the fact that under the selective review procedures implemented by securities regulators the prospectus may not even be reviewed by the securities regulators with which it is filed, and (iii) the fact that (according to industry participants) the prospectus is not read by investors and often may not even be given to them.

<sup>&</sup>lt;sup>69</sup> I include the investment fund itself in the meaning of the word "intermediary".

A key component to moving towards an issuer-based integrated disclosure model is to ensure that there are appropriate statutory civil remedies for misrepresentations.<sup>70</sup> A companion requirement is to provide for international reciprocity in respect of the enforcement of judgments.

# 13.3. Timely Disclosure - Level Playing Field

The suggestions for an issuer-based integrated disclosure model are based on the belief that information is a major equalizing force. Access to it in an open and timely manner is a crucial ingredient to leveling the playing field. Information is a valuable commodity. Systems that are designed or are permitted to operate in a way that withhold the timely dissemination of material information result in fundamental unfairness to the marketplace as a whole.

With today's technology and communications systems that provide ready access to all, there is no reason why individual investors should be put at an informational disadvantage to institutional investors in terms of access to the information. There is no longer any need to rely on intermediaries such as investment dealers and advisers to disseminate material information to the marketplace or to structure the regulatory system so that they have first access to it. This information can and should be directly and publicly available:

- (i) through the Internet (or another universally accessible communications facility);
- (ii) from issuers and regulators at no cost; and
- (iii) obtainable through intermediaries.

Interesting developments in the marketplace are occurring that make it practical for the first time in history for the regulatory system to provide for equality of access to information and there is increasing pressure from vocal individual investors to be allowed such access.

Examples of what is happening in this area include the opening of "analyst calls" to individual investors who are able to listen in on issuers' calls with research analysts and key institutional investors on both a real-time and a deferred-time basis.

Another example is the ability to attend shareholder meetings via Internet access on both a real-time and deferred time basis. An example of this is the Annual Shareholders Meeting of BCE Inc. where for the last two years shareholders who were unable to be physically present at the meeting were able to "be present" through the use of Internet facilities that enabled them both to hear and see what was transpiring at the annual meeting as it occurred. Shareholders who were not able to "attend" in real time were able to access the proceedings on a deferred-time basis. It is not unreasonable to expect that one day facilities of this nature will be extended to allow "distant" shareholders to ask questions at the meeting in the same manner as those who are physically present at the meeting are able to.

The extension of this type of facility to enable consumer/investors to "attend" informational and other types of meetings in respect of the various types of investments fund participations that they own would be a significant initiative to provide consumer/investors with the tools they need to make decisions that will help them achieve their objectives.

-

<sup>&</sup>lt;sup>70</sup> Proposals for a Statutory Civil Remedy for Investors in the Secondary Market were published for comment in the OSC Bulletin dated May 29, 1998 at (1998) 21 OSCB 3335. The text of the proposed legislation appears in the OSC Bulletin at (1998) 21 OSCB 3367.

Yet another example is the growing practice of issuers to give advance notice of when earnings or other announcements are to be made and to provide telephone or Internet access to information facilities or services that enable the public to hear, first-hand and on a real-time basis, the information that is being released.

The application of improvements in technology and communications facilities brings an entirely different dimension to the ability to make timely, meaningful and relevant disclosure to investors of the material facts (and of the changes in the material facts) that they need in order to make reasoned investment decisions both initially and on a continuing basis.

Despite this all of our regulatory efforts at reform in this area are still focused on the "paper world" with arguments over whether one should have to use "plain language" and what information should be permitted to be kept hidden in regulatory or issuer files rather than provided to investors.

It is submitted that we need to use the tools that are now readily available to bring about, or at least to increase the likelihood of bringing about, effective disclosure. I use the term "effective disclosure" as describing timely, meaningful and relevant information that has been communicated to investors and is understood by them, or at least is capable of being understood by investors who are reasonable persons acting reasonably. Section 17 of this Review includes some specific suggestions designed to achieve effective disclosure.

It is acknowledged that there will always be situations of unequal knowledge and of informational asymmetry but at least the facts would be available for all to have access to equally.

Whether people choose to use the information and what their individual capabilities are to evaluate the information they have will depend on the respective abilities of the individuals involved. The important thing is that the choice of whether to use the information or to get help from an intermediary if help is needed to understand it, is theirs to make. Section 14 of this Review includes suggestions for enhancing the knowledge and awareness of consumer/investors with a view to reducing the knowledge gap between those who are aware and those who are not. Section 15 of this Review includes suggestions for enhancing the knowledge, awareness and proficiency of the financial intermediaries who deal with the public which should benefit the consumers/investors to whom they provide investment advisory and financial planning services.

#### 14. KNOWLEDGE AND AWARENESS OF CONSUMER/INVESTORS

# 14.1. What is Needed to Make a Difference

Two areas that I was asked to address in this Review relate to:

- 1. What facilitating role governments could play to bring about changes that would increase individual consumer/investor sovereignty and control over their assets?
- 2. What would enable consumer/investors to better identify their goals, the means of achieving them and to assess risk?

The simple answer to both questions is "education".

While I have some hesitation in focusing on "education" as a solution to the above questions because of the tendency to look to education as the universal remedy for all sorts of problems, the reality is that education is often the most effective solution. This is so because education reduces the knowledge gap and it empowers people to take charge of their own well-being.

It is particularly significant that consumer/investors have themselves identified education as a means to enhance their abilities to plan and manage their financial resources and to improve their personal economic circumstances. This illustrates a receptivity to the solution that should make its implementation particularly effective.

Examples of this receptivity are found in research conducted in January, 1996 for the Canadian Securities Institute regarding investor attitudes to learning. This research indicates that 63% of Canadians *strongly* agree (and 27% somewhat agree) that they need to become better at investing if they are to have enough money to retire comfortably. 54% of Canadians *strongly* agree (and 34% somewhat agree) that learning about investing should be their top priority.<sup>71</sup>

The November 1997 Yankelovich Survey<sup>72</sup> also found that investors feel there is a lot they have to learn about making good investment decisions. Investors want to learn more about investing and they feel that the securities industry should do more to educate investors. According to the Yankelovich Survey, only 4% of investors think they know "everything they need to" in order to make good investment decisions while 59% feel they know "just some" or "very few of the things necessary" to make good decisions. 81% of investors believe the securities industry should be doing more to educate the public about good investing.

According to the Survey of Canadians' Economic and Financial Understanding prepared in December 1997 by Gregg, Kelly, Sullivan & Woolstencroft: The Strategic Counsel Inc. for the Canadian Bankers Association and the Canadian Foundation for Economic Education, Canadians recognize the importance of an understanding of the economy and the relationship of such an understanding to their financial prospects and personal economic circumstances. More than two-thirds of the population believe that a better understanding of the economy and how it functions would enhance their abilities to plan and manage their financial resources and improve their personal economic circumstances. However, the authors of the Study conclude that Canadians exhibit little depth of understanding about the economy, economic indicators and their influence on their personal financial and economic circumstances. They report that fewer than one-in-five Canadians can be categorized as having "some" knowledge about the economy, with more than eight-in-ten emerging as having "only some" or "extremely limited" knowledge. Their conclusion is that Canadians need more information and that the self-described levels of understanding and confidence overstate actual understanding and confidence.

\_

<sup>&</sup>lt;sup>71</sup> The Canadian Securities Institute is the educational arm of the IDA. The attitudes reflected in this research study are said to be held by a broad cross-section of people - from those under 30 years of age through 60 plus, from people whose households make less than \$30,000 to those who make more than \$100,000, from those who have less than high school education to those who are university graduates. This research study is based on an Angus Reid Group's Omnibus Survey that was conducted in January 1996 by telephone among a representative cross-section of 1,500 Canadian adults. The results are considered reliable +/- 2.5 percentage points 19 times out of 20. The Canadian Securities Institute used the information derived from this Research Study in making its decision to launch the Investor Learning Centre in February 1996.

<sup>&</sup>lt;sup>72</sup> See Footnote 40.

<sup>&</sup>lt;sup>73</sup> This survey of 1,500 Canadians was conducted on a nation-wide, proportionate basis between November 3 and November 10, 1997 among adult Canadians, 18 years of age and older. The margin of error for a sample of this size is +/-2.6 percentage points, nineteen times out of twenty. In response to the needs identified in this survey, the Canadian Bankers Association in April 1998 launched its program called "Building a Better Understanding", the purpose of which is to provide useful financial information that is relevant and meaningful to the lives of Canadians. Ten (out of a planned series of 14) easy-to-read reference booklets are available in French and English on its website at <a href="http://www.cba.ca">http://www.cba.ca</a> or by calling 1-800-263-0231. These booklets are also available in alternative formats (Braille, Large Print, Compact Disc and Audio Tape). The Canadian Bankers Association advises that over 80,000 booklets have been ordered since the Building a Better Understanding Program was launched. Their website has received over one million hits and their call center has handled over 22,000 calls.

The information-needs index derived from the abovementioned Study reveals that not only are confidence levels related to a range of financial activities somewhat low (with only up to four-in-ten Canadians describing themselves as very confident about their competence and abilities in managing a range of financial activities), but informational needs are high in virtually all areas of financial management among **both** those who describe themselves as "at least somewhat confident" and those who report "a lack of confidence". The conclusions of the Study also state that:

"[e]fforts aimed at building a better understanding of both "big picture" economic indicators and their influence on understanding both the economy and personal financial management combined with a direction to sources of information that will enhance confidence in personal financial management will be extremely useful to Canadians. The data ... all points to a clear need within the population for more information that will improve Canadians' knowledge and provide a clearer context for personal decision-making." (Emphasis added)

Another study conducted for the Canadian Securities Institute in November of 1997 confirms that investors want to understand the implications of national and international events for the stock market. This finding corresponds with the findings in the abovementioned Survey that was conducted for the Canadian Bankers Association and the Canadian Foundation for Economic Education.

It would therefore appear from the foregoing that one does not have to overcome the hurdle of persuading people that what they need is education. One simply has to find an effective and efficient way to provide the appropriate education.

# 14.2. Filling the Information Gap - Youth

As noted in Section 11, many people are inadequately equipped with the basic life skills they need to function in the world in which they find themselves. There is no need for this situation to continue. There is evidence (as noted in subsection 14.1) that people do not want this situation to continue. There is a need to put in place measures to ensure that current and future generations are equipped with the basic life skills to function in today's world. This is an area where coordinated government action could marshal the forces necessary to bring about the needed changes.

The goal of our education system needs to be to create a lifelong learning process that will be a tool that people can continuously use to build on their knowledge and their ability to apply it to make decisions where they feel competent and confident.

The success of this learning initiative is dependent upon the following:

- parents clearly seeing the value and the need for addressing the skills needed for economic life in our school system;
- ministries and departments of education recognizing and assigning the priority that this area of education deserves;
- teachers recognizing the value and contribution that this area of instruction can bring to the lives and futures of their students; and

-

<sup>&</sup>lt;sup>74</sup> See the observations made by Gary Rabbior, the President of the Canadian Foundation for Economic Education, regarding this matter. His observations are set out in Appendix B to this Review.

effective leadership and coordination to chart the course for a better future.

Educators have a crucial role to play in this learning initiative. They bear the first line of responsibility for ensuring that educational programs (starting in kindergarten) equip individuals with the basic life skills they need for informed decision-making that will enable them to determine their own economic well-being including the ability to understand information that is communicated to them. Without this fundamental ability to understand information that is communicated to them, none of the other consumer protection remedies (particularly those that are based on disclosure) will work effectively.

# Identification of Needed Competencies

I believe that the starting point is to identify the skills or, in other words, the competencies that people need in the knowledge-based world in which we live. Competency is a term that is used to refer to the skills that are required to put knowledge to work to achieve a result. Competency is the catalyst that is needed to make effective use of information.

I believe that we need to bring together a group of key people (including people from the education field, from business and industry, from government and from the Canadian public) to identify the key competencies that are needed and the indicators that will measure them. Once this is done, it should be a relatively simple task to design, or to modify and/or supplement existing, education programs to ensure that the people who emerge from our schools do so with the necessary set of life skills that equip them for decision-making in the world in which they live.

# Raising Literacy and Numeracy Skills

One of the first areas that will need to be addressed is the issue of raising literacy and numeracy skills. Without remedial work in this area it will be difficult for both youth and adults alike to learn and to function effectively.

Appendix B sets out some of the significant conclusions from the IALS Report regarding the linkage between lifelong learning skills, economic success and the ability to capitalize on and adjust to structural change.

# **Empowering Teachers**

A teacher's confidence and knowledge in a subject area is one of the biggest factors that affects whether something will be taught and how well it will be taught.

The matter of preparing teachers to help prepare students for economic life is one that needs to be addressed right at the beginning of the learning initiative. I am advised that there is currently no program in Canada that prepare teachers in this respect.

### Teaching Methodology

A related area is that of the teaching methodology that is used. I believe that there is a need to shift the focus from teaching to learning. *There is a need to concentrate on helping people to develop lifelong learning skills by helping them learn how to learn.* These are the skills that are needed in the workplace today and it is what our children will need when they enter it.<sup>75</sup>. To do this, there is a need

\_

<sup>&</sup>lt;sup>75</sup> See Appendix B - IALS Report.

to shift the focus from what is described as "didactic information dissemination" to "discovery learning". Discovery learning is a term that is used to describe the ability to apply principles to problem-solving.<sup>76</sup>

# **Provision of Resources**

We need to concentrate on obtaining the necessary resources to support curriculum opportunities and teacher capability. This is discussed in Appendix B to this Review.

# **Curriculum Design**

Curriculum design is crucial to the success of any initiative aimed at better equipping people to deal with the requirements of today's workplace and the related decision-making that is required to function in today's world. Appendix B to this Review includes some suggestions that were made by, and resulted from discussions with, industry participants with expertise in curriculum design. The key principles underlying these suggestions are:

- the need to integrate outcomes related to economic life and understanding into other courses of study;<sup>77</sup>
- the need to make reliable, knowledgeable, non-partisan support and assistance available to provincial government departments and ministries;
- the need for curriculum, starting at an early age, to provide sequential development opportunities over the years of schooling so that concepts and fundamentals that are introduced in the early grades can be expanded, applied and integrated into a broader range of courses and incorporated into day-to-day life activities and decisions;
- the need to support teachers by providing, as part of their basic or continuing teacher training programs, development and training in economic fundamentals so that they feel confident with the curriculum proposals in this area.

The importance of the linkage between the opportunity for instruction (curriculum design), the ability to provide instruction (training and resources) and the learning outcomes cannot be overemphasized.

# 14.3. Filling the Information Gap - Adults

#### Identification of Needs

As noted in subsections 14.1 and 14.2, consumer/investors have themselves identified the problems they have and what they need to fix it. They have recognized that their needs are integrated and that their lack of awareness and their lack of ability (or the lack of confidence in their ability) to make the basic economic decisions of life that they are called upon to make on a daily basis are serious impediments to their achievement of their objectives of lifetime financial self-sufficiency for themselves and their family. Many people want to do something about this. This is an evolving process and not everyone is at the same stage in the process.

<sup>&</sup>lt;sup>76</sup> This teaching methodology is often not encountered until graduate school and sometimes not even then.

<sup>&</sup>lt;sup>77</sup> Mathematics and numeracy skills are an example. Addition, subtraction, multiplication, division, fractions, percentages, formulas, relationships, probabilities and so on are all integral parts of the world of money and finance and personal financial planning.

## Seeking Help - Advisers

In the first stage (once they decide that they need to do something) people usually turn to an adviser. They often just want "someone to tell them what to do" thereby either abdicating or at least sharing the responsibility for the choices that are made and for whether these choices reflect their goals.

Sometimes they have not even identified their goals or prioritized them. Sometimes people take a perverse sense of pride in removing themselves from the decision-making, saying things like "I don't read prospectuses - that's someone else's job - I rely on them - I can't even balance my cheque book." It is often not long before these people become anxious about whether they have done the right thing particularly if they see a higher rate of return posted somewhere. They bounce from product to product and sometimes from adviser to adviser, usually at a substantial financial cost to them.

# Seeking Help - Seminars

Sometimes people attend an investment or retirement planning seminar thinking that they are going to learn what they need to know to make informed decisions. They do not recognize that many of these seminars are just opportunities for selling products or for making contacts to sell products, either directly or via follow-up contacts. Sometimes they think that because the seminar is held in a school or workplace there is reason to believe that the seminar carries with it an endorsement of the merits of the seminar and that there is no product-category bias, product-bias or firm-category bias. Sometimes they are persuaded by the speakers at these seminars that they need to follow very aggressive investment strategies that expose them to risks that they often do not understand and that put their life savings (including their homes) in jeopardy.

This situation is very unsatisfactory. It is unsatisfactory for the consumer/investors who find themselves in it. It is unsatisfactory for most of the advisers who are trying to deal responsibly with those who consult them. I believe that from both perspectives the solution will be found through education.

# Seeking Help - Enhancing Knowledge and Awareness

Fortunately this is an area where industry participants, regulators and consumer/investors respectively see a need for investor education. Indeed, investor education - i. e. increasing people's knowledge and awareness - is one of the core recommendations of the 1995 Report and of this Review.

However, before a lot more money is spent on investor education, I think we need to stand back and focus on what we mean by this term. Is "investor education" only an euphemism for selling more product in the guise of advice, the cost of which is added to the product? I do not think so. Is "investor education" only about providing consumer/investors with information about firms, products and services? I do not think so.

The experience of the last four years has broadened the focus of what is needed in the area of investor education. This is seen in the survey results that are referred to in subsection 14.1. The results show that people have recognized that they need more than just "investor education" or "information". They show that people have recognized that they do not really know what to do with the information that is conveyed to them and that they have an insufficient understanding about the implications of much of the information that is provided to them. Most of them feel overwhelmed with the information that is given to them or do not understand how to use it or its relevance to them or to what they are doing. Investor education must therefore address these needs.

### Challenges Faced - Taking Charge

Most people recognize that they have to be more proactive in managing their affairs but they feel ill-prepared to do so whether directly or through an intermediary adviser. They do not know how (or lack confidence in their ability) to go about systematically setting their objectives and implementing a plan to achieve their objectives. They do not understand (or lack confidence in their ability to understand) the relationship of return to risk. They do not know how (or lack confidence in their ability) to monitor their performance effectively against their objectives. They do not know how (or lack confidence in their ability) to adjust their activities effectively to meet their objectives better. Sometimes this is due to a lack of specific knowledge. Sometimes it is due to a lack of understanding of how to apply the knowledge that they have to solving their real-life problems and to testing the effectiveness of their solutions.

# Challenges Faced - Changing Workplace

Most people also know that their workplace is changing. They are faced with the need to learn new skills in order to remain employed. Often they find it difficult to adapt to the changing workplace culture which expects people to be part of a "learning organization" and to engage in lifelong learning. They have not had to learn anything since they left school. Helping people learn how to learn is a key factor in sustaining employment and productivity. Substantial resources are invested in training and education programs in order to ensure an ever-growing set of competencies and skills for employees. Coincidentally, one of the areas that is receiving increased attention in the workplace is the need for employee education in financial planning matters as new compensation and benefit plans are introduced that shift the benefits, as well as the risks, to employees.

# **Providing Supplementary Learning Opportunities**

There is a convergence of all of these disparate factors - indeed, they are different sides of the same coin - that points to the need to provide Canadians with supplementary learning opportunities that are structured to enable Canadians, competently and with confidence, to make the basic economic life skill decisions that they are called upon to make in their daily life. Decisions about investing are only one aspect of these economic life skill decisions.

The simplest way to provide Canadians with supplementary learning opportunities would be to extend the process outlined in subsection 14.2 in relation to the youth learning programs to the people who no longer "go to school". Most adults need, or could benefit from, the same type of learning program that is recommended in subsection 14.2 for their children. These basic life skills programs would show people how to be lifelong learners, a characteristic that is increasingly essential in the knowledge-based world in which we live and work. <sup>79</sup> It would enable people to learn what they need to know and how to apply it to achieve their very fundamental and practical objectives of financial self-sufficiency.

There is a natural synergy that flows from the suggestion to extend the process outlined in subsection 14.2 beyond the primary and secondary grade schools. For example:

<sup>&</sup>lt;sup>78</sup> There is little point in providing people with a graph or a chart if they do not know how to read it or if they do not know the difference between a linear or arithmetic chart and a logarithmic or percentage chart and how a linear chart can be misleading by creating a false visual impression of the information it purports to show. There is little benefit in providing people with benchmarks if they do not know what a benchmark is or if they do not understand its relevance to their own decision-making.

<sup>&</sup>lt;sup>79</sup> See Appendix B to this Review - The Role of Literacy in Economic Life.

- 1. The curriculum for the adult learning program would basically be the same as for the youth learning program although the time that it would take to move through the adult learning program would be substantially shortened and would be adapted to the appropriate levels of existing knowledge for the adults taking the courses.<sup>80</sup>
- 2. The training programs that are developed for teachers of the youth learning program could be used to train additional people who could conduct the adult learning programs whether in the schools, in the workplace or in other locations.
- 3. The existing infrastructure of the schools could be used thereby enabling the delivery of the program in every community across the country.
- 4. The existing infrastructure of the workplace could be used to offer the adult learning program.
- 5. The existing investor learning center facilities could be used to offer the adult learning programs.
- 6. Distance learning techniques which use videos, video conferencing, telephone conferencing, CD-ROMS, computer programs, audio tapes and other means of delivery could be used to make the adult learning programs available to those who prefer self-directed learning either alone or as supplementary to attendance in person.<sup>81</sup>
- 7. National television and radio and national or local newspapers and magazines could be used to offer adult learning programs to those who prefer to learn in their own homes or to supplement adult learning programs offered in schools or elsewhere.<sup>82</sup>
- 8. The quality of the adult learning program could be maintained.
- 9. People attending the adult learning programs would have confidence in the independence of the information and the problem-solving techniques. In other words, concerns about product-category bias, product bias and firm-product bias could be eliminated.
- 10. A body made up of the key people referred to in subsection 15.8 could be constituted. This non-partisan body of key people (including people from the education field, from business and industry, from government and the regulatory world and from the Canadian public) would:
  - (i) periodically review the curriculum for the adult and youth learning programs to ensure its continued relevancy and suitability;

<sup>80</sup> An example of what I have in mind is language training programs. There are different programs for people who are "beginners" and for those who have varying degrees of familiarity with the language.

55

<sup>&</sup>lt;sup>81</sup> These techniques also lend themselves to adaptation for use in conjunction with the youth learning program. An interesting extension of distance learning programs is seen in a recent announcement by Tutornet.com (<a href="http://www.tutuornet.com">http://www.tutuornet.com</a>) of the start of a service using accredited teachers to provide assistance to students. This service offers scheduled live tutoring help in mathematics and science for a monthly subscription fee of US\$59 for unlimited access. Scheduled tutorials are held both in the afternoon and the evening. The curricula are derived from the Fairfax County, Va. School System Programs.

<sup>&</sup>lt;sup>82</sup> This coverage can also be adapted to target different segments of the audience including youth. It can also be used effectively to heighten people's awareness of the need to learn how to learn and of the ease of doing so. A model that I have in mind is the "Participation" advertising program which identifies very serious health problems that virtually everyone has or will have but offers simple, low or no-cost solutions like walking.

- (ii) offer recommendations respecting the curriculum;
- (iii) review specific programs for suitability; and
- (iv) set standards to which anyone presenting an adult or youth learning program would be required to adhere.

The suitability review referred to in items 10 (i) and (iii) above would include a review of the conformance to these standards.<sup>83</sup> The suggestion in item 10 (iv) above should add greatly to the credibility of the programs offered and should help to overcome the distrust that many consumer/investors have of industry-organized programs and should also help them to overcome their reluctance to attend industry-organized programs.

The body referred to in item 10 could also serve as (or oversee the operations of) a "better business bureau" that could be consulted by consumer/investors who have had a negative experience with an adult learning program or who want to check on the accreditation of the program, including the accreditation of the person(s) presenting or teaching the program.

Additional benefits flow from the foregoing. These include the following:

- 1. Financial advisers would benefit from a better informed client. One of the outcomes of both the youth and adult learning programs would undoubtedly be the recognition of the need to start planning and investing at a very early stage and how to select and work with a financial adviser. This recognition would likely increase the client base of financial advisers.
- 2. The need for enhanced "employee education" in the workplace about the choices employees are faced with in connection with compensation and retirement benefit plans could be substantially met or supplemented by the adult learning programs. The adult learning programs should enable people to make better decisions in these areas.
- Increased productivity of workers should result from people having learned how to learn and how to build on their learning and knowledge base and to apply it to a broad cross-section of matters both to solve problems and to innovate.
- 4. Canadians should be better placed to compete in the rapidly changing, knowledge-based global environment that exists today.
- 5. As consumer/investors, the ability to look after themselves and to use the information that is provided to them to make investment and other decisions should be greatly enhanced.
- 6. These outcomes should enable governmental and regulatory resources to be refocused into other areas. For example, a more knowledgeable and aware consumer/investor base should facilitate the implementation of a broader range of governmental initiatives in the area of pension and other retirement benefit programs including the increased privatization thereof.

\_

<sup>&</sup>lt;sup>83</sup> The suggestions in item 10 would help address the concerns that exist about investment seminars that were discussed in the 1995 Report and that were a major contributing factor to the substantial losses that investors incurred in various matters that have been the subject of hearings before the OSC. The two most recent egregious proceedings relate to the activities of Bruce Warrington (reported at (1997) 20 OSCB 1454) and Dino DeLellis (reported at (1998) 3 OSCB 305). Also see the decision of Cavarzan, J. in the action brought in the Ontario Court of Justice (General Division) by Druiven et al against Bruce Warrington, Associated Financial Planners Limited and Moneysem Financial Consultants Inc. [1998] O.J. No. 647 (Gen. Div.) (QL).

Even if the youth learning program were not to be extended to an adult learning program, adults would benefit from the youth learning program in that parents who help their children to learn are likely to learn themselves as a result of working with problem-solving, application-related learning techniques. However, parents would learn more effectively if there were some supplemental training to help them do so. Also, as parents with school-aged children are not the only segment of the adult population who are lacking in basic life skills, there is a need to extend the learning opportunity beyond this group.

#### Fundamental Tool

This type of learning opportunity is the most fundamental tool to make available to consumer/investors. With this tool consumer/investors will be able to understand and to apply information to enable them to make decisions competently and with confidence.

What better objective could there be?

# 14.4. Elements of a Successful Adult Learning Program

### Teaching Methodology

A lot has been written about adult learning programs. The best/preferred practices in youth learning methodology and the best/preferred practices in adult learning methodology seem to be fusing. This perhaps is due to the shift in emphasis from "didactic information dissemination" to "discovery learning" i.e. the ability to apply principles to problem-solving.<sup>84</sup>

One of the key differences between adult learning programs and youth learning programs is that often adult learners are more focused and motivated having seen the relevance and the benefits of knowledge and information. This challenges the program to be relevant and effective since adults usually will not suffer in silence. Adults are very critical or complimentary of programs based upon the utility of the program to their needs.

### Factors to be Considered in Adult Learning Program Design

During the course of talking with industry participants and educators about investor education programs for adults several observations were made that are relevant to the design of these programs. These observations are also relevant to the design of learning programs for financial advisers and planners. Most importantly they are relevant to the disclosure documents that are provided to investors and to registration requirements.<sup>85</sup> The observations are summarized in Appendix B to this Review.

# 14.5. Specific Learning Tools for Consumer/Investors

During the course of talking with industry participants and educators about investor education programs several suggestions were made about specific tools and initiatives that could help consumer/investors achieve their objectives. These suggestions are summarized in Appendix B to this Review.

-

<sup>&</sup>lt;sup>84</sup> It may also be due to the fact that for this generation, both adults and youth have received a similar style of education which differs from the style that prevailed for the Depression/World War II parents and their baby-boom children.

<sup>&</sup>lt;sup>85</sup> The disclosure documents include the prospectus, annual and interim reports, confirmations and account statements. The registration requirements relate to know-your-client/suitability requirements, education and proficiency requirements, confirmations and account statements.

# 14.6. Consumer/Investor Education Strategy

The recommendations for consumer/investor education include both short-term and long-term strategies. The proposals in subsection 14.2 relating to youth learning programs are examples of a long-term strategy. The proposals in subsection 14.3 relating to adult learning programs are a blend of both short-term and long-term strategies. What is important to recognize is that:

- the process is a continuing one of lifelong learning;
- this concept of continuous learning needs to underlie all learning programs; and
- great care needs to be taken to ensure that the person who is learning has a positive learning experience.

Industry participants and educators suggested some principles to ensure the success of any consumer/investor education strategy. These are summarized in Appendix B to this Review.

### 15. FINANCIAL PLANNING AND ADVICE

### 15.1. Education and Proficiency Standards for Industry Participants

Section 14 of this Review discusses the need for enhanced education measures for consumer/investors and suggests initiatives in this respect. Section 15 of this Review deals with the corresponding need for enhanced education and proficiency standards for industry participants and suggests initiatives in this respect. These initiatives center around the creation of a Non-Partisan Standards Council which is more particularly described in subsection 15.10 of this Review. This Non-Partisan Standards Council would form a core component of the self-regulatory structure proposed in Section 16 but would not itself be a self-regulatory organization.

One of the most essential components of consumer protection relates to the adequacy of the education and training that the intermediaries who deal with the consumer/investor must have. It is of the utmost importance to the consumer/investor that the matter of identifying the requirements for this education and training be proceeded with without delay. These requirements need to be based on what the intermediary is *actually* doing. In order to identify what the education and training requirements should be, it is necessary to start by first identifying what the intermediaries are actually doing, and then to identify the competencies and skills that they need to do this.

# 15.2. Beyond Product Sales

In order to make this identification, it is important to recognize the substantial changes that have occurred in the distribution side of the financial services industry. The focus of all participants in the financial services industry is firmly on asset gathering, asset allocation services and asset management. The strategic shift from a transactions-based business to a fee-based, relationship-driven, financial services business by virtually all participants in the financial services industry is well underway.

The trends that were outlined in Section 2.02 of the 1995 Report have come about. The asset management services offered by the various categories of dealers compete directly with the

\_

<sup>&</sup>lt;sup>86</sup> See Section 5 of this Review.

comparable services offered by the independent mutual fund organizations, financial institutions and investment counsel firms. The "wrap account" has become the dealers' entry mode into the world of portfolio management. The result has enabled the dealers to capture all, or an increasing portion, of the income stream that flows from portfolio management. For the most part this has been done without the dealers having to give up any part of the income stream that they obtain from their distribution services. 88

Function, product and advice have blurred and fused. With this fusion has come a corresponding shift in emphasis from the success of the transaction to the success of the portfolio. This has resulted in the need for the traditional trading-oriented sales representative to evolve into, or to be replaced with, a relationship-oriented manager whose role is focused on acting as a conduit for internal or external money managers.

A major factor that has brought about these changes is the changing needs of the consumer/investor.<sup>89</sup> Consumer/investors have recognized the need to adopt sound personal financial management practices. A large number of consumer/investors have recognized that because of their lack of personal financial expertise (or because of their lack of confidence in their expertise), they will need to turn to someone for advice.

This has created vast business opportunities for those in the financial services sector to convert personal financial management know-how into an economic good that adds value to the customer and for which there is a market. This conversion has heightened the need for increasing the knowledge and awareness of industry participants who seek to provide this personal financial management know-how. If industry participants do not increase their knowledge, awareness and ability to meet the needs of their clients, consumer/investors will likely turn elsewhere for the help that they will need.

I believe that the future for industry participants who seek to provide personal financial management know-how to their clients lies in positioning themselves to benefit from a more educated consumer/investor who is ready to make independent decisions, who will feel increasingly competent and confident to do so and whose ability to do so will be greatly enhanced by the ready availability of, and access to, low-cost technological help and service providers.

As noted in Section 2.02 of the 1995 Report, the changes that are occurring,

"will require significant investments in education, systems, training and back office operations and the modification of compensation structures to focus their base on assets under management rather than exclusively on commissions generated on

<sup>&</sup>lt;sup>87</sup> The expression "wrap account" is used as being descriptive of and as a part of a generic class of fee-based money management products which include in-house broker-managed discretionary accounts, traditional multi-adviser/menudriven wrap accounts, trust company individually managed accounts (IMAs), some investment counsellor segregated or pooled managed accounts, wrap mutual funds, broker-owned/managed funds, asset allocation products and traditional mutual funds.

<sup>&</sup>lt;sup>88</sup> The lack of awareness of consumer/investors to the fees and charges that they are paying and to the impact that these fees and charges have on their total return and on their ability to meet their goal of lifetime financial self-sufficiency is a major contributing factor to this. The recommendations in Sections 9, 13, 14, 17 and 18 should result in a more knowledgeable consumer/investor. However, whether consumer/investors will use this knowledge and awareness to produce better value for themselves remains to be seen.

<sup>89</sup> See Sections 9 and 14 of this Review.

specific transactions as has been the traditional basis. From the regulatory point of view, it is my opinion that it will be necessary to ensure that the regulatory system is designed to reflect the increased requirements in respect of educational requirements, training, systems and back office operations to support the shift to an advisory relationship."

So far, there has been little evidence of fundamental change in any of the areas referred to in the above quotation.

Currently, the recognition of the need to be able to offer a broad range of advice and services, with the ability to utilize a broad variety of products and services to implement a plan, is being addressed by firms and their sales representatives obtaining dual or multi-licenses. In addition, an increasing number of sales representatives who are licensed only to sell "single products" like mutual funds or life insurance are taking the necessary courses to be licensed to sell a full range of securities products plus the necessary courses that will give them a financial planning designation. Some industry participants (such as financial institutions) are encouraging their employees (or making it mandatory) that they take courses that will provide them with the necessary knowledge base to facilitate the provision of a full range of financial planning and investment advisory services to customers/clients. Some professional firms are establishing investment advisory firms that are licensed to provide investment advisory services. Various courses have been supplemented by adding an advisory component but they all start from the firm/product oriented base of the particular sectors of the financial services industry.

However, there is a need:

- (i) to take a fresh look at these courses in the light of the changed environment;
- (ii) to move beyond the firm/product oriented base of the particular sectors of the financial services industry and the courses offered by these sectors; and
- (iii) to make sure that they are designed to produce the needed competencies and skills to reflect the broad range of advice and service that is now being offered.

This need applies both to entry-level courses and to the specialized courses that lead to a financial planning designation.

### 15.3. Growing Awareness of Need for Change

There are signs of a growing awareness that the continuation of the status quo is not sustainable. These signs (some of which have been discussed earlier) are seen in the concerns listed below that were expressed to me by industry participants.

# Level Playing Field Concerns

- 1. The lack of a level playing field with some products, services and industry participants being regulated under securities legislation, some products. services and industry participants being exempt from compliance with the requirements of securities legislation, and some products, services and industry participants being subject to less onerous requirements under other legislation or to no requirements, while nevertheless competing at the same time with each other for the same consumer/investor dollar.
- 2. The lack of awareness on the part of consumer/investors that redress mechanisms vary depending on who the consumer/investor deals with and the type of product or service that is acquired.

- 3. The lack of awareness of consumer/investors to product-category bias, product bias and/or firm-category bias depending on which industry participant is consulted.
- 4. The different rules of fair practice and business conduct standards being applicable to industry participants and to the products and services that they offer.
- 5. The number of regulatory bodies and self-regulatory bodies that an industry participant must deal with or be licensed or registered with or become a member of.
- 6. The uneven standards of regulation, supervisory oversight and enforcement.

# Competency and Standards Concerns

7. Concern about the sufficiency of the educational and proficiency standards for entry level courses in the securities industry and life insurance industry and the lack of entry level courses in other sectors of the financial services industry. These concerns center on the fact that most of these course were designed for a transaction-related sales activity rather than for the provision of the ongoing financial planning and investment advisory services on a continuing basis that industry participants now hold themselves out as providing.

# Financial Planning Concerns

- 8. Concern about the unregulated use of terms such as "financial planner", "financial consultant", "financial adviser", "investment adviser" and other terms designed to imply expertise in financial planning and investment advisory services. These concerns are combined with a concern about the lack of standards to be met by firms and individuals before being able to offer, or to identify themselves as offering, financial planning and investment advisory services.
- 9. Concern about the sufficiency of the educational and proficiency standards for advanced courses leading to a "financial planer" designation.
- 10. Concern about the unregulated practice of financial planning and investment advisory services by professionals such as lawyers and accountants.

In summary, most of these concerns fall into two categories. These are:

- concerns about standards and competency (of which financial planning concerns are a subset);
   and
- (ii) concerns about the adequacy of the current regulatory and supervisory system,

to meet in each case the needs of consumer/investors and of industry participants.

### 15.4. The Reasonable Expectations of Consumer/Investors

Consumer/investors expect (and should be entitled to expect) that when they turn to a financial planner/investment adviser:

- (i) the advice they get will be based on a full client-needs assessment;
- (ii) the financial planner/investment adviser will have the capability to make that assessment;

- (iii) the financial planner/investment adviser will be able to (and will) provide them with advice that is in their best interests;
- (iv) this advice will be based upon a full range of money management services and products being available to implement the plan; and
- (v) the ability to carry out transactions to implement the plan and to provide ongoing monitoring and reporting services is and should be ancillary to the foregoing.

If this is not what the "service offer" to the client is, then there should be unequivocal, up-front, full disclosure to the client so that there is no question about the limited service offer that is being provided.

What these reasonable expectations add up to is the need for a fresh approach to educational courses and proficiency requirements and to registration requirements.<sup>90</sup>

# 15.5. Recognition of the Economic Interests in the Status Quo

The environment has not been conducive to taking a fresh approach to educational courses and proficiency requirements despite the need to do so. While each of the course providers has made substantial investments in, and improvements to, their courses during the last four years, the separate courses still center, for the most part, around the product categories that were reflective of the era before financial deregulation. During this period, each of the "four pillars" established its own educational and proficiency courses. With the exception of the trust industry, "these product-oriented courses still continue to be offered by the respective educational arms of the former "four pillars".

These separate courses are a major source of revenue for the industry associations that sponsor them. Accordingly, sustaining these courses and the need for them is of major economic significance to the industry associations. Unfortunately, the underlying desire to protect this revenue source has been the predominating, unspoken factor in ongoing discussions about the need to adopt enhanced education and proficiency standards as well as a contributing factor to the relatively sudden formation of the Financial Planners Standards Council of Canada ("FPSCC") and to the subsequent negotiations that took place to settle the disputes over the right to use the CFP (Certified Financial Planner) trademark in Canada.

It is important to keep these basic economic interests in mind when considering whether the FPSCC as it is currently constituted adequately addresses the need for a recognized financial planning standard for Canada.

Before discussing the FPSCC's role in establishing a recognized financial planning standard for Canada, it is important to identify what we are seeking when we talk about a "recognized financial planning standard for Canada" and how to go about obtaining it.

<sup>&</sup>lt;sup>90</sup> Elements of the suggested fresh approach are set out in the balance of this Section 15 and in Sections 16 and 17 of this Review.

<sup>&</sup>lt;sup>91</sup> The courses offered by the Trust Companies Association are merged with those offered by the Institute of Canadian Bankers.

# 15.6. Confusion of Issues

It is very easy when talking about "financial planning" (or any variation of that term) and about "financial planners" (or any variation of that term) to confuse the issues. This is particularly so when talking about what the education and proficiency requirements relating to this subject should be. The reason for the confusion is because there are several aspects to the issue of education and proficiency requirements. These aspects relate to:

- What is the activity that the education and proficiency requirements are designed to equip people for?
- Are we talking about being a "generalist" or a "specialist"?
- Are we talking about entry-level education and proficiency requirements or about advancedlevel education and proficiency requirements?
- Are we talking about entry-level education and proficiency requirements for a "generalist" or about advanced-level proficiency requirements for a "generalist"?
- Are we talking about entry-level education and proficiency requirements for a "specialist" or about advanced-level proficiency requirements for a "specialist"?

One could endlessly debate these questions and their permutations and combinations. However, given the evolution of the marketplace that has taken place, I think that the "activity" for which we need to be designing education and proficiency requirements is the "activity" of being able to meet the reasonable expectations of consumer/investors that are outlined in subsection 15.4 of this Review.

Because this "service offer" was not the focus when the current education and proficiency requirements were developed, the existing courses need to be reviewed and, where necessary, redesigned. Some work in this respect is currently under way.

I think that given the "service offer" that the public expects, the entry-level courses need to be designed to equip the industry participants who take them with the ability to provide the financial planning and investment advisory services and the services related thereto which are referred to in subsection 15.4 of this Review.

This will probably mean that the graduates of these courses will be "generalists" in financial planning and investment advisory services. As outlined in the observations referred to in subsection 15.9 of this Review, the "generalist" courses should probably be tiered. A suitable descriptive term should be found to describe this entry-level "generalist" state and should be required to be used uniformly throughout the industry to avoid confusing consumer/investors.

If this recommended shift in educational and proficiency requirements is made, I do not think there is a need for separate stream of training for a "specialist". It seems to me that the model used in the medical profession which builds specialized training and practice requirements on top of general training and practice requirements should be sufficient and should avoid confusing consumer/investors. Again, a suitable descriptive term should be required to be used uniformly throughout the industry to describe the areas or levels of specialization.

-

<sup>&</sup>lt;sup>92</sup> See Appendix C to this Review.

What I am suggesting is another example of addressing the need to integrate, simplify and streamline regulatory requirements to serve the needs of the marketplace and of consumer/investors better.

### 15.7. Additional Confusion

The controversy surrounding the establishment and the ongoing operations of the FPSCC has been another factor that has tended to confuse, and perhaps inadvertently to obscure, the basic issues about financial planning. This controversy culminated in the Canadian Securities Institute and the Institute of Canadian Bankers withdrawing from the FPSCC in the spring of 1998.

These withdrawals were perceived by some as not being in the public interest. The withdrawing organizations received a lot of criticism for their actions. They are being urged to reconsider the matter of their withdrawal. Most of the criticism has centered around the view that it would be desirable for there to be only one "financial planner designation" and for this designation to be the "CFP". There was a lot of talk about the undesirability of continuing to have an "alphabet soup" of designations.

With respect, I think the criticism misses the substantive issues that underlie the valid objectives reflected by this view. These objectives (with which I agree) are that:

- 1. There is a need to establish a recognized financial planning standard that everybody who calls themselves a financial planner (or uses any other term that implies financial planning and investment advisory expertise) must meet and comply with on a continuing basis.
- 2. It is desirable that there be a designation of the professional status of the individual as a financial planner/investment adviser that is clearly recognizable by the public.

The area of my disagreement relates to what the appropriate starting point is for establishing a recognized financial planning standard for Canada. The critics have started from the assumption that the requirements for obtaining and maintaining the CFP designation is the appropriate starting point to satisfy the abovementioned objectives. I think we need to examine whether their assumption is correct. To do so, we need to take a close look at whether in fact the FPSCC requirements do constitute the appropriate financial planning standard for Canada in light of today's needs. Legitimate questions have been raised about this matter by respected industry educators. These questions deserve to be addressed.

In doing so it needs to be kept in mind that the CFP standards were developed in the days prior to financial deregulation taking effect and prior to the industry shift from a transaction-based approach to a client-needs based approach. In addition, some of the courses that have been accredited by the FPSCC have an inherent product-category bias and were designed from the perspective of a transaction-based service offer rather than today's integrated client-needs, advisory-based service offer.

I think that before we endorse any standard, we need, as discussed in subsection 15.4 of this Review, to take a fresh approach to the requirements for educational courses and proficiency training and to adapt the requirements to meet the public's current expectations of financial intermediaries.

In the interim, a moratorium should be placed on the use of the term "financial planner" and like terminology which has become very misleading.<sup>93</sup> I recognize that it is easier to say this than to actually do it. Therefore, we need to expedite the basic work that needs to be done.

### 15.8. Establishing a Financial Planning Standard

I wish to emphasize that I am not saying that the CFP standard adopted by the FPSCC could not be found to be an acceptable standard for Canada or that it could not be adapted so that it would be. I am simply saying that until we go through a rigorous process to determine what an acceptable financial planning standard for Canada is, it is premature to be looking at the FPSCC standard and its proprietary CFP designation as being the exclusive financial planning designation for use in Canada.

It was a recommendation of the 1995 Report that a process of the nature that I am again suggesting here be engaged in. The organizations that have withdrawn from the FPSCC have also recommended that a process of this nature be engaged in. I believe that at least some of the other members of the FPSCC would also support such a process.

### Non Partisan Identification of Financial Planning Key Competencies

The starting point is to identify the skills or the competencies that people providing financial planning and investment advisory services to others need to have in order to meet the integrated needs of their clients.<sup>94</sup>

I think we need to put in place a process similar to the one I suggested in subsection 14.2 of this Review in connection with identifying the competencies that consumer/investors need to have to help them achieve their objectives and with respect to designing the curriculum for learning programs aimed at producing such competencies.

Here, I think we need to bring together a non-partisan group of key people<sup>95</sup> (including people from the education field, from business and industry, from governments and the regulatory world, and from the Canadian public) whose initial task would be to identify:

- (i) the key competencies that are needed by the client-contact professionals who will provide financial planning and investment advisory services to others; and
- (ii) the indicators that will measure such competencies.

Once this is done, it should be a relatively simple task to design, or to modify and/or supplement existing, education programs to ensure that the people who emerge from these programs do so with the necessary set of professional skills to equip them to provide financial planning and investment advisory services to others. The process of doing so will involve starting with the entry level programs and working through them and through the current "financial planning programs".

<sup>93</sup> The unregulated use of this terminology is very distressing to the industry participants who have earned their financial planning designations.

<sup>94</sup>As noted in subsection 14.2 under the subheading, *Identification of Needed Competencies*, competency is a term that is used to refer to the skills that are required to put knowledge to work to achieve a result. It is the catalyst that is needed to make effective use of information.

<sup>95</sup> This group could form the Non-Partisan Standards Council referred to in subsection 15.10 of this Review and undertake the additional activities described in subsection 15.10.

Again, this need to review the sufficiency of the entry-level courses was identified in the 1995 Report. The 1995 Report recommended that the current industry education and proficiency requirements be adapted to reflect the industry's change in approach from a transaction-based one (with the advice being ancillary to the transaction) to a relationship-based one focused on the provision of ongoing financial planning and investment advisory services (with the transactions being ancillary to the advice).

Some of the background work has been done or is in the process of being done. This work should provide a useful starting point and should make the task less daunting.

# 15.9. Observations Relating to Education and Proficiency Programs for Industry Participants

A lot of the same observations that are made in Section 14 of this Review in connection with requirements for effective consumer/investor education apply equally to the requirements for effectively equipping industry participants with the competencies and skills necessary to engage in financial planning and investment advisory services. Some additional observations have been made by industry participants and should be taken into consideration as well. These observations are listed in Appendix C to this Review together with some additional observations of my own.

#### 15.10. Need for a Non-Partisan Standards Council

There is a need for a non-partisan standards council ("Non-Partisan Standards Council") made up of a group of key people (including people from the education field, from business and industry, from governments and regulators and from the Canadian public) that will oversee the ongoing development of the financial planner/investment adviser standard that results from the process identified in subsection 15.8 of this Review and the modifications that are necessary to keep it current. The activities of the Non-Partisan Standards Council would include:

- (i) identifying the requisite competencies and skills;
- (ii) identifying the indicators of these competencies and skills:
- (iii) developing the prototype curriculum;
- (iv) accrediting the respective courses;
- (v) performing the same activities for the continuing education programs; and
- (vi) overseeing the examination process.

Other possible activities could include:

- (i) regulating the descriptive terms relating to the provision of financial planning and investment advisory services;
- (ii) identifying best practices;
- (iii) developing an industry-wide code of ethics and business conduct that persons who provide financial planning/investment advisory advice are required to adhere to; and
- (iv) identifying the circumstances that would lead to the withdrawal of the ability to use the designation and accreditation and overseeing the procedures relating to such withdrawal.

These last four activities would depend on what the status of the Non-Partisan Standards Council is and would depend on its relationship with the Single SRO described in Section 16.

### Status of the Non-Partisan Standards Council

I do not think that there is any need for the Non-Partisan Standards Council to be a separate self-regulatory organization for the reasons outlined in Section 16. I think Non-Partisan Standards Council should be independent of but linked to the Single SRO and thereby subject to regulatory oversight and supervision. In management terms, it would be an independent part of the Single SRO with a "direct report" to the Regulator. More thought needs to be given to how this would be made to work and what the Non-Partisan Standards Council's interaction with the Single SRO would be.

### Regulatory and Supervisory Role of Regulators

I think that the role of securities and other regulators should be confined to ensuring that there is a responsible non-partisan standard-setting process in place and working. The accrediting of courses relating to proficiency and educational requirements would become the function of the Non-Partisan Standards Council. The accreditation procedure would involve ongoing review by the Non-Partisan Standards Council to assure that the accredited courses continue to meet the standards and objectives of the Non-Partisan Standards Council.

# Common Standard but No Monopoly

There is no question that it would be desirable to have a common standard throughout Canada for financial planning and investment advisory practitioners. This can be achieved very simply and directly and without conferring a monopoly on what in effect is a commercial enterprise, by requiring that everyone who engages in financial planning and investment advisory services obtains the CFP designation. No other country has done this and there is no reason why Canada should when there are other means available to achieve the desired objectives.<sup>96</sup>

#### 15.11. The FPSCC and the CFP

It should be kept in mind that the "CFP" is only one of several financial planning designations that are used in various countries. There is fierce competition among the different designations in other countries as well as in Canada. Questions have been raised in other countries about the sufficiency of the CFP requirements as well as in Canada.

It should be kept in mind that the FPSCC simply imported the financial planning concept that was developed in the United States in 1985 by the Denver-based Certified Financial Planners Standards Board. The Denver organization registered the "CFP" designation as its trademark for its concept around the world. The Denver organization enters into franchise arrangements with organizations in other countries granting the right to use the CFP concept and trademark in such countries and it charges fees for the privilege of doing so including an annual licensing fee. The Denver organization provides an outline of 175 topics that the teaching programs should cover. This outline serves as the accreditation framework for the teaching programs outside the United States. The Denver organization also requires that the franchisee administer a certifying exam and that each of the CFP licensees adheres to a code of ethics as well as commits to thirty hours of continuing education per year.

\_

<sup>&</sup>lt;sup>96</sup> See subsections 15.8 and 15.10 of this Review.

The national franchisee in Canada is the FPSCC. The FPSCC is responsible for protecting against trademark infringements, collecting and remitting a portion of the licensing fees to the Denver parent group, accrediting teaching programs, developing and administering qualifying exams and monitoring compliance with the code of ethics and the continuing education prescriptions.

### Critics of the FPSCC Standard

Critics of the FPSCC standard observe that it is an imported standard - a standard that was developed for a system that is not evolving as our system is. It is a franchise. It doesn't take into account the state of evolution of our financial system or its evolution for the future. The critics observe that the US origin of the standards comes from the insurance industry and that the curriculum is designed to sell products - and not to advise. The curriculum is fixed. It cannot be changed. The curriculum simply specifies the topics to be covered rather than the topics with the indicated learning outcomes. The critics criticize the exam process saying that the whole exam is multiple choice. The exam reflects only knowledge-based memorization questions and does not test, or adequately test, the ability to apply the knowledge to solve the kinds of problems that financial planners and investment advisers are required to deal with in their day-to-day practice. The critics observe that the examination process that is used is efficient if you are just trying to "roll in the bucks for minimal costs". They say that the FPSCC is a marketing game rather than a standards council.

The critics of the FPSCC query how much work the FPSCC has done in trying to identify and design a financial planning/investment advisory standard based on meeting client needs. They allege that there is no transparency in the program accreditation procedures and that accreditation was basically granted in exchange for the agreement to become a member of the FPSCC.

Concern has also been expressed about the governance procedures of the FPSCC and its lack of resources to do an effective job. In addition, concern has been expressed about regulators using their powers to make a commercial interest viable by forcing industry participants to become licensees of the FPSCC. Concern has also been expressed about the substantial outflow of the licensing fees to the Denver organization.

# Consumer/Investors and Industry Are Not Well-Served by a Monopoly

These concerns and criticisms aside, neither consumer/investors nor industry participants are well-served by a monopoly being granted to any organization. There is a less intrusive way to arrive at the objectives that are outlined in subsection 15.7 of this Review which are to establish:

- (i) a publicly-recognized financial planning standard, and
- (ii) a publicly-recognized designation of the professional status of a financial planner/investment adviser.

Consumer/investors and industry participants deserve the independent, competency-focused perspective that a Non-Partisan Standards Council of the nature described in subsection 15.10 of this Review would bring to the issues. It is important that neither they nor the regulators be misled by the use of the word "Standards" in the name of the FPSCC and that people look behind the name to what it is in fact.

### 16. THE RIGHT TO DEAL WITH THE PUBLIC

# 16.1. Registration and Licensing Requirements

Registration is one of the two cornerstones on which our current securities regulatory and supervisory system is based. The term "registration" covers matters such as suitability for initial and ongoing registration, regulatory capital requirements, insurance and bonding requirements, education and proficiency requirements, know-your-client/suitability requirements, advertising requirements, fair dealing requirements, record-keeping requirements, financial reporting and audit requirements, requirements relating to internal systems and controls, customer protection fund participation, and various other requirements. These requirements are sometimes referred to as conditions of registration.

Persons and firms who deal in securities or who provide advice about dealing in securities are required to be registered under applicable securities legislation to do so. Persons who sell insurance are required to be licensed under applicable insurance legislation. The requirements for registration or licensing must be met initially and on an ongoing basis.

Section 16 of this Review deals with issues relating to registration or licensing.<sup>97</sup> It contains proposals for changes in the current multi-registration requirements to create an integrated registration system centered around advice-giving/distribution activities. This registration system is a core part of the integrated regulatory structure recommended in Sections 10 and 12 of this Review that could be achieved by agreement among all levels of government.

# 16.2. Simplification of Regulatory Requirements for Registration

The evolution of the marketplace which has resulted in function, product and advice blurring and fusing provides a unique opportunity to take a fresh approach to the requirements that must be met by all intermediaries who deal with the public.

Here again, requirements can be simplified, streamlined and coordinated to eliminate the need for multi-registrations, for multi-regulators and for multi-self-regulatory organizations while at the same time providing for adequate, uniform and functional oversight, supervision and regulation.<sup>98</sup>

This simplified approach to registration is consistent with, and is a fundamental part of, the recommendations that I have made in Sections 10 and 12 for an integrated regulatory and supervisory system to reflect the combination, within each of the "four pillars", of the respective separate functions to which each pillar formerly was restricted prior to the deregulation of the financial services industry.

Today's marketplace would be better served by a registration system that reflects the integration of function, product and advice that has occurred and that centers around advice-giving rather than product-licensing. The continued maintenance of the current registration requirements with separate regulatory bodies, centered around product distribution, is no longer necessary, appropriate or desirable.

The proposals in Sections 10 and 12 of this Review for centralized, coordinated, streamlined and functional regulation would include the elimination of the separate registration requirements

\_

<sup>&</sup>lt;sup>97</sup> The terms "registration" and "licensing" are often used interchangeably in the industry.

The burden of multi-registrations with conflicting and overlapping requirements cannot be overestimated. One organization advises that it is subject to regulation by more than 85 regulators in Canada. See Footnote 47.

with separate regulatory and/or self-regulatory bodies. The need for such separate registrations and memberships is becoming increasingly problematic for the industry and its regulators given the integration and consolidation that is occurring in the industry.<sup>99</sup> As noted in Section 10, further work needs to be done to develop the framework for the proposed integrated regulatory system that would focus on:

- (i) prudential and solvency concerns that relate to various activities (such as, for example, banking, insurance and liability trading) of the respective institutions; and
- (ii) the advice-giving/distribution activities of the intermediaries; specific product knowledge would be subsumed under the framework of the advice-giving/distribution activities.

The framework for the registration requirements relating to the advice-giving/distribution activities is outlined in this Section 16.

# 16.3. Registration Framework

The registration framework that I propose centers on two concepts:

- (i) requirements for a single registration with the Regulator; and
- (ii) requirements for a single membership in the Single SRO;.

The term "Regulator" refers to the integrated regulator contemplated by the proposals contained in Sections 10 and 12 of this Review.

The term "Single SRO" refers to the single self-regulatory organization (described in this subsection 16.3) that would be recognized by the Regulator for the purpose of mandatory membership.

Another example of the integration and consolidation that is occurring is seen in the announcement of the proposed acquisition by First International Asset Management Inc. of a 49% interest in Beutel Goodman and Co. Ltd. (See Susan Heinrich, The Financial Post, July 31, 1998, First International buys Beutel stake.) Michael Simonetta, chief operating officer of First International is quoted in the article as saying, "We've now accomplished [our] objective". The article indicates that this objective is to own a piece of a company that operates in each of the asset management areas and that he was referring to mutual funds, pension fund and high net-worth investment counselling. The article states that First International has also purchased a 100% interest in Groupe Financier Concorde Inc., of Quebec City and a 60% interest in Toronto-based financial planning franchisor, Money Concepts (Canada) Ltd.

<sup>&</sup>lt;sup>99</sup> One example of the integration and consolidation that is occurring on the distribution side of the industry is reflected in the reported entering into of a letter of intent respecting the acquisition by Equinox Financial Group Inc., a Toronto-based distributor of insurance and other financial products, of a controlling stake in Trillenium Investors Services Inc. (See Shirley Won, The Globe and Mail, July 30, 1998, *Equinox to take stake in Trillenium*.) In this article, the Chairman of Equinox Financial Group Inc., in explaining the reason for the acquisition is quoted as saying:

<sup>&</sup>quot;About half the 1,400 financial advisers associated with Equinox make mutual funds available to their clients, but they do so now through arrangements with other dealers. ... The Trillenium investment means that Equinox's complete network of advisers will be able to easily add mutual funds to their range of offerings. ... The mutual fund area is a very large, growing area. It's another product line that we must have available. Since we didn't have that access, they had to go out and make their own separate deals with another broker-dealer. ... There is an awful lot of consolidation going on in the insurance industry and among networks of mutual fund dealers."

The registration framework contemplates that all intermediaries who are involved in advice-giving and distribution activities would be required to be registered with the Regulator and to be a member of the Single SRO. No other registrations or memberships would be required.

### The Single SRO

The Single SRO would operate on a national basis. The membership requirements of the Single SRO for such matters as capital, insurance and bonding, education and proficiency, contingency fund participation, maintenance of books and records, supervisory procedures, and so on would be tiered. These requirements would be based on what the members of the Single SRO *actually* do. It should be possible to reduce the number of different categories of registration and membership without adversely affecting consumer protection.

In this respect there would be fundamental basic provisions that apply across the board relating to matters such as entry-level education and proficiency requirements, basic good faith requirements, rules of fair practice and business conduct, financial and other reporting requirements, advertising, client confirmations and reporting, audit requirements including surprise audits, compliance reviews, and enforcement proceedings.

The provisions for basic entry-level requirements, including those for education and proficiency, could take into account "equivalency" requirements that other self-regulatory professional bodies like The Law Society of Upper Canada and the CICA have. These equivalency requirements could satisfy some of the Single SRO's requirements although in some cases, such as in the case of insurance and bonding requirements, there would probably be a need to arrange for additional coverage.

The objective should be to make the requirements of the Single SRO and of the bodies regulating the practice of law and accounting complementary to each other rather than duplicative or conflicting.

The objective of the exercise would be to ensure that it does not make a difference to consumer/investors (at least from the perspective of adequate regulatory oversight and supervision) with whom they deal in terms of there being minimum requirements that must be adhered to or minimum redress remedies available in the event of problems.

Membership by intermediaries in other industry organizations would continue to be voluntary. The functions of these other organizations would be confined to those of an "industry" association.

However, the stock exchanges would continue to perform their market regulation role and organizations such as The Law Society of Upper Canada and the CICA would continue to perform their respective roles relating to the legal and accounting professions. Coordinating protocols would be entered into to provide for cooperation in the case of oversight and supervision activities, including compliance and disciplinary matters.

A key factor in bringing about the Single SRO would be to provide for an appropriate governance structure. Proposing a governance structure is beyond the scope of this Review. However, the experience of the last four years indicates that it is important - at least from a psychological perspective - to establish a new organization into which the self-regulatory role of the existing self-regulatory organizations could be rolled as opposed to requiring people who are non-members of an existing self-regulatory organization to join such organization.

The focus on financial planning and investment advisory functions of the intermediary as being a basic primary part of the services provided by the intermediary, with the range of products to implement the intermediary's services as being ancillary, enables a rationalization of the intermediation process that reflects what has occurred in the marketplace and enables the entire process to be integrated within the

ambit of a single strong effective self-regulatory organization that will operate on a national basis with all participants being a member of it.

It is important to recognize that a failure to rationalize the regulatory and self-regulatory system will involve real and immediate costs (both in the short-term and in the long-term) which ultimately will be paid for by consumer/investors. All businesses transfer their costs to their customers. If we continue to maintain separate "product silos" and to center regulation on "product knowledge" instead of centering it on advisory relationships that are based on the assessment of a client's integrated need, the impact on consumer/investors of advice remaining tied to product-bias will be negative both in terms of direct costs and in terms of lost opportunity costs. There is no reason to permit this to happen.

# 16.4. Alignment of the Compensation System

Hand-in-hand with rationalizing the regulatory and self-regulatory system to enable a better matching of client needs, is the need for the industry to redesign its reward systems. For the most part, the current systems reward product sales as opposed to rewarding the ability to meet the client's integrated needs. Part of the re-alignment of the reward system needs to focus on ways to encourage referrals when this is the appropriate thing to do from the client-needs perspective. Otherwise, the normal tendency of the intermediary will be to select whatever reflects the particular niche of the intermediary, including a tendency to select (or be tempted to select) the product that will offer the highest revenue stream to the intermediary.

In the current system, what the client ends up with depends to a large extent on who the client first approaches. This occurs even in the case of a fully-informed client who has a plan. Most people do not appreciate that the advice they get may be skewed by their entry point or by how the individual's compensation is structured within the firm or institution. Even when the reward system is based on the productivity of units within firms or institutions, there is still a bias to serve the client needs through the particular entry-point unit.

When you have a system that requires all intermediaries to be registered, it should be easier to develop a reward system that encourages referrals when that is the appropriate thing to do from the client-needs perspective. The parameters for the reward system need to be developed and these parameters need to include ensuring that there is clear disclosure to the client of the revenue stream that will flow to or for the benefit of the referring firm and representative.

The reward system is a systemic problem that needs to be addressed by the industry. If it is not addressed, a major credibility gap between the industry and its clients is likely to occur. This credibility gap is likely to increase as clients become increasingly knowledgeable and aware.

### 16.5. Problems with the Current Registration System

There are some serious problems with the way the current registration system is operating.<sup>100</sup> Despite the fact that registration of the persons and firms who deal with the public is one of the two cornerstones<sup>101</sup> on which the securities regulatory system is structured, very limited regulatory

<sup>&</sup>lt;sup>100</sup> The comments in this Section relate only to the registration system under the Securities Act (Ontario). While I have been told that there are similar problems under other legislation of the Province of Ontario such as insurance legislation and under securities and other legislation of the other Provinces, I have not been able in the time constraints applicable to this Review to follow-up on these comments to gain an understanding of whether there are problems and, if there are, the magnitude of any such problems.

<sup>&</sup>lt;sup>101</sup> The other cornerstone is disclosure.

resources have been directed to registration. One consequence of this has been that delays in registration have plagued the industry and its regulators for years. The problems have been exacerbated as the number of applications have increased.

Because of the legitimate concern about the impact of the delay in processing applications for registration on people's livelihoods, the pressure to streamline the processing of applications for registration has increased. One of the means of dealing with this pressure has been to adopt, as an administrative practice, the practice of registering virtually everyone who applies for registration unless someone comes forward and demonstrates beyond a reasonable doubt that the applicant is not suitable for registration or continued registration. This administrative practice has now become the prevailing regulatory perspective.

This regulatory perspective to registration regards an applicant as being entitled as of right to be registered and to remain registered under the Securities Act (Ontario) unless and until someone can demonstrate that the registrant is not suitable or that the registration is objectionable. This approach to registration changes the current statutory onus under the Securities Act (Ontario) which requires regulators, before granting registration, to determine that the applicant is suitable for registration and that the proposed registration or amendment to registration is not objectionable.

This current approach to registration is problematic from the perspective of consumer/investors and the industry. It is problematic because it subsumes the interests consumer/investors (who are encouraged by the industry and the regulatory structure to place their trust and reliance in a registrant to act in their best interests) to those (albeit a minority) who would abuse such trust and reliance. It subsumes the interests of the vast majority of honorable firms and individuals who merit the trust and reliance of consumer/investors to those who tarnish the industry through their abuse of trust.

There are current proposals to amend the Securities Act (Ontario) to reverse the statutory onus that is contained in Section 26 of the Act to determine the suitability of an applicant for registration and that the registration is not objectionable *before* granting registration.<sup>102</sup> These proposals raise substantial public interest concerns. The fact that the proposed amendments may be in line with current administrative practices or with the legislation in other Provinces does not address the concerns that exist.

There is a need to identify better ways than we currently have to determine suitability for registration and continued suitability for registration. This should be the focus of ongoing work.<sup>103</sup>

Failure to address the issue of suitability for registration helps no-one. When problems arise, it brings the whole regulatory system and the whole industry into disrepute despite the fact that the problems center around a relatively small number of people and firms. However, for the consumer/investors who are affected by the actions of these firms and people, this is no comfort. Their problems are devastating.

<sup>&</sup>lt;sup>102</sup> See Section 17 of An Act to amend the Securities Act - Draft. Notice of the proposed amendments to the Securities Act (Ontario) was published in the August 14, 1998 OSC Bulletin at (1998) 21 OSCB 5078. The text of the proposed amendments appears at (1998) 21 OSCB 5149.

<sup>&</sup>lt;sup>103</sup> One suggestion that was made is to increase the onus on the sponsoring firm to determine suitability for registration, to monitor suitability on an ongoing basis, and to be accountable if it turns out that the person in question is not suitable.

### 16.6. Specific Concerns

Several concerns that center around registration matters and the related administrative procedures have been expressed by industry participants and consumer/investors. These concerns all give rise to questions about how effective the registration provisions are to provide for the protection of consumer/investors. The concerns which have been expressed by industry participants and/or by consumer/investors include the following:

# Suitability for Registration

- 1. Concern (as noted above) has been expressed that issues relating to the suitability of the person or firm to deal with the public get caught up in regulatory expediency. The view has been expressed that the registration focus is on administrative processing<sup>104</sup> rather than on dealing with the substantive issues of determining the suitability of the firm or person for registration to deal with the public and determining that such registration is not objectionable.
- 2. Concern has been expressed that the criteria for determining suitability for registration and for determining that the registration is not objectionable are not evident. In this respect, industry participants and members of the public respectively question how people can be registered and re-registered in the face of situations that most people regard as making an applicant unsuitable for registration. These situations include:
  - (i) a history of financial difficulties including bankruptcy;
  - (ii) prior regulatory or professional disciplinary proceedings that have resulted in a reprimand or suspension or revocation of registration;
  - (iii) pending regulatory or professional disciplinary proceedings;
  - (iv) prior or current proceedings involving fraud or breach of fiduciary obligations:
  - (v) misappropriation of assets or other wrongdoing relating to the handling of money, the trading in securities and the provision of advice;
  - (vi) mental illness; and
  - (vii) substance abuse.

### **Termination for Cause**

3. Concern has been expressed about the speed with which people who have been terminated for cause by one registrant are re-registered with another registrant without there first having been an investigation or hearing to determine the suitability for re-registration. The concern is that this procedure, which permits ongoing registration while an investigation is conducted, presents a risk to the public. There is evidence from reported cases and disciplinary proceedings that the regulatory rationale for this procedure (which is based upon the legitimate objective of regulators not wanting to get into the middle of a commercial dispute) ignores the risk to the public and exposes the public to risk. This regulatory rationale of not wanting to take sides in what regulators regard as a commercial dispute, combined with the regulatory preoccupation of

<sup>&</sup>lt;sup>104</sup> See also the concerns noted in this subsection 16.6 under the sub-heading "Administrative Processing".

not wanting to deprive people from earning their livelihood in the securities industry, often has the unintended result of facilitating further wrongdoing by the registrant in question.

- 4. Concern has been expressed about the seeming lack of understanding at the regulatory level of just how bad **some** situations are even when clearly documented problems are brought to their attention. The tragedy of a failure to take prompt action is that members of the public are often hurt by what the registrant does in the interim. Further hardship results from the fact that by the time regulatory action is taken, the registrant often has either absconded with investors' assets or is in financial difficulty.
- 5. Concern has been expressed about the lack of support provided by regulatory authorities to dealers who terminate a sales representative for cause and report the matter to the regulatory authorities. Responsible registrants find themselves having to justify their actions to the regulatory authorities who question whether the registrant did in fact have cause to terminate and who then expedite the transfer of the terminated sales representative's license to another registrant without first investigating the situation to ascertain the facts of the matter. This often leaves the registrant who is responsible for the representative's wrongful acts that were committed while in the employ of the registrant in an extremely difficult position. The registrant may find that it is denied access to the necessary books, records and premises to verify client positions and the magnitude of problems. Often the representative that is terminated for cause institutes legal proceedings against the registrant hoping that a settlement can be negotiated that will let the representative off the hook. In one situation (which was described as being the worst the registrant had ever seen) the fact of re-licensing by the securities authorities was argued as proof that there was no cause for termination and that the representative had been wrongfully dismissed.

The combination of the regulatory attitude and the likelihood of ensuing legal actions as described above results in a reluctance by registrants to terminate a "bad rep" for cause in order to avoid having to report the circumstances and be caught up in a process fraught with problems. This result puts the public at risk and seriously undermines the effectiveness of registration as a key factor in consumer protection.

### Close or Onerous Supervision Orders

- 6. There is concern about the regulatory practice of using "close supervision" or "onerous supervision" terms and conditions and orders as a proxy for the required statutory determination that the registration in question is suitable and not objectionable. Experience has shown that:
  - (i) these terms and conditions and orders are not always complied with;
  - (ii) there is either no follow-up or monitoring of compliance with these terms and conditions and orders or there is inadequate follow-up or monitoring;
  - (iii) the public and other registrants are often not aware of the existence of these terms and conditions and orders; and
  - (iv) the public often suffer losses that are not recoverable as a result of transactions that were effected after (as well as before) re-registration.
- 7. Another concern expressed about the regulatory practice of using close supervision or onerous supervision terms and conditions and orders as a proxy for the required statutory determination that the registration in question is suitable and not objectionable is that courts sometimes view the fact of re-registration or continued registration as evidence that the regulatory authorities

have reviewed the substance of the matters and have determined that the matters complained of are not serious enough for the regulatory authorities to revoke the registration. Where the courts conclude, rightly or wrongly, that the regulatory authorities did not think there was anything wrong with what the registrant did, the courts tend not to sustain the consumer/investor's action for redress. From the perspective of the consumer/investor seeking redress for matters arising out of a registrant's wrongful acts, it would be better if the registrant's registration is suspended so that improper inferences are not drawn from the fact of continued registration. <sup>105</sup>

Experience has shown that there is a wide gap between the regulatory authorities' expectations of a registrant's obligations under a close or onerous supervision order and the reality of such close or onerous supervision. The misery that can occur as a result of a registrant's continued registration is extreme. The misery that can occur as a result of a registrant's continued registration is extreme.

### **Independent Contractors**

8. Concern has been expressed about the licensing of sales representatives who operate as independent contractors without adequate provisions being made to ensure that their activities and those of their employees are subject to adequate oversight and supervision. There are a variety of issues that flow from this including the issue of the name in which the representatives carry on business. In some jurisdictions, the dealer name is not required to be used. In other jurisdictions, the dealer name is required to be used. I believe that at best the situation is confusing. At worst it is misleading. From the public's perspective, the situation is unsatisfactory. It is compounded by the fact that regulatory approach differs from province to province.

Other problems in connection with the licensing of sales representatives who operate as independent contractors without adequate oversight and supervision relate to the maintenance of trust accounts, record-keeping, access to books, records and premises, insurance and bonding and capital requirements.

# **Education and Proficiency Requirements**

9. Concern has been expressed about the tendency to grant exemptions from educational and proficiency requirements based on experience that is not truly a proxy for the educational and proficiency requirements. This occurs mainly when people seek to be recognized as an investment counsel in the category of a portfolio manager.

\_

<sup>&</sup>lt;sup>105</sup> The concern that has been raised illustrates how issues that get caught up in regulatory expediency can often backfire. A regulatory decision to issue an onerous supervision order instead of suspending registration often reflects a decision of expediency from the perspective of the regulatory authorities. The regulators know that the order will not be challenged by the registrant and they assume that the order will have the effect that they want - namely, that no-one in the industry will employ the person because the terms and conditions are so onerous. Accordingly, everything should be just fine but in actual fact the result often operates to the prejudice of the consumer/investor who is harmed by new or recurring wrongful acts.

<sup>&</sup>lt;sup>106</sup> See for example the Order and Settlement Agreement *In the Matter of the Securities Act R.S.O. 1990, c. S.5, As Amended and In the Matter of Dino P. DeLellis, William R. Kennedy and The Height of Excellence Financial Planning Group Inc.*, (1997) 20 OSCB 4327.

<sup>&</sup>lt;sup>107</sup> See for example the Decision and Reasons *In the Matter of the Securities Act R.S.O., Chapter S.5, As Amended and In the Matter of Dino P. DeLellis, William R. Kennedy and The Height of Excellence Financial Planning Group Inc.,* (1998) 21 OSCB 305.

10. Concern has been expressed about the adequacy of the educational and proficiency requirements to provide people with the skills and competencies that they require both at the entry-level and as a specialist, and on a continuing basis. These concerns and the recommendations for dealing with them are the subject of Section 15 of this Review. However, one concern that is not addressed in Section 15 is the concern about people who leave the industry for a period of time and then seek to return to the industry. Such people are currently permitted to do so within certain time frames. There are current proposals to reduce these time frames, to bring them into line with the less onerous requirements of the self-regulatory organizations.

I believe that in a rapidly changing environment, there is a need to ensure that after being away from practice for a period of time, people's skills are refreshed before they are again permitted to deal with the public.

I believe that it is particularly important to address this issue in view of the transition that has taken place from a sales transaction-focused business to a client-needs focused financial planning/investment advisory service business as the skills set and the issue of the currency of the skills set differ substantially from that which is required in a sales transaction-focused business. In addition, the constant introduction of new instruments adds to the need to ensure that people's skills are updated before they are again permitted to deal with the public.

### Inadequate Systems and Controls

11. Concern has been expressed about the fact that registration is granted without there being any assurance that the registrant has an adequate infrastructure in place to run the business and to provide the services that the registrant intends to provide. There are no requirements that the registrant have the necessary technology, systems, controls and procedures in place before commencing business. There is no requirement to ensure that operating and monitoring procedures sufficient to meet the registrant's obligations to the public and to the regulatory authorities are in place before commencing business. Experience has shown that sometimes there is a lack of understanding of what a "trust" account is and of how to ensure that trust accounts are properly established. This state of affairs seriously undermines the consumer protection measures that registration requirements are intended to provide.

#### Administrative Processing

12. There is a perception that the regulatory focus is on "pushing the paper", collecting the fees (which constitute a major expense for the industry and a major source of revenue for regulators) and reducing the complaints about the time it takes to process an application. People view with concern the system of administrative processing of registration applications that assigns applications to review officers based on alphabetical order according to the name of the applicant rather than grouping the applications according to the type of application in question. They view this mechanical approach as tending to produce less than an optimal assessment of the substantive matters that are involved in the applications. They also view with frustration the business process system inefficiencies that require the issuance of separate cheques in payment of multiple registration fees instead of being able to make an omnibus payment.

13. There are complaints about the lack of helpfulness on the part of regulators in terms of providing guidance to the registrant about how to comply with standards that are expected to be met.<sup>108</sup>

It is important to note that the foregoing matters are ones that industry participants and/or consumer/investors have raised.

# 16.7. Industry Standards for Intermediaries

The matters outlined in subsection 16.6 reinforce the fact that the current fragmented regulatory and self-regulatory system and, in particular, the provisions relating to registration requirements are not functioning as effectively as they should in order to deal with the needs of consumer/investors in the reality of today's marketplace. In fact, in some cases, the interests of consumer/investors are adversely affected

### Suggestions to Deal with Registration-Related Concerns

Here are some suggestions designed to improve the situation. These suggestions (which are intended to flesh out the proposals previously made for restructuring the regulatory and supervisory system) would produce the most effective results if they were to be made a part of such proposals. Even if the restructuring proposals were not to be implemented., the suggestions would also help improve the current system

The suggestions are intended only to be indications of what could be done and are not intended to be an exhaustive set of principles. More work is required to be done in this area. The suggestions reflect my own views and those of a cross-section of industry participants who are supportive of initiatives that would result in a simplified, streamlined, integrated regulatory and supervisory system that is not product-centered but enables a broad range of products to be made available to satisfy client-needs.

#### Structural

Establish the Single SRO described in subsection 16.3 of this Review.

### Initial and Ongoing Registration and Membership Standards

2. Adopt standards that all intermediaries are expected to meet as a condition of initial and ongoing registration and membership in the Single SRO. In addition to standards of good conduct and fair practice (which would include adherence to a rigorous code of ethics) these standards would include basic operating standards and procedures.

<sup>&</sup>lt;sup>108</sup> This concern may in part be related to, or be a consequence of, the concern which is noted in this subsection 16.6 under the subheading, "*Inadequate Systems and Controls*".

I note that in October of 1997, the OSC published two reports prepared by its Compliance Team. The purpose was to inform mutual fund dealers and their advisers of the types of deficiencies noted by the OSC in performing compliance examinations in the hope that the incidence of non-compliance would decline. The first report is entitled "Deficiencies Identified in OSC Compliance Examinations of Mutual Fund Dealers". The second report is entitled "Deficiencies in Regulatory Filings of Registrants Who Are Not Members of a Self Regulatory Organization". These reports are contained in the materials for "Dialogue with the OSC" - October 20, 1997.

- 3. Do a better job of screening intermediaries before they are accepted into the industry aimed at ensuring that the intermediaries will deal honorably with the public.<sup>109</sup> There should be a duty of care on member firms to screen, monitor and report on unsuitable intermediaries.<sup>110</sup>
- 4. View registration and membership in the SRO as a privilege and not as a right. Applicants should be required to establish their suitability for registration and membership. Perhaps a practical way of doing this would be to identify factors that will give rise to questions of suitability. Where these factors exist, applicants would have to establish why registration should be granted. The onus would be on the applicant to establish suitability in the face of the existence of factors that give rise to questions of suitability. A high level of "proof" would be required to satisfy this onus in the face of these factors.

Where there are no questions as to suitability and the applicant meets the other requirements for registration and membership in the Single SRO, the application could be granted.

Similar requirements would apply to transfer applications.

Where there is any question as to whether an applicant has been terminated for cause, no transfer should be processed until the matter has been fully investigated and the applicant has satisfied the onus of proving suitability for re-registration.

Appropriate procedures for dealing with these requests for transfer in a timely manner should be developed.

If close/onerous supervision orders are used,<sup>112</sup> they should be used only when the firm assuming the obligations in question has provided a secured guarantee or a performance bond adequate to cover the risks to the public of permitting the person who is subject to the close/onerous supervision order to continue to deal with the public. The funds derived from the secured guarantee or performance bond should be available to reimburse the public for losses incurred as a result of wrongful acts of the person who is subject to the close/onerous supervision order.

If close/onerous supervision orders are used, the terms and conditions of these orders should be made public and clients should be informed of their existence both at the time they are made and by noting the existence of the orders in confirmation forms and account statements.

<sup>110</sup> It was observed that the registration or licensing of intermediaries is the first line of consumer protection and that licensing involves screening. Monitoring the actions of intermediaries is the next line of defense. Disciplinary action is the last line of defense.

<sup>&</sup>lt;sup>109</sup> Many people expressed the view that the strength of the industry will be in the quality of the advisers and how well they are equipped to deal with the public. The recommendations here and in Section 15 of this Review are aimed at enhancing the quality of the advisers - the intermediaries.

Examples of circumstances that give rise to questions of suitability are set out in subsection 16.6 of this Review under the subheading, "Suitability for Registration".

<sup>&</sup>lt;sup>112</sup> A threshold question that needs to be addressed is whether close/onerous supervision orders should be used at all. There are serious questions about these orders. Firstly, if the situation merits a close/onerous supervision order being issued, there is reason to question whether the individual or firm should be dealing with the public at all. Secondly, as noted in subsection 16.6 of this Review under the subheading, "Close or Onerous Supervision Orders," there is reason to question not only whether these orders work at all but whether they result in unintended consequences. Perhaps consumer protection and industry integrity merit a zero tolerance attitude comparable to what exists in other industries (such as the airline industry) where public safety is at risk.

Appropriate procedures should be developed for dealing with frivolous and vexatious allegations.

5. Enhance the rules of fair conduct and business practices to bring them into line with the professional standards referred to below under the subheading *Educational and Proficiency Standards*.

### **Educational and Proficiency Standards**

- 6. Enhance the educational and proficiency requirements for intermediaries as described in Section 15 of this Review. Educational and proficiency requirements should:
  - (i) Incorporate values-based learning aimed at building commonly-shared high ethical standards that justify and sustain the consumer/investor's trust in and reliance on the integrity of the participants in the intermediation process.
  - (ii) Emphasize the professionalism that comes with providing financial planning and investment advisory services based on integrated client needs.
  - (iii) Focus on the need for high standards and obligations on the part of those who advise and implement their advice through the sale of product.
  - (iv) Focus on the duty of care and other fiduciary obligations.
  - (v) Address ethical issues including those relating to compensation matters.
  - (vi) Result in a high competency level that recognizes that the complexity of intermediaries' activities today demands a higher skill set than may have been required in the past.
  - (vii) Include basic business training on:
    - (a) how to manage a service business that is based on providing advice and implementing the advice; and
    - (b) how to conduct operations.

This training would be tiered to the activities that are to be carried on and the responsibilities that are to be assumed but with an overall understanding of the checks and balances that apply to the intermediary in question.

(viii) Include basic training in compliance procedures including practice and procedures aimed at managing the intermediary/client relationship.

Again, the training would be tiered to the activities that are to be carried on and the responsibilities to be assumed by the intermediary but with an overall understanding of the checks and balances that apply to the particular registrant with whom the intermediary is associated so that the intermediary can understand his, her or its role in the compliance procedures and in the management of the relationship with clients.

A model that offers helpful guidance in this respect is the new "practicePRO" program that the Lawyers Professional Indemnity Company has developed. 113

# **Basic Operating Standards and Procedures**

7. The operating standards and procedures should be aimed at ensuring that each intermediary has an infrastructure in place that is adequate to run the business and to provide the services that the intermediary intends to provide.

The operating standards would include requirements to have and to use the industry-prescribed standards for technology, systems, controls and procedures before being permitted to commence business.

The purpose of these requirements is to ensure that internal operating and monitoring procedures sufficient to meet the intermediary's obligations to the public and to the regulatory authorities are in place.

Consumer protection demands that such infrastructure be in place before intermediaries are permitted to deal with the public.

A related benefit would be the ability to speed up the processing of transactions between firms and to reduce the overall costs of doing so.

These operating standards should apply to third party "back-office" service providers as well as to internal "back offices". Routine and surprise compliance audits by the Single SRO should extend to third party "back-office service providers as well as to member firms.

"... we need to start addressing the root causes of claims. When we look at these root causes, we discover that many claims stem from fundamental errors made by lawyers - errors that can easily be corrected but do involve a change in behaviors on the part of the lawyer. practicePRO helps this change to happen. Our second reason revolves around the reality of change. Forces such as technology, a more demanding consumer client, globalization and others are reshaping the environment in which lawyers practice. To thrive, lawyers need to adapt; we need to see change as an opportunity not a threat. practicePRO is intended to help facilitate this change management process by, for example, helping lawyers integrate technology into their practice. So, in a nutshell, practice PRO is intended to help lawyers adapt and thrive."

The practicePRO program is initially focusing on managing the lawyer/client relationship and managing conflict of interest situations. In describing the former, the Summer 1998 issue of lpic news (Vol. 4, No. 2) at page three states,

"It's a fact that lawyers today practice in an increasingly complex world. The demands on members of the legal profession, and the related risk of a claim, have never been greater. And the costs associated with a claim are significant - both in financial terms through deductibles, increased insurance premiums and surcharges, and on the personal front in dealing with issues such as stress, professional stigma and lost opportunity to work on other matters. The best way to protect yourself against these realities is through proactive action to minimize and even avoid risk. And the best way to start is by managing the lawyer/client relationship effectively from start to end."

I note that these same comments could have been written about the investment fund industry and the provision of financial planning and investment advisory services to the public.

<sup>&</sup>lt;sup>113</sup> In describing the practicePRO initiative, the June 1998 special edition of lpic news stated that practicePRO is "... about being proficient, professional and progressive. That's what the PRO in practicePRO stands for." The article goes on to explain that the reason for the initiative is:

# **Compliance Programs**

8. Enhance the compliance programs both internally within firms and externally through third party and Single SRO reviews.

Third party "back office" service providers should also be required to enhance their compliance programs and operating procedures in order to provide their services to registrants and members of the Single SRO. The compliance programs and operating procedures of these third party "back office" service providers should be subject to review and approval by the Single SRO. 114

# Operation of Branch Offices and Sub-Offices

- 9. Adopt measures designed to ensure that issues related to the independent operation of branch offices and sub-offices including:
  - (i) the use by registered representatives of personal corporations and of separate corporations for branch offices or sub-offices;
  - (ii) the use of franchise arrangements; or
  - (iii) the use of any other arrangements of an "independent contractor" nature;

do not result in any lessening of consumer protection and do not relieve the dealer from liability for the conduct of its representatives. 115

Here it will be important to address system control/business process issues that relate to a variety of matters such as, for example, things as fundamental as client statement processing.

A lot of the administrative processing (such as the generation and review of client statements) that is required under IDA rules to be done by the head office of an IDA member firm as part of its required systems control and oversight procedures, is in the case of non IDA-member firms, permitted, and indeed often required, to be performed in the branch offices or sub-offices of non IDA-member firms. This is largely attributable to the high commission payouts that prevail in non IDA-member firms. The result of the high commission payouts is that the firms transfer the obligation to do the administrative processing to their representatives who operate in branch offices or in sub-offices.

<sup>&</sup>lt;sup>114</sup> These suggestions are based on the principle that the best way to deal with a problem is to prevent it from happening in the first place.

<sup>&</sup>lt;sup>115</sup> The CSA Committee on Distribution Structures is currently addressing these issues. In a speech made on July 6, 1998, Milestones and Challenges for Mutual Funds in the Coming Years, David A. Brown, the Chair of the OSC, said that it is expected that the CSA shortly will introduce for comment

<sup>&</sup>quot;...rules that will ensure that a mutual fund dealer will be responsible for the conduct of all registrants who are sponsored by it. Our objective will be to ensure that investors will have recourse against the dealer if its sales people fail to comply with proper standards. While we recognize the need for flexibility to permit tax planning and encourage entrepreneurial initiatives by sales forces, we will not permit the creation of a corporate structure that will shield the dealer from liability for the conduct of the sales people."

It will be important to broaden these initiatives to recognize the proposals for the integration of activities outlined in Sections 10, 12, 15 and 16 of this Review.

This raises issues as to:

- (i) the adequacy of the firm's oversight and supervision procedures;
- (ii) the adequacy of system controls; and
- (iii) whether there is a potential lessening of consumer protection.

Another area of system controls and business process matters that needs to be addressed relates to the liability of the dealer respecting signature guarantees given by "independent contractors" to make good on claims where the endorsement that has been guaranteed is forged and/or fraud is involved.

# Misleading Advertising

- 10. Adopt measures designed to:
  - (i) prevent misleading advertising;
  - (ii) pro-actively monitor compliance with the measures; and
  - (iii) take timely, effective disciplinary action if there is a failure to comply.

These measures aimed at curbing misleading advertising should extend both to actual statements that have been made that are misleading and to statements that have been omitted to be made where the omission makes what has been said misleading.<sup>116</sup>

### Misleading Names

11. Adopt or enhance measures designed to ensure that the names of firms are not misleading. There are several problems that relate to the names of firms.

The first problem relates to the names of firms generally. Some firms use words in their names that suggest that the firms have some official, non-profit, organizational status when they do not.<sup>117</sup>

The second problem relates to the use of virtually identical names by different corporations that are affiliated, where one firm is a member of the IDA or of a stock exchange and the other firms are not. It is often unclear to consumer/investors which firm is the one that they are dealing with and whether the transactions are covered by the CIPF.

A third problem relates to the use or non-use of the dealer's name on business cards by representatives of the dealer. This problem is related to the matters referred to in item 9 above under the heading "Operation of Branch Offices and Sub-Offices.

<sup>&</sup>lt;sup>116</sup> Zero commission advertisements are an example of misleading advertising.

<sup>&</sup>lt;sup>117</sup> Examples include the use of words like "Standards", "Council" and "of Canada" in the firm name.

### Complaints Mechanisms

12. Adopt or enhance measures requiring member firms and their representatives to keep a record of complaints that are made by clients and the resolution of the complaints. As a minimum, these records should be available for inspection. These measures should require that settlements with a client, disciplinary action and payments to investors should be required to be reported to the Single SRO and to the Regulator.

How complaints are dealt with and the matter of reporting them to the Single SRO, the Regulator and the public is one that needs to be addressed without delay. It is part of an "early warning" system for suitability for registration and continued registration. It is an essential element for consumer protection. The lack of an adequate complaints mechanism combined with lack of disclosure to the public about problem firms and/or representatives is another example of the knowledge gap that exists and that operates to the disadvantage of consumer/investors. 119

These matters are controversial. The fact that these matters are controversial should not be permitted to allow the present situation to continue where consumer/investors are not aware of very serious problems that exist with firms and/or their representatives.

# **Disciplinary Procedures**

- 13. Enhance disciplinary procedures to deal with breaches of the rules of the Single SRO and to improve the timelines for investigations and hearings. 120
- 14. Adopt a penalty structure designed to encourage adherence to the rules of the Single SRO.<sup>121</sup>

### 16.8. Compliance with Sales Practices Rule

The matter of monitoring compliance with, and taking appropriate proceedings for breaches of, the sales practices rule (NI 81-105) is a matter that needs to be dealt with promptly. There is concern that some people are not complying with the rule. This is putting the firms that are complying with the rule at a competitive disadvantage and there is increasing pressure on these firms to find a way around the sales code rule.

Examples of allegations of non-compliance that have been made include allegations that:

(i) although travel and accommodation expenses are being billed and cheques or VISA charge slips are being received "in payment", the cheques and VISA charge slips are not being processed for payment;

<sup>119</sup> This knowledge gap is part of the informational asymmetry referred to in subsection 9.2 of this Review.

<sup>&</sup>lt;sup>118</sup> See subsections 16.5 and 16.6 of this Review.

<sup>&</sup>lt;sup>120</sup> A suggestion that was made related to publicizing disciplinary proceedings. This has a dual advantage of making other industry participants aware of practices that they may not realize are wrong and of possibly acting as a deterrent to others who engage in such practices. While disciplinary proceedings are periodically referred to in Bulletins such as the OSC Bulletin, relatively few people have access to the publication. There is a need to develop better communication measures including the timely posting of information on websites and newsletters to member firms and registrants.

<sup>&</sup>lt;sup>121</sup> See subsection 16.09 of this Review.

- (ii) substantial gifts (like sails for a boat) are being made to representatives;
- (iii) despite the prohibition in the sales practices rule, some fund management organizations are continuing to deal directly with sales representatives in offering sales incentives;
- (iv) computer software is being "sold" by some fund management organizations to sales representatives but the purchase price is not collected;
- (v) the sales practices rule is being interpreted as not applying to the sale of segregated funds including those segregated funds that are publicly offered mutual funds with an insurance wrapper;
- (vi) the situation described in item (v) above is providing an incentive to mutual fund dealers (who are licensed to sell insurance products) and others to sell segregated funds instead of mutual funds in order to find their way around the sales practices and restrictions that the sales practices rule was designed to address.

#### 16.9. Penalty Provisions

Most people do not think that the consequences of breaching the rules of a self-regulatory organization or the provisions of applicable legislation are serious enough. They do not think that the consequences of breaching such rules operate either as a deterrent to doing so or as an incentive to ensure compliance by others with the rules. Furthermore, the consequences do not offer any recompense to people who have suffered loss as a result of the failure to comply with the rules.

Several suggestions and observations were made by industry participants about the penalty structure. These suggestions and observations include the following:

- 1. When a representative is found, after due process, to have breached the rules of the self-regulatory organization or the provisions of applicable legislation, then the firm with which the representative is associated should also be found to be in breach because it has breached its obligations to supervise and to exercise oversight over the activities of the representative.
- 2. Fines/penalties ought to reflect a disgorgement of profits. It was observed that despite the fines/penalties imposed by the regulators or by the self-regulatory organizations, the trades remain profitable to both the representative and the representative's firm. Therefore, the revenues generated as a result of the acts of the defaulting representative ought to be required to be paid over by way of fines/penalties or restitution.
- 3. It was observed that the problem with the way the current system is administered is that there is often no real penalty for wrongdoing and no real business cost to unethical representatives and the firms that employ them. It was also observed that it is important to create examples that are so punitive that people simply will not breach the rules or fail to observe the standards of good practice.
- 4. It was observed that when a representative or firm's registration is suspended, terminated, revoked or canceled, the trailing commissions (which are a form of asset-based sales

<sup>&</sup>lt;sup>122</sup> "Fines" are seen as simply part of the cost of doing business. Temporary suspensions are seen as simply a vacation.

commission) continue to be paid. If this is the case, 123 it raises questions about the issue of splitting commissions with non-registrants and questions as to whether the representative and the firm ought to continue to benefit from wrongful acts.

- 5. It was observed that when a representative's or firm's registration is suspended, terminated, revoked or canceled, registration or licensing under the Insurance Act continues. The representative and/or firm in question simply sells life insurance products and segregated funds or continues to conduct business through a spouse's or some other person's registration. It was suggested that there should be appropriate orders restraining these activities and action taken to enforce the orders.
- 6. It was observed that when a representative's registration is suspended, terminated, revoked or canceled, there is no order restraining the person's ability to act as a consultant or to teach sales methods to industry participants and that there should be appropriate orders restraining these activities.

These suggestions and observations indicate a need to do some fundamental rethinking about the consequences of wrongful acts.<sup>124</sup> Some of the issues raised indicate that the relevant rules of the Single SRO or of the regulatory authorities should include provisions that restrain registrants or member firms from retaining the services of a representative or firm whose registration has been suspended, terminated, revoked or canceled.

Again, the focus of the rules should be on consumer protection rather than on the right of the registrant to earn a living in the "securities" industry. These comments all presuppose that any decision to suspend, terminate, revoke or cancel registration will have been made after a hearing where the registrant has had an opportunity to be heard and which has been duly convened and conducted.

#### 16.10. Other Related Issues

There are some other matters that would enhance consumer protection that could also be dealt with as "registration" matters or "Single SRO" matters. However, I have chosen to address them in other parts of this Review. These matters include:

- (i) the use of enhanced "know-your-client/suitability" procedures;
- (ii) the use of enhanced confirmation/point-of-sale forms;
- (iii) the use of enhanced account statements;
- (iv) the use of nominee accounts, on-book/off-book transactions and powers of attorney;
- (v) performance reporting standards;
- (vi) custody of assets held in registered plans, wrap accounts, segregated funds, equity-linked universal life policies and other plans or products; and

<sup>&</sup>lt;sup>123</sup> One would assume that trailing commissions would cease to be paid when the account(s) in question are transferred to others. However, this assumption may not be correct.

<sup>&</sup>lt;sup>124</sup> Some suggestions are included in subsection 16.7 of this Review as well as in this subsection 16.9.

(vii) consumer redress mechanisms.

#### 17. MEETING CONSUMER/INVESTOR INFORMATION NEEDS

# 17.1. Disclosure as a Cornerstone Strategy

Disclosure is one of the two cornerstones on which our current regulatory and supervisory system is based. It is also a key consumer protection strategy. However, disclosure only works effectively when it is used in combination with other key strategies, such as education, and when there is a competitive environment in which consumer/investors have a real choice as opposed to simply having a choice of the same thing packaged under different labels, with or without a price difference.

There is a lot of concern about the current disclosure system. These concerns relate to whether the disclosure system is working as well as it should be and what can be done to make it work more effectively.

One observation that is frequently made is that disclosure is still aimed at the intermediary rather than at the end investor. If one accepts that this is so (which I believe it is currently reasonable to do) it reinforces the fact that:

- (i) consumer/investors are relying on the advice and recommendations of their respective financial advisers;
- (ii) the obligations of financial advisers to consumer/investors are fiduciary obligations based on the financial advisers' provision of advice to their respective clients that is tailored to meet the integrated needs of consumer/investors, as opposed to a relationship that is simply based on selling product to such clients; and
- (iii) there is a need to ensure that the interests of financial advisers are aligned with those of the end investors who are their clients.

Other observations that are frequently made about disclosure are that:

- (i) there is too much paper;
- (ii) people feel overwhelmed by the paper;
- (iii) people do not want all the paper/information; and
- (iv) people will not read the material that is provided to them.

# 17.2. Effectiveness of Disclosure as a Consumer Protection Strategy

I believe that the key to the effectiveness of disclosure as a consumer protection strategy lies in helping consumer/investors learn how to make effective decisions and how to make use of information in doing so.

This presupposes that consumer/investors have first been motivated to do so and that the quality of the information that is provided is sufficient to enable them to do so.

Hopefully, this motivation will be one of the outcomes of the educational initiatives referred to in Sections 14 and 15 of this Review. This Section 17 contains additional suggestions designed to facilitate better decision-making by consumer/investors.

### Key Role for Intermediaries

There is a key role for intermediaries to play in helping consumer/investors who are their clients to:

- (i) identify their goals, objectives and tolerance for risk;
- (ii) prepare an implementation plan;
- (iii) monitor performance against the plan; and
- (iv) make appropriate adjustments to accomplish their objectives or to modify their strategy if circumstances change.

Each of these activities involves important elements of an effective disclosure strategy as well as activities that can be used as indicators to measure the effectiveness of such a disclosure strategy.

What I am emphasizing is that disclosure is about more than documents. It is about having ready access to information and using it to make decisions effectively, with a full understanding and comprehension of the significance (or lack of significance) of the information that is provided or that is not provided.

For consumer/investors who are reliant on financial advisers for help in connection with the four foregoing matters, it is essential that the terms of the "service offer" clearly outline the services with which the consumer/investor can expect to be provided. These include the identification of the consumer/investor's integrated needs, the preparation of a written plan, the reporting mechanisms and the ongoing monitoring and adjusting that may be necessary.

Suggestions are included in this Section 17 for adapting know-your-client/suitability procedures, confirmation/point-of-sale forms and account statements<sup>125</sup> to better align them to meet the needs of consumer/investors.

Although these matters have customarily been regarded merely as "standard forms" required by regulators, the know-your-client/suitability procedures, confirmation/point-of-sale forms and account statements are much more than that. They are an integral part of the services that consumer/investors need if they are going to become better decision-makers and if they are going to be able to achieve their personal financial goals. They are an integral part of the services that consumer/investors should expect financial advisers to provide in return for the remuneration that they receive from consumer/investors.

The time for consumer/investors to focus on the adequacy of these services is at the time of that they initially retain their financial adviser. The quality of these services is what differentiates financial advisers and is part of the value provided by the financial adviser. If the consumer/investor does not "negotiate" the provision of these services up front by ensuring that they are included, it is unlikely that

88

<sup>&</sup>lt;sup>125</sup> It has not been customary to think of these documents as being part of the disclosure system. However, I think that these documents are, and should be recognized as, an integral part of an effective disclosure system. The suggestions in Section 17 of this Review are aimed at enhancing these documents so that they will be an integral part of such a disclosure system.

financial advisers will voluntarily provide them at a later date as at that point such services will simply be direct expenses that will reduce the profitability of the financial adviser.

While some of the suggestions contained in this Section 17 regarding know-your-client/suitability procedures, confirmation/point-of-sale forms, account statements and ongoing monitoring procedures can be reflected in regulatory and self-regulatory requirements, the quality, usefulness and effectiveness of these procedures, reporting and monitoring are very much a matter of the "service offer" that is provided and the use that is made of them by consumer/investors. It is essential that consumer/investors recognize this and the corresponding need for them to engage themselves in the processes.

### 17.3. The Elements and Dimensions of Disclosure

There is a need to make the information that is provided to consumer/investors timely, meaningful, relevant and readily accessible to them and to their advisers and to enhance the usefulness of the information to them.

To do this, it is important to keep in mind that there are three key elements of "disclosure". These elements relate to:

- the identification of what information is relevant information;
- the communication of the relevant information; and
- the comprehension or understanding of the relevant information.

All three of these elements must be present in order to have "disclosure". If any of the three elements is missing, the result of what you have is not "disclosure". <sup>126</sup>

There are many dimensions to "disclosure". The prospectus is just one dimension of disclosure. Other dimensions include:

- 1. Education that is aimed at improving the ability of consumer/investors to identify and comprehend the information that is communicated to them and to understand what is relevant and not relevant to their own personal circumstances.<sup>127</sup>
- Education and proficiency training for intermediaries that is aimed at improving their ability to identify and comprehend the information that is needed to enable them to meet the needs of their clients - remembering that the intermediation function has shifted (or is in the process of shifting) from being primarily sales-related (with advice being ancillary) to being advisory (with sales being ancillary).<sup>128</sup>

One intermediary in illustrating the importance of "disclosure" in the intermediary-client relationship talked in terms of disclosure being the difference between "honesty" and "rigorous honesty". An example of "honesty" is: "I *told* him/her the facts", whereas an example of "rigorous honesty" is: "I made sure that he/she *understood* the facts."

<sup>&</sup>lt;sup>127</sup> Specific recommendations relating to consumer/investor education are contained in Section 14 of this Review and in Section 15.04 of the 1995 Report.

<sup>&</sup>lt;sup>128</sup> Specific recommendations relating to education and proficiency training for intermediaries are contained in Sections 14 and 15 of this Review and in Sections 15.01, 15.02 and 15.03 of the 1995 Report.

- 3. Regulation of managers of investment funds. 129
- 4. Development of common standards that are consistently and universally applied so that the information that is provided or used is comparable. 130
- 5. Regulation of sales communications and advertising to ensure that sales communications and advertising are not inherently misleading.<sup>131</sup>
- 6. Use of "know-your-client" information, confirmations and account statements to facilitate better decision-making. 132
- 7. Use of continuous disclosure provisions, or "integrated disclosure provisions" if the recommendations in Section 13 of this Review are implemented, to ensure that the marketplace (i.e. the collectivity of all investors) is kept informed about material changes and about operations through the inclusion in annual and interim reports of the fund manager's discussion and analysis of the operations of the investment fund (MD&A).<sup>133</sup>

## 17.4. Problems with the Prospectus Disclosure System

As noted in Section 17 of the 1995 Report, the current prospectus disclosure system is not working as well as it should be. There are several factors that contribute to why the current prospectus disclosure system still is not working as well as it should be despite the substantial improvements that the investment fund industry has made since the 1995 Report in the way that information is presented. These factors relate to problems with the information that is or is not included, awareness problems with the information that is or is not included, and to problems arising under securities legislation. More specifically:

#### Information Problems

- 1. There is a problem with the fragmentation of information in the basic disclosure documents. Key information is currently divided among three documents the simplified prospectus, the annual information form and the financial statements. In many cases only the simplified prospectus is actually delivered. However, except in the legalistic sense of "information being incorporated by reference", the simplified prospectus does not actually contain all of the relevant information. The relevant information is divided among the three documents.
- 2. A lot of the information that investors are actually given is superficial in nature.
- 3. Some of the information that investors are looking for is not required to be given to them.

<sup>&</sup>lt;sup>129</sup> I have suggested that the activities of managers of investment funds be regulated through the registration process as opposed to attempting to regulate their activities through the prospectus process. Specific recommendations for this are contained in Section 14 of the 1995 Report and in Section 19 of this Review.

<sup>&</sup>lt;sup>130</sup> Specific recommendations relating to the development of common standards are contained in Section 24 of the 1995 Report and in Section 16 of this Review.

<sup>&</sup>lt;sup>131</sup> Specific recommendations for this are contained in Section 24 of the 1995 Report.

<sup>&</sup>lt;sup>132</sup> Specific recommendations are contained in subsections 17.12, 17.8 and 17.10 of this Review.

<sup>&</sup>lt;sup>133</sup> Specific recommendations are contained in Section 17 of the 1995 Report and in Section 13 of this Review.

4. There is a lot of generic information that is common to all investment funds that is required to be included in the documents. This adds to the complexity and length of the documents despite the efforts made by the investment fund industry in recent years to present information plainly, simply and in a user-friendly manner.

## Issues Relating to Lack of Awareness

- 5. There is a lack of awareness about what information to look for in the disclosure documents.
- 6. There is a lack of awareness about the implications that flow from the information that investors have or do not have.
- 7. Many financial intermediaries do not perceive the information in the prospectus disclosure documents as being relevant to investors.
- 8. If/when financial intermediaries deliver the simplified prospectus, they often do not identify it as a document that contains the information about the investment fund(s) that an investor should have **before** making a decision to invest.
- 9. The term "prospectus" is not a familiar term to most people so they do not recognize the documents as being something that should help them in making their investment decisions.
- 10. Consumer/investors are often prepared to trust others to make fundamental decisions for them. In doing so they assume that the education, training and proficiency of the intermediary has qualified the intermediary to do so and that the intermediary's licensing, registration or accreditation is sufficient to merit the trust that they place in the intermediary. Consumer/investors also assume that in making decisions for them, the intermediary has made all the due diligence inquiries that are appropriate to the investment decisions being made and that the intermediary has no conflicting interests or incentives to act other than in the best interests of their clients. Sometimes this trust is warranted; at other times it is not. The situation often results in consumer/investors being vulnerable.

#### Securities Law Aspects

11. Securities laws (unlike most consumer protection laws) do not require that the prospectus be delivered to investors **before** the transaction is entered into. However, the time to receive, read and comprehend the information in the prospectus is **before** the investment decision is made.

## 17.5. Premises Underlying 1995 Report Recommendations for Changes in the Disclosure System

The basic premise behind the recommendations in the 1995 Report and in this Review for changes in the disclosure system is that information which is only deemed to have been delivered to consumer/investors but which has not *actually* been delivered to consumer/investors cannot be considered to have been disclosed to them.<sup>134</sup>

Information that has not *actually* been delivered to consumer/investors has not been communicated to them and, in the absence of delivery and communication, consumer/investors cannot be considered to have understood such information.

-

<sup>&</sup>lt;sup>134</sup> This view is contrary to the approach reflected in current securities laws.

A further premise is that information that has not been delivered to consumer/investors until *after* they have made their investment decision is of little use to them in making an investment decision.<sup>135</sup>

Delivery after the fact, combined with the concept of deemed delivery, both operate to the disadvantage of consumer/investors who attempt to seek redress as a result of failure to disclose material information or as a result of negligent advice.

In other words, these delivery provisions of applicable securities law result in the information in the various documents being able to be used to protect the investment fund, its manager(s) and intermediaries as opposed to being a consumer protection measure aimed at enabling consumer/investors to make effective decisions or to give their informed consent to various transactions.

Accordingly, these delivery provisions of applicable securities law are inconsistent with a consumer protection strategy that is aimed at promoting effective decision-making by consumer/investors. Instead of doing this, the current disclosure system has the opposite effect of allowing the key disclosure documents to be converted into liability-limiting documents.

Lack of actual disclosure also has serious consequences from a variety of other perspectives. One of the most serious consequences flows from the fact that in common and statutory law the "disclosure" of conflicts of interest and of other related party transactions will permit certain self-dealing to take place without the corresponding liability to account for the profits or losses thereby incurred.

If the pertinent information relating to the disclosure of conflicts of interest has not been delivered to investors, it certainly has not been communicated to them nor has it been understood by them. It is therefore doubtful that consumer/investors will be considered to have given their informed consent to the transactions. This result could have significant financial consequences for fund managers and their affiliates in the event of subsequent legal actions that seek an accounting.

Another consequence of lack of disclosure is the impact that it has on the ability of consumer/investors to understand the suitability of their investments for the purpose of meeting their goals and objectives.

Two other premises that underlie my recommendations respecting disclosure are that:

- (i) it is possible to have full, true and plain disclosure which is simply presented; and
- (ii) it is possible to combine information, communication and understanding.

I believe that as the knowledge and awareness level of consumer/investors increases (which is happening rapidly) consumer/investors are realizing that what they need is more information rather than less - but they want this information to be timely, relevant, readily comprehensible and accessible. *In other words, they are looking for quality information, plainly and simply presented, that will help them make effective decisions either with or without the help of a third party adviser.* 

It is therefore important that the regulatory strategies recognize this changing state as consumer/investors make the transition from a state of dependency on intermediaries to make decisions for them to empowering themselves to make effective decisions with or without professional assistance.

<sup>&</sup>lt;sup>135</sup> To expect consumer/investors to seek out material information is akin to negative option billing. It is advantageous and convenient for the supplier but not for the customer.

The recommendations in the 1995 Report are aimed at providing for delivery of timely, relevant and meaningful information to people in manageable segments to help them make effective decisions.<sup>136</sup>

## 17.6. The CSA Proposals for Mutual Fund Prospectus Disclosure

The recent CSA proposals for mutual fund prospectus disclosure<sup>137</sup> reflect many of the 1995 Report recommendations in this respect. Examples include:

- the separation of generic information and educational information from the prospectus disclosure information;
- the use of non-repetitive plain language;
- the inclusion of performance information shown on a year-by-year calendar year basis with comparison to the relevant indices;
- the inclusion of summary financial information;
- the use of websites for access to information;
- the recognition of the need to provide enhanced annual and interim reporting to investors about the operations of the investment fund.

While the CSA proposals go a long way to address the concerns that have been raised about the current mutual fund prospectus disclosure system, they stop short of permitting all of the relevant information to be included in a single document that is actually delivered to investors.

Instead, the proposals require that only a fund summary actually be delivered to investors. The fund prospectus is required to be delivered to investors only if they ask for it. The information that may be included in the fund summary is restricted and the information that is included must be presented in a prescribed order.

The CSA's intended purpose in restricting the information and in prescribing its order is to facilitate comparison. While there may be theoretical merit in this objective, there is a practical concern that the restrictions, in focusing on form rather than on substance, may get in the way of providing meaningful disclosure about the investment fund.

In the attempt to standardize the presentation of information for ease of comparison, the requirements may have the unintended result of making the information so "boilerplate" that it will cease to be read at all, to have meaning, or to be relevant.

The requirements may also have the unintended effect of stifling efforts to find better ways of communicating concepts to investors and of differentiating the specific fund(s) described in the fund summary from those offered by others.

\_

<sup>&</sup>lt;sup>136</sup> See Section 17 of the 1995 Report.

<sup>&</sup>lt;sup>137</sup> See subsection 3.5 of this Review.

In addition, the likelihood of using the fund summaries for comparison purposes is somewhat remote as there are now other facilities available that permit more ready comparisons as well as more up-to-date ones. At the most basic level if one is simply looking at the numbers, there is the information published in newspapers. This information shows extensive comparative information on a category-by-category basis. More detailed analysis and comparison is readily available through programs and services such as Globefund, Paltrack and Bell Charts. Globefund includes a free charting feature that enables a user to compare the performance of mutual funds against indexes like the TSE 300, Dow Jones or Standard and Poor or against any other mutual funds that the user is interested in.

The comparative and analytical information that is now readily available to investors at no incremental cost is a relatively new development in the marketplace that has been made possible by the significant advances in technology and communications facilities that have occurred in recent years. Technology and communications facilities offer more meaningful and up-to-date disclosure alternatives than the current regulatory requirements. This is an example of why some fundamental rethinking of the CSA disclosure proposals is desirable.

There are other concerns about the CSA proposals for mutual fund prospectus disclosure. These include:

- (i) a concern that the CSA proposals lag the marketplace which is recognizing the desirability of putting all of the relevant information about a particular fund or group of funds into one document that is actually delivered to investors;<sup>138</sup>
- (ii) a concern that the proposed fund summary document is too long and that it is too rigid in what it must and must not contain;
- (iii) a concern that relevant information will not be contained in the document that is *actually* delivered to investors but instead will be found only in a document that remains in regulators' files yet nevertheless will be deemed to have been given to them for liability purposes; the fact that the CSA proposals include time constraints for delivering the additional information if it is requested by investors does not address the fact that investors still have to ask for the information rather than simply having it in hand, in one place, clearly and plainly written, without repetition; and
- (iv) a concern that the requirements for outlining the investment objectives and policies of the investment fund do not result in appropriately identifying the portfolio risk of the investment fund.<sup>139</sup>

It should be recognized that the CSA's proposed fund summary document is not a substitute for the point-of-sale document recommended in the 1995 Report, the purpose of which was to highlight, at the point of sale, the key investment facts reflected in the decision that was made by the investor to invest in the fund in question. The point-of-sale document recommended in the 1995 Report presupposed the delivery of the fund prospectus and that the information in the fund prospectus formed the basis for the investor's decision.

-

<sup>&</sup>lt;sup>138</sup> An example of this is seen in the Simplified Prospectus dated February 2, 1998 issued by Fidelity Investments Canada entitled "Your guide to investing in Fidelity funds". In a press release relating to the issuance of this prospectus dated March 23, 1998 and entitled "New Educational Prospectus Praised by Investors, Regulator", Fidelity Investments Canada stated that "Canadian investors who reviewed the new Fidelity prospectus in focus groups almost unanimously endorsed the usefulness of the document."

<sup>139</sup> See Section 25.03 of the 1995 Report.

One of the most fundamental concerns about the CSA proposals for mutual fund prospectus disclosure relates to the fact that the fund summary document is not required to be delivered to investors **before** the investment decision is made. The failure to make such changes undermines the effectiveness of disclosure as a consumer protection strategy.

## 17.7. Enhancement of the CSA Proposals for Mutual Fund Prospectus Disclosure

While a lot of good work is reflected in the CSA proposals for mutual fund prospectus disclosure, they stop short of providing all the help consumer/investors need to make effective decisions with or without the help of an adviser.

With some changes, the CSA proposals could be modified to be of real benefit to consumer/investors in helping them to make effective decisions. The suggested modifications presuppose<sup>141</sup> that:

- 1. Generic and educational information will be eliminated from the prospectus and presented in a separate document.
- 2. Summary financial information will be included in the prospectus in lieu of the full financial statements.
- 3. Full financial statements will be posted on a website that is freely accessible to the public and clearly identifiable on a fund-by-fund basis.
- 4. Non-repetitive plain language will be used.
- 5. If a prospectus covers several funds in one document, the key information about each fund (as indicated in the CSA proposals) will be included on separate pages.
- 6. Marketing material will not be included in the prospectus or wrapped around it.
- 7. Performance information shown on a year-by-year calendar year basis with comparison to the relevant indices will be included.
- 8. Websites that are freely accessible to the public will provide for access to the prospectus for the fund(s) on a clearly identifiable basis.
- 9. Enhanced annual and interim reporting to investors about the operations of the investment fund(s) will be provided to investors.
- 10. All information will be permitted to be delivered in electronic and/or paper form according to the preference of the respective investors.

-

<sup>&</sup>lt;sup>140</sup> This is an area where legislative change would be needed in order to require such delivery. However, there is nothing that prevents industry participants from adopting procedures designed to ensure that delivery is made prior to the investment decision being made.

<sup>&</sup>lt;sup>141</sup> The CSA proposals reflect the assumptions outlined in items 1, 2, 4, 5, 6, and 7 of subsection 17.7. They do not fully reflect the remaining assumptions although in some cases (such as item 9) there is an indication that this will be the subject of ongoing work. Item 8 is not precluded by the CSA proposals. Item 10 will hopefully be permitted by the guidelines respecting the use of the Internet for the delivery of information to investors which are expected to be issued shortly.

## Suggested Enhancements to CSA Proposals

Based on the foregoing assumptions, the changes that I suggest be made to enhance the CSA proposals are as follows:

- 1. A plainly written generic, educational information document about investment funds including:
  - (i) what they are;
  - (ii) how they work;
  - (iii) how they are regulated;
  - (iv) how they are taxed; and
  - (v) what to look for in, and how to use, a fund prospectus, annual and interim reports, confirmations and account statements;

should be required to be delivered to investors as early as possible in the investment process. All of this information would be in one document. It could be a stand-alone document or it could be included as a separate part of the complete offering document referred to in item 2 below.

- 2. All of the relevant information should be required, or at least permitted, to be included in one simple but complete offering document the fund prospectus. This document should be required to be plainly and simply written in a non-repetitive manner. The complete offering document should be required to be *actually* delivered to investors *before* they invest in the fund(s).
- 3. A term sheet (point-of-sale outline) should be delivered either before or with the confirmation or could be included as an integral part of the confirmation.<sup>143</sup>
- 4. Material documents should be required to be posted on websites with their availability noted for investors. Examples of material documents include the constating documents of the investment funds and management contracts.
- 5. Disclosure relating to the overall portfolio risk of an investment fund should be required to be included in the fund prospectus and, if used, the fund summary.

It should be kept in mind that presenting information plainly and simply does not require that relevant information be eliminated or be re-categorized as non-relevant information.

The basic information is not complicated. However, a whole industry has grown up around making it appear so. Fund managers, distributors, financial planners, lawyers, accountants and regulators have all had a hand in this. However, when you strip away generic and educational information, full financial statements, repetitive information, summaries of summaries and marketing material, it is possible, practical and helpful to investors to put all of the remaining information into one offering document that will provide investors with the plain and simple facts, in a way that can be readily understood. The purpose of doing this is to enable people in possession of all of the facts to make an informed investment decision that will suit their needs. A lot of people play lip service to this but the reality is that it suits many people's purposes to keep investors uninformed and in the dark.

<sup>&</sup>lt;sup>143</sup> See subsection 17.8 of this Review.

It should also be kept in mind that the requirements to include summaries often tend to result in repetitive information that people do not read as opposed to providing useful information that people do read. A person should only have to read clearly-presented information once in a document.

#### 17.8. Confirmation Forms

One of the key dimensions to disclosure is the confirmation statement.

This is so despite the fact that confirmation statements:

- (i) are not commonly thought of as being part of the disclosure system; and
- (ii) their utility is limited by the fact that the disclosure requirements for confirmation statements have not been updated to reflect the major changes that have occurred in the last eleven years in the commission structure as a result of the introduction of deferred sales commissions, waived or zero front-end sales commissions and trailing commissions.

The result of this unfortunate combination of circumstances is that the confirmation statement is the most underutilized opportunity for communicating key information to investors.

With the increased popularity of selling investment funds on a deferred sales commission basis, confirmation statements no longer reflect that any commission is being "charged in respect of the trade".

The reason is that the statutory requirements for confirmation statements<sup>144</sup> are being literally interpreted to conclude that because the investor is not directly paying the intermediary's compensation in respect of the trade, the intermediary is not "charging a commission in respect of the trade" and accordingly, no disclosure is required in the confirmation statement of the amount that the intermediary is receiving either at the time of the sale or on an ongoing basis in respect of the trade.<sup>145</sup>

In addition, the statutory requirements for confirmation statements are being interpreted as not requiring disclosure to be made of the deferred sales commissions that are payable if the investor redeems during the currency of the deferred sales commission period. The basis for this is that such transaction(s), if there are any, will be different trades and therefore no disclosure of the deferred sales commission is required in respect of the current trade.

Similarly, when mutual funds are sold on a waived or zero front-end sales commission basis with a high trailing commission, there is no disclosure made in the confirmation statement of the payment of sales commission in the form of the ongoing trailing commission that is paid with respect to the investor's

-

<sup>&</sup>lt;sup>144</sup> Section 36 of the Securities Act (Ontario) sets out the requirements for confirmation statements. Similar requirements are included in the securities legislation of the other provinces and territories of Canada.

<sup>&</sup>lt;sup>145</sup> When mutual funds are sold on a deferred sales commission basis, the manager of the fund pays the intermediary a sales commission at the time of the sale. This sales commission is usually 5% (but it may be more or less depending on the mutual fund in question). The intermediary also receives additional sales commissions on an ongoing basis by way of trailing commissions that are paid to the intermediary by the manager. These trailing commissions, depending upon the mutual fund in question, range from 25 basis points to 150 basis points (and may even be higher) with the usual trailing commissions being 50 basis points or 100 basis points. Generally, the trailing commissions that are paid when mutual funds are sold on a deferred sales commission basis are half the amount of the trailing commissions that are paid when the mutual fund is sold on a front-end sales commission basis (or a zero or waived front-end sales commission basis).

investment in the fund on the basis that the payment is not coming directly from the investor and therefore no "commission is being charged in respect of the trade".

The same situation applies in the case of funds that describe themselves (or are described) as being "no load funds" despite the fact that the managers of these funds pay commissions in the form of trailing commissions to intermediaries on an ongoing basis with respect to the investment in question. 146

In these circumstances, it is no wonder that consumer/investors have no idea of the cost of investing or of the impact that this cost has on their total return because no information is given to them about the cost of investing.<sup>147</sup>

## Recommendations Relating to Confirmation/Point of Sale Disclosure

Although it has not been customary to regard them as such, confirmation statements are truly a "point-of-sale" disclosure document. It is here that the cost of investing needs to be highlighted.

It is my recommendation that priority should be given to updating the disclosure requirements for confirmation statements and to making these confirmation statements an integral part of the disclosure system.

The intermediary's compensation in all its dimensions<sup>148</sup> needs to be clearly disclosed together with the impact of this compensation on the investor's total return over a period of years. In addition, all charges that may be payable by the investor need to be clearly disclosed.<sup>149</sup> This is essential information for consumer/investors to have to assist them in making effective investment decisions.

The practice in the last eleven years or so (which has been based on a literal interpretation of disclosure requirements that were designed with a different compensation structure in mind) has resulted in the industry's compensation structure becoming less than transparent. This situation should no longer be allowed to prevail over the interests of consumer/investors in knowing what it is costing them, directly and indirectly, to invest.

The confirmation statement/point-of-sale disclosure document<sup>150</sup> should also include other key information such as:

Many investors think that they are not paying any sales commissions or charges. One industry participant with whom I talked told me that there are "financial planners" going into high schools and telling the students that there are no fees or sales charges for investment funds. Several firms that sell mutual funds on a waived or zero front-end sales commission basis leave the impression in their advertisements that because there are no immediate sales commissions payable by the investor there is no cost to the investor notwithstanding the substantial sales commissions that are payable by the manager of the fund to the intermediaries in the form of trailing commissions relating to the investment.

<sup>&</sup>lt;sup>146</sup> I am not aware of any regulator having focused on or having taken issue with the industry's interpretation of the statutory requirements for confirmation statements.

<sup>&</sup>lt;sup>148</sup> This would include wrap fees, transaction fees and account maintenance fees as well as sales commissions paid by the investor or by the manager and trailing commissions.

<sup>&</sup>lt;sup>149</sup> This would include early redemption fees, deferred sales commissions or charges, administration fees charged on sales transactions or withdrawals, and termination fees.

<sup>&</sup>lt;sup>150</sup> Some intermediaries currently use variations of a point-of-sale disclosure form. Joe Killoran, a consumer advocate, for several years has urged the adoption of a mutual fund checklist for consumers. He urges consumers "[not to] buy or be sold

- whether the investment fund is a proprietary fund or a third party fund;
- whether the investment is insured by the CDIC;
- whether the investment is covered by customer protection funds such as CIPF, CompCorp or other such funds;
- the period during which withdrawal rights may be exercised;
- the period during which an action may be brought for damages based on misrepresentation in the prospectus;
- the manner in which redemption instructions may be given (e.g. in writing, by fax or by phone or by e-mail) and the time by which such instructions must be received for same day processing;
- the importance of retaining the confirmation statement for the investor's records.

The suggestions in this subsection 17.8 apply to all forms of investment funds with the necessary modifications to reflect the different types of investment funds. They also apply to wrap accounts of whatever nature and kind.

### 17.9. Consumer/Investor Checklist

A useful tool for consumer/investors would be to develop a consumer/investor checklist based on the suggestions made in subsection 17.8 of this Review for confirmation statements/point of sale disclosure documents plus some additional information, such as management expenses, the fund category and the relevant benchmark(s) for the fund category. This checklist would be a companion piece to the generic educational and information document referred to in subsection 17.7 of this Review. It would be designed to be used by consumer/investors as a type of self-help worksheet as opposed to being a form required to be provided by an intermediary.

This checklist would be used to highlight the implications of the information given. For example, it would highlight that there are some "buyer beware" considerations to take into account when a proprietary fund or an intermediary's wrap program is recommended. These considerations include:

- (i) whether there are internal requirements that sales representatives have a certain proportion of clients' assets invested in these funds or programs; and
- (ii) whether it will make a difference to the amount of future income on retirement that the sales representative will receive.

Requirements of this nature raise the question of whether the advice given to the consumer/investor is the best advice for the consumer/investor or the best advice for the sales representative whose immediate and deferred compensation may be affected by the amount of assets under administration that are invested in such funds or programs.

a mutual fund without it!" His checklist includes many of the items suggested in subsections 17.8 and 17.9 of this Review for inclusion in a confirmation/point-of-sale disclosure form and in a consumer/investor checklist.

There is also the question of whether it will be more difficult, take longer and/or be more costly for consumer/investors to withdraw their assets if they wish to transfer their accounts to other intermediaries.

#### 17.10. Account Statements

Another key dimension to disclosure is the account statement. Again, it has not been customary to view account statements as being part of the disclosure system. However, they are an essential part of it. Account statements provide an important opportunity to give investors the information that will answer their most basic question which in this context is "just tell me how am I doing".

Consumer/investors should be able to use their account statements both to check the accuracy of the recorded transactions and to monitor whether their portfolio allocation is appropriate to what their goals are. In this respect, there is a need to develop some measures that will help consumer/investors understand how they are doing. The information provided by these measures will help them to evaluate how their investment objectives are being met and whether modifications are needed.

Performance information by itself is insufficient for this purpose. This is particularly so when performance information is given at the fund level and without any analysis of the risk that the fund is exposed to in order to achieve its results.

All too often the information that is necessary to enable consumer/investors to monitor how they are doing against their plans and whether their portfolio allocation is appropriate for what their goals are is missing from account statements. In addition, many investors find their account statements difficult to read.

Intermediaries are increasingly recognizing the need to make account statements more readily understandable and some are trying to make them more useful for their clients. These efforts should be encouraged. In addition, the voluntary efforts should be supported by minimum regulatory requirements.

The present requirements for account statements<sup>151</sup> are primarily transaction-oriented rather than advisory-oriented. They do not reflect the shift by intermediaries from a transaction-based relationship with their clients (with the advice being ancillary) to an advisory-based relationship with their clients (with the transactions being ancillary). They do not reflect the substantial bundling of services that is occurring through wrap accounts, pooling and asset allocation services, and fund-of-fund arrangements.

In particular, the exemption that is provided in the present securities legislation requirements for account statements that allows mutual fund dealers to send account statements only once a year needs to be reviewed. Several industry participants have commented on the inadequacy of this provision from the perspective of the consumer/investor.

The subject of account statements, what should be in them, how frequently they should be sent and who should send them is a complex one. This is an area where substantially more work needs to be done. <sup>152</sup>

<sup>&</sup>lt;sup>151</sup> See for example Section 123 of the Regulation made under the Securities Act (Ontario).

One industry participant described the issue of account statements as being a critical issue. This person described account statements as being a crucial tool to enable self-reliant people to be responsible for their own affairs and as being

In talking with industry participants about what would help consumer/investors better to identify their goals, the means of achieving them and to assess risk, a frequently given answer was "better know-your-client procedures" and "better account statements". The two go hand-in-hand. The use of know-your-client procedures is discussed in subsection 17.12 of this Review. It is important to read the suggestions for account statements in this subsection 17.10 with the recommendations in subsection 17.12 of this Review for know-your-client/suitability procedures.

Without attempting to answer the above-noted questions regarding account statements, here are some suggestions made by industry participants for including information in account statements that would benefit consumer/investors.

- 1. Regard account statements as an opportunity to report to clients on how they are doing according to their stated investment goals and targets within the ambit of risk that they are willing to tolerate. This reporting would be in addition to providing statements of the investment portfolio and of the changes therein. This type of enhanced reporting is consistent with the intermediary's shift in emphasis from a transaction-based relationship to an advisory-based relationship. The current form of account statements is still reflective of transactional reporting only. The current form of reporting needs to be brought into line with the type of reporting that investment advisory clients usually receive.
- 2. Include individual internal rate of return numbers and time-weighted rate of return numbers in order to enable the client to evaluate the client's performance in particular funds vis-à-vis various benchmarks and other investments.<sup>153</sup>
- 3. Include appropriate capital market benchmarks with the relevant fund-specific category benchmarks for the same period for comparison purposes.
- 4. Show risk-adjusted returns. Performance numbers that have not been adjusted for risk are misleading.<sup>154</sup>
- 5. Illustrate through the use of charts and graphs the correlation of these performance numbers and benchmarks. Logarithmic or percentage charts should be used instead of linear or arithmetic scale charts.
- 6. Include attribution analysis information to show how the returns have been derived.

an equally important tool for those who are reliant on third parties. This person observed that it is essential that the transactional and the advisory sides be very clear and that it was necessary for both the industry and its regulators to focus on this.

"And when Ms. Modigliani combined risk and return, she found that past risk-adjusted return is a decent indicator of future risk-adjusted return. It's a lot better than just looking at returns. "Selecting funds on the basis of risk-adjusted performance is better than selecting based on total return," Ms. Modigliani says."

<sup>&</sup>lt;sup>153</sup> See Scott Mackenzie, Advisor's Edge, October 1998 - Understanding performance figures.

<sup>&</sup>lt;sup>154</sup> See Robert McGough, The Wall Street Journal, August 7, 1998, *Heeding Risk-Adjusted Returns May Pay*. This article refers to a research study by Leah Modigliani, a U.S. equity strategist at Morgan Stanley Dean Witter & Co. which looked at 660 stock funds that had been in existence for the full 10 years through the end of 1997. The study looked at all types of stock funds. The article reports that "she found that a top total return is absolutely, positively *not* a good predictor of future returns. But she found that past risk is an extremely *good* indicator of how risky a fund will be in the future." The article goes on to state that:

- 7. Show after sales charges returns.
- 8. Show the impact of income tax on returns in taxable accounts assuming various marginal tax rates.
- 9. Show clients the amount by which their account has appreciated/depreciated:
  - show the amount of the account at the beginning of the period;
  - add dividends and interest received during the period;
  - subtract withdrawals during the period;
  - add contributions made during the period;
  - show the market value at the end of the period; and
  - show the value of the account at inception.
- 10. Show the amount of compensation (including the amount of the trailing commissions) paid to/received by the dealer/representative during the account period in respect of the client's account.
- 11. Include graphs comparing average account balances to the management fees, expenses and charges.<sup>155</sup>
- 12. Show how much the client has paid on account of management fees, expenses and other charges (including fees, charges and expenses for transactions, wrap accounts, insurance fees, pooling arrangements, asset allocation services and other arrangements of every nature and kind) per \$10,000 invested. Show the impact of these fees, charges and expenses on a cumulative basis over time.
- 13. Different account statement formats for different types of accounts are probably required.
- 14. The frequency of account statements might vary depending on the type of accounts involved.

Enhanced account statements of the nature described above are problematic for intermediaries whose advice may be tempered by the product-generated fees that they currently receive on account of proprietary or third party products (including wrap accounts, pooling arrangements and asset allocation services) particularly as the layers of costs eat into capital.<sup>156</sup> Meaningful disclosure to investors is

<sup>&</sup>lt;sup>155</sup> Many people do not have any idea of the fact that they are paying management fees and operating expenses in respect of their investment in a mutual fund because the performance numbers that are shown are after deduction of such fees and expenses. The person suggesting the inclusion of this graph observed that this graphical illustration would heighten people's awareness of, and sensitivity to, fees and charges with the result that consumer/investor pressure would cause fees and expenses to come down by one or two basis points every year.

<sup>&</sup>lt;sup>156</sup> In Letter #197 dated June 30, 1998 published by KPA Advisory Services Ltd. entitled *Today's Financial Food Chain: Getting The Customers On Top*, Keith Ambachtsheer writes:

<sup>&</sup>quot;Because of the power of compounding, investment costs have a major impact on terminal wealth in long term investment programs. Thus for example, a common rule of thumb is that every reduction of 1% in long term

bound to serve as an impetus for the industry to realign its compensation system, the need for which has been discussed earlier in this Review.

Governments and regulators need to ensure that meaningful disclosure occurs.

## 17.11. Consolidated Reporting and Account Statements

The desire of consumer/investors for assistance in consolidating and evaluating information about their investment performance will present a business opportunity for independent third party consolidators/custodians who position themselves to capture consumer confidence by not having relationships with investment funds, investment advisers or dealers that compromise or may be viewed to compromise their independence.

As consumer/investors become increasingly aware of the importance of having:

- (i) investment objectives and goals;
- (ii) a strategic plan for achieving their investment objectives and goals;
- (iii) a means to monitor how they are doing against their plan; and
- (iv) a means to identify what realignment (if any) is necessary;

they will increasingly be looking for help in consolidating the information that they receive from various sources.

They will also be looking for measurement and evaluation services that they can trust to be objective, unbiased and unskewed to any product or service. The measurement and evaluation services that they will be looking for include sector and objective analysis, money manager evaluation, trading cost analysis, how effectively portfolio transactions have been executed and report generation facilities showing comparisons to benchmarks and related graphs.

In addition to independence and objectivity, consumer/investors will be increasingly conscious of the need to ensure the confidentiality of the information provided to them and the privacy of their information in the consolidator's data bank.

All of these information needs of consumer/investors are currently available from a variety of sources. However, in order to provide consumer/investors with the benefit of this information there is a need for a mechanism to consolidate the disparate parts, free from any product or service bias.

It would be most advantageous for consumer/investors to be able to access this consolidation service separate and apart from the requirement to participate in products such as the various types of wrap accounts which add yet another layer of product management cost for consumers that negatively impacts their capital.

return on retirement saving leads to a 20% reduction in the ultimate pension it can buy. This proposition is not opinion. It is simple fact."

## 17.12. Know-Your-Client/Suitability Procedures

Know-your-client/suitability procedures are the starting point for enabling consumer/investors to make use of the information that is provided to them:

- (i) by the financial planners/investment advisers with whom they deal;
- (ii) by the prospectuses that they receive;
- (iii) by the annual and interim reports that are sent to them;
- (iv) by the confirmation statements/point-of-sale disclosure documents that are issued to them;
- (v) by their account statements; and
- (vi) from other sources.

The know-your-client/suitability procedures are the starting point because it is here that consumer/investors focus on what their particular needs, objectives and goals are. All of the other elements and dimensions of disclosure flow from this focus. The know-your-client/suitability procedures are a critical dimension of the disclosure system although they are not commonly thought of as being a part of it.

The know-your-client/suitability procedures have their origin in the provisions contained in applicable securities legislation. These procedures require dealers and advisers to make the enquiries that will enable them to establish the identity and, where applicable, the creditworthiness and reputation of each client. Dealers and advisers are also required to make the enquiries that are appropriate, in view of the nature of the client's investment and of the type of transaction being effected for the client's account, to determine the general investment needs and objectives of each client and the suitability of a proposed purchase or sale for that client.

The know-your-client/suitability procedures are generally satisfied by the dealer or adviser completing a standard form with or without the involvement of the client in question. Often the form is not signed by the client and is not shown to the client. It is not mandatory that a copy of the form be given to the client. The know-your-client/suitability procedures vary from dealer to dealer. Some dealers (such as discount brokers) and some banks require that a detailed account application form be filled out and signed by the client before opening an account. The know-your-client/suitability procedures followed by advisers often differ from those followed by dealers and reflect more involvement with the client in identifying the client's investment objectives and goals.

The know-your-client/suitability requirements (despite the fact that they apply to advisers as well as to dealers) have been developed primarily with transaction-based relationships in mind rather than advisory-based relationships. The procedures have been used mostly as a means of protecting the intermediary from liability based on claims arising out of transactions entered into on the client's behalf rather than being used as a mutually advantageous basis for managing the client relationship. However, as industry participants move towards an advisory-based relationship with clients, they are

<sup>&</sup>lt;sup>157</sup> See for example subsection 114 (4) - (7) of the Regulation made under the Securities Act (Ontario).

<sup>&</sup>lt;sup>158</sup> Some clients object to completing these forms because they feel that it is an intrusion on their privacy and that the information will be used for unwelcome cross-marketing purposes.

increasingly recognizing that the know-your-client/suitability procedures provide a useful foundation for establishing, monitoring and maintaining a client-needs based advisory relationship.

The time is therefore ripe to adapt the know-your-client/suitability requirements for use as a critical tool in the establishment, monitoring and maintenance of an ongoing advisory-based relationship with clients. Such action would be of mutual benefit to consumer/investors and intermediaries. This would require adapting the know-your-client/suitability requirements so that they would form the basis for:

- identifying and assessing the integrated needs of each client;
- developing a plan aimed at achieving such client's needs; and
- providing ongoing monitoring to determine the effectiveness of the plan to meet the identified needs and objectives of the client.

Here a useful model that might be used is the one used by pension funds and other institutional accounts. This model starts with the preparation of an agreed-upon statement of investment goals that identifies (i) how these goals will be achieved, (ii) the indicators, benchmarks and performance measures that will be used to measure the achievement of these goals, (iii) the frequency and type of reporting to the client, and (iv) the method and frequency of the ongoing review of the statement of investment goals to maintain its currency or to adapt it to meet changing circumstances.

Some intermediaries have already adapted this institutional model for use with consumer/investors, recognizing that the starting point for a mutually successful advisory-based relationship is with the development of a true financial plan which clearly identifies client needs, objectives, goals, timeframes, implementation procedures, review and monitoring procedures, agreed-upon indicators, benchmarks and performance measures that will be used to measure the achievement of these goals, the frequency and type of reporting to the client, the method and frequency for ongoing review and updating of the plan to maintain its currency or adapt it to meet changing circumstances.

The account statements referred to in subsection 17.10 of this Review would complement the know-your-client/suitability financial plans and provide consumer/investors with the information they need to understand how they are doing in achieving their goals and the degree of risk to which they are exposed.

Appendix D to this Review contains a summary of some observations that have been made by industry participants with respect to the know-your-client/suitability procedures and financial planning generally that are aimed at making the procedures and practices more useful and effective for consumer/investors and for industry participants.

As can be seen from the industry observations set out in Appendix D, there are a lot of different approaches as to how to help clients develop appropriate, dynamic plans that will enable them to better identify their goals, the means of achieving them and to assess risk. This is what the know-your-client/suitability procedures are all about. This is what an advisory-based relationship based on client-needs is all about. This is what the purpose of disclosure in all its elements and dimensions is all about. This is what consumer/investor education is all about. This is what education and proficiency training for industry participants is all about.

The dynamic tools that technology has made available serve to facilitate both the reliant consumer/investor and the self-reliant consumer/investor to utilize all of the converging elements to more effectively identify their goals and how to achieve them within the risk parameters that they are

comfortable with. All of the elements and dimensions of disclosure converge to enable consumer/investors, with or without the help of intermediaries, to make effective decisions on a fully-informed basis.

## 18. IMPROVING DISCLOSURE

## 18.1. Better Disclosure of Fees and Charges

Surveys show that consumer/investors are generally not aware of the fees, charges and expenses that they pay with respect to investing in investment funds. Part of the reason why consumer/investors are not aware of the fees, charges and expenses that they pay is because the amount they pay is "deducted at source" and all returns are stated on a net basis.

Fees, charges and expenses are akin to a hidden tax. When people do not realize that they are paying it, they do not question it. This factor, combined with the very strong performance that most investment funds have enjoyed in the last several years, has masked the impact of the increasing fees, charges and expenses that are borne by consumer/investors.

In an environment where costs are simply passed on to the consumer/investor, there has been little incentive for fund managers to operate more efficiently or to ensure that trading costs are kept to a minimum while ensuring that the best prices are obtained on the purchase and sale of portfolio securities.

Some industry participants have advocated regulatory intervention to require "all-in" pricing of investment funds. By this they mean that the management fee charged by the investment fund manager would include the provision of all services required by the investment fund and that there would be no additional expenses charged to the fund. This method of pricing was popular some years ago but the industry has moved away from it. Today, some fund groups achieve "all-in pricing" by agreeing to absorb all expenses over and above a stated amount.

I suggest that before resorting to regulatory intervention, an effort should be made to improve the disclosure to consumer/investors of the fees and charges they are paying and the impact on their total return over a period of years. <sup>159</sup> I would therefore recommend that the management expense ratio be "unbundled" to disclose as separate line items both in dollar amounts and percentage terms:

- the advisory fee;
- the business administration fee;

<sup>&</sup>lt;sup>159</sup> See *Today's Financial Food Chain: Getting The Customer On Top*, referred to in Footnote 156. In the same letter, Keith Ambachtsheer states:

<sup>&</sup>quot;... most customers still don't understand the devastating impact high fees have on long term investment results. While this economic reality is finally beginning to impact how large pension funds are managed, it still has not sufficiently penetrated into the smaller fund arena. As to the mutual fund sector, a recent survey of 2000 Canadian mutual fund holders found that an astonishing 40% of the customers surveyed were not aware they were paying any fees at all!"

- the distribution fee, including:
- (i) the deferred sales commissions that are paid, and
- (ii) the trailing commissions that are paid;
- the insurance fees (if any);
- the operating charges (broken down into their major components);
- the total of these amounts and the proportion the total represents of the average net assets of the investment fund;
- the impact of the amount of these fees and charges per \$10,000 invested in the investment fund over a period of years; and
- a graph or chart illustrating the aggregate amount of these fees and charges over a period of years compared to the average assets under administration during the period in question.

Where wrap accounts, asset allocation services, fund-of-fund arrangements, segregated accounts or other bundled service arrangements are involved, the fees, charges and expenses should be similarly "unbundled" and shown.

I also think it would be helpful to consumer/investors to show these amounts in comparison to a relevant benchmark. With the fund categorization project referred to in subsection 3.8 of this Review, it should be possible to provide appropriate comparison numbers related to the investment funds that are highest in the class, lowest in the class and the mean.

This detailed information should be included in the prospectus (including any fund summary) as well as in the annual and interim reports and perhaps even in the account statements.

The theory behind providing this enhanced disclosure is that when investors see and understand the nature of the costs that are being passed on to them they will question the appropriateness of these costs being borne by them and/or look for lower cost alternatives. This awareness should reduce the information gap and should result in costs coming down. Whether this is what will happen is beyond my ability to predict. However, it certainly will not happen as long as there is no transparency of the fees and charges and of their impact. <sup>160</sup>

The current cost structure of investment funds raises the question of whether consumer/investors who invest on a front-end sales commission basis should bear the costs of distribution that are attributable to investors who invest on a deferred sales commission basis. I do not think it is appropriate that they do so. I also think there is a need to focus on the impact that fund-of-fund arrangements, asset

<sup>&</sup>lt;sup>160</sup> Some industry participants think that it will never happen. They say that neither the media nor the regulators will press the issue, that financial advisers, for obvious reasons, will not and that consumer investors will remain passive. However, a contrary view is expressed by Keith Ambachtsheer in his letter referred to in Footnote 156. In this letter, Keith Ambachtsheer observes that the profile of the issues (one of which relates to the impact of fees and charges on investment returns) is rising. He states that the issues are:

<sup>&</sup>quot;receiving increasing attention from the media and from the regulators. An increasing number of pension fund trustees are 'getting it'. Eventually, retail investors will begin to 'get it' too."

allocation services, segregated funds and fund-cloning have on unitholders of the core fund(s) that are involved

The current cost structure of investment funds once again raises the question of whether consumer/investors should be bearing any costs of distribution. The rebating of fees and commissions seems to be increasing as investors begin to question the cost of participating in investment funds, particularly when their investments in an investment fund are substantial. This whole area of rebating needs to be looked at from the perspective of whether it is fair to the other investors in the investment fund and to potential investors.

## 18.2. Timely Disclosure of Portfolio Holdings

There are increasing requests from investors that the statements of the investment portfolio of an investment fund and of the changes therein be filed more frequently and be more readily available for review.<sup>161</sup> Technology has made it feasible to do this.

This portfolio information is available but it generally is provided only to proprietary information service gatherers. It is time to make changes in this area to provide for equality of access to, and timely disclosure of, this portfolio information to all investors and to make the information available through the Internet. It should be possible to provide for more frequent periodic reporting of portfolio information without putting people in the position of being able to "front-run" the portfolio manager. <sup>162</sup>

## 18.3. Disclosure of Ownership Positions in Portfolio Holdings

With the increasing activism of institutional investors (both for their own account and for their managed accounts) there is a need to re-examine the continued existence of statutory and other exemptions from the "early warning" requirements and from insider reporting requirements with respect to the ownership of, or the exercise of control or direction over, securities of an issuer.

In an age where information translates into knowledge and power, these exemptions seem to be incompatible with ensuring that there is full, true and plain disclosure on a continuing basis in the marketplace about who owns, or exercises control and direction over, the securities of an issuer once a certain threshold is crossed. Technology makes it feasible to keep track of this information and to disclose it in a timely manner. Exemptions based on the passivity of an investor (as the current and proposed exemptions are based) are increasingly problematic, particularly with the increasing use of passive investing as an investment style.

These issues relate to the securities regulatory regime generally and not just to investment funds. They are of particular relevance to consumer/investors as the exemptions can operate to obscure for consumer/investors the significance of information about who owns, or exercises control or direction over, the securities of an issuer that are held in the investment portfolio of an investment fund.

Another contributing factor that makes it difficult to identify how concentrated ownership control and direction is relates to the fact that some holdings in managed accounts are not included in publicly available information as a result of the fact that there is not a common regime for money management

<sup>161</sup> These statements are currently required to be filed with regulatory authorities on a semi-annual basis and to be made available only to investors who request the same.

<sup>162</sup> Current provisions that permit non-disclosure of specific information regarding positions that are in the process of being accumulated or disposed of would remain in effect.

and that the holdings in some types of managed accounts are not subject to public reporting requirements.

A third factor that contributes to the obscuring of information about concentration of ownership, control and direction, is that under the current disclosure requirements relating to the disclosure of major portfolio holdings by investment funds in the securities of any one issuer or issuers, no disclosure of this information is made in documents that are actually delivered to investors.

The combination of these factors results in investors generally not being aware of the fact that:

- (i) the investment fund in which they invest holds a significant position or positions in securities of any one issuer or issuers; or
- (ii) the manager of their investment fund and/or its associates and affiliates owns, or exercises control or direction over, a significant position in any one issuer or issuers.

#### 18.4. **Voting Rights**

Another issue with respect to the ownership of portfolio securities relates to disclosure of how portfolio securities owned by an investment fund are voted. There is currently no requirement to disclose this information. There are also few requirements respecting such voting. 163

Given the increasing concentration of ownership, control and direction over securities by investment funds and other institutional investors, the voting of portfolio securities is a significant issue. This is particularly so as the marketplace increasingly recognizes the significant economic value that attaches to the right to vote.

Individuals who have chosen to pool their investments in a collective investment vehicle have given up one of the fundamental rights that flow from the ownership of securities - namely, the right to vote such securities. By pooling their investments, individuals have unwittingly conveyed their voting power and by doing so have placed enormous power in the hands of professional money managers, some of whom are not independent of other financial and commercial interests.

This is an area that has received little attention in Canada (and perhaps elsewhere).

As concerns about concentration of power, shareholder activism and the impact of the short-term focus of money managers, corporate managers and others grow, the voting of securities held in an investment fund is an area that can be expected to give rise to calls for change. Although these are issues that impact consumer/investors, the matters that they raise are beyond the focus of this Review. Nevertheless, I would like to highlight two matters pertaining to these issues that should be kept in mind.

The first relates to ensuring that there is no question that the economic value of the vote unequivocally belongs to the investment fund.

The second relates to the need to give some thought as to how to deal with voting rights attached to the securities held in the investment fund's portfolio. A starting point would be to enhance disclosure

<sup>163</sup> An example of one of the few requirements is found in Section 2.04 (1) (g) of National Policy No. 39 which provides that a mutual fund shall not without the prior approval of the securities authorities purchase securities for the purpose of

exercising control or management of the issuer of such securities.

requirements respecting the fund manager's voting policy. This would require a fund manager to articulate clearly in the fund's prospectus what the fund manager's policy is with respect to the voting of portfolio securities. If the fund manager deviates from its stated policy, the fund manager would be required to explain why in the annual and interim reports to investors. Depending on materiality, a timely disclosure announcement might be required to be made at the time when the voting rights that deviated from the fund manager's stated policy were exercised.

Disclosure requirements of this nature should be sufficient. Requirements to pass the votes through to unitholders would seem to be adding a degree of complexity and expense that is unwarranted.

## 18.5. Annual and Interim Reports

There is continued concern about the length of time it takes to issue year-end and interim reports. This concern was raised in the 1995 Report.<sup>164</sup>

The 1995 Report recommended that the annual report be issued within 60 days of the fiscal year end of the investment fund and that interim reports be required to be issued within 45 days of the end of the relevant interim period. Industry participants advised that these time delays were realistic in 1995 and that they would address the need for more timely information to be given.

With the advances in technology and reporting systems that have occurred since 1995, it is difficult to understand what the impediment is to implementing these recommendations.

In addition to the recommendations relating to shortening the time delays for the issuance of annual and interim reports, the 1995 Report contained several recommendations for enhancing the information contained in these reports. The importance of these measures in enabling consumer/investors (with or without the help of their advisers) to make sure their investments are aligned with their personal goals and objectives cannot be overestimated. Access to timely, enhanced information is an integral part of the investment process. The provision of this enhanced information in a timely manner should be high on the priorities of the investment fund industry in differentiating its "service offer" to investors.

#### 18.6. Expanding Public Access to the Internet

There is a need to explore ways of increasing the public's access to the disclosure information that is filed with securities authorities or that is otherwise made available to consumer/investors. As the use of technology increases, it is desirable to consider ways to ensure that every segment of the community is able to access the information in the marketplace. An interesting initiative is that of America Online in the United States that is targeted at parents and schools.<sup>166</sup>

Although the number of people with direct access to the Internet is increasing there are still a fair number of people who would benefit from facilities being provided in government offices, at various business sites and in public places such as libraries, community centers and schools. Increasing

\_

<sup>&</sup>lt;sup>164</sup> See Section 17.04 of the 1995 Report.

<sup>&</sup>lt;sup>165</sup> See Section 17.04 of the 1995 Report.

<sup>&</sup>lt;sup>166</sup> In August of 1998, America Online launched a "Sign Up Your School" campaign with the Family Education Network. The online service gives schools free websites. In announcing the campaign, AOL Chairman Steve Case invited every school district in the nation to take advantage of this free resource and get connected.

access to and enhancing Free-Net Internet service providers<sup>167</sup> is one means to broaden access to the information in the marketplace.

It is encouraging to note the significant efforts reflected in the federal government's "Connecting Canadians Initiative". This initiative, according to the advertisements regarding the same, is aimed at "helping Canadians to become the most connected people on earth, ready for the jobs and opportunities of today's knowledge-based economy".

## 18.7. Expanding Use of the Internet to Disseminate Information

Technology offers the means to reduce the information gap between those who know and those who do not know. So much of the regulatory focus has been on restricting or limiting the use of the Internet instead of on how to use it to bring about effective disclosure - i.e. timely, meaningful and relevant information that has been communicated to all investors - not just market professionals - and that is understood by them or at least is capable of being understood by investors who are reasonable persons acting reasonably.

Prospectuses, annual and interim reports, press releases, timely disclosure announcements, material change reports and material contracts should all be required to be available on publicly accessible websites. Facilities and standards need to be developed for the minimum length of time that the information remains posted and for archiving the information. Most industry participants with whom I spoke are supportive of enhancing the electronic availability of information. However, they would like to see relief from also having to provide paper copies of materials to those who acknowledge that they are satisfied with electronic access. This is a reasonable request.

The development of streaming media services such as Broadcast.com should be encouraged and enhanced in order to provide equal access to information for all.<sup>169</sup>

For example, some of these services allow the public to listen in on analyst calls although not the question and answer period that follows.

Consumer/investors need to be alerted to the availability of these services and of how to access them.

<sup>&</sup>lt;sup>167</sup> Free-Nets are supported by ongoing and one-time donations. For example, The Toronto Free-Net (according to its website) is supported by the ongoing donations of Rogers Shared Services, Ryerson Polytechnic University, Ontario Network Infrastructure Program, Pathway Communications and Free-Net users. The Toronto Free-Net was made possible by many one-time donations, including donations from 3com Canada Inc. and Bell Canada. The public libraries provide computer terminals through which The Toronto Free-Net may be accessed. Members can access The Toronto Fee-Net from their own computers as well. However, there are a limited number of access ports, limited capacity and limited speed.

<sup>&</sup>lt;sup>168</sup> The Toronto Stock Exchange is taking a leadership role in this respect. It has proposed guidelines for websites of listed companies. These guidelines go beyond current securities law and timely disclosure requirements. The main recommendation is that every listed company should maintain a corporate website to disclose investor relations information electronically in addition to the usual means of dissemination. The guidelines are available from the Market Regulation page of The Toronto Stock Exchange's website at <a href="http://www.tse.com">http://www.tse.com</a>. The comment period on the proposals expires October 31, 1998.

<sup>&</sup>lt;sup>169</sup> Broadcast.com and National Media Corp., have announced their agreement to webcast 24-hours per day live and to provide on-demand streaming video channels. As television and computers rapidly converge into a single box, services such as this should act as a catalyst in disseminating information of the nature described above "on demand" and to a very broad audience.

#### 18.8. SEDAR

The availability since 1997 of information on the SEDAR website has been a major step forward in making it easier for consumer/investors and for market participants to access certain market information. This facility should continue to be enhanced.

There are some commonly voiced concerns by industry participants about the number of problems they encounter when trying to file documents. If there are problems, it is important that they be identified and remedied. It serves no-one's interest to have filers complaining about the system while the operators of the system assert that there are no problems.

Other concerns relate to the difficulty that users who access the system have in finding the information that they are looking for. The difficulty relates to the fact that there is no attempt to isolate key documents such as the prospectus from extraneous documents such as the letters that accompany the filing of the prospectus.

Industry participants have suggested that there should there should be two levels of access to information. The first level of access would provide access to the key documents that have been filed about the issuer such as its prospectus, annual and interim reports, material change reports, timely disclosure announcements, press releases and material contracts. The second level of access would have all of the other information that is currently filed but is not of particular interest or relevance to consumer/investors.

Related to the above concern is the fact that the indexing system makes it difficult to find information about specific issuers. This problem is mostly encountered in the case of investment funds where the filed documents relate to more than one issuer. Consumer/investors would benefit if improvements in the indexing and cross-indexing search system were made.

#### 19. FUND GOVERNANCE

#### 19.1. Fund Governance

One of the most critical issues affecting consumer protection is fund governance. This subject has received little attention. Perhaps one reason is that the subject is abstract and intangible. There seems to be a certain lack of enthusiasm among a number of industry participants for focusing on fund governance.

The closest that many consumer/investors come to thinking about the subject of fund governance is to ask if their money is safe. Consumer/investors basically assume that somebody is "in charge" and is looking after things and that there is adequate regulatory oversight.

The reality is that there are relatively few requirements respecting the establishment and structuring of investment funds and there is little oversight of the ones that are established. The number of new investment funds has more than doubled in the last four years. There are now more investment funds in Canada than there are stocks listed on Canadian stock exchanges. Many of the new entrants to the investment fund management business have had little experience in investment fund management. Many of the investment funds are too small to be economically viable. The competitive pressures on

-

<sup>&</sup>lt;sup>170</sup> The seed capital for a publicly offered investment fund is \$150,000. Industry participants tell me that the break-even point for the costs and expenses of running an investment fund in Canada requires the minimum assets under administration to

fund managers have caused them to manage for sales. This has caused them to concentrate on the need to gain access to distribution in order to gather assets whereas at one time their focus was directed primarily to the prudential management of other people's money.

#### 19.2. The Establishment and Structure of Investment Funds

As noted above, the current regulations contain few requirements respecting the establishment and structuring of investment funds. <sup>171</sup> There is no requirement for the investment fund manager to be registered with any securities authority in order to establish and operate investment funds. In addition, there is no requirement for the investment fund manager to have "outside directors". There are no requirements for investment fund managers with respect to minimum regulatory capital, insurance and bonding coverage, management resources, competency and proficiency of personnel, internal systems, controls and procedures, or for procedures to monitor the adequacy and effectiveness of the internal systems, controls and procedures. In addition, there is no requirement for investment funds sponsored by the investment fund manager to have independent boards or to put in place other mechanisms designed to ensure that an independent perspective is brought to bear on transactions and operations. There is little discussion of this subject in prospectuses and there is little discussion of how conflicts of interest are managed and monitored. The little discussion that there is, is often buried in that part of the prospectus documentation that is not delivered to consumer/investors.

It is important to keep in mind that the manager and trustee of an investment fund are usually the same entity or are affiliates of each other. It is important to keep in mind that the manager, trustee and its affiliates provide all requisite services to the investment fund on terms that for the most part are unilaterally imposed. Coupled with this is the fact that most mutual funds in Canada are for tax reasons structured as trusts and there is no constating legislative structure in Canada for trusts that is comparable to the constating legislative structure that exists for corporations. The terms of the trust are unilaterally set by the manager-trustee.

One needs to consider the likelihood of prudential concerns arising in a structure that permits all of the functions that are required to be carried out in respect of an investment fund to be carried out by related parties on terms that in effect are unilaterally imposed without there being some degree of review by unrelated persons as to the manner in which the obligations are being carried out or without there being any measures to ensure the integrity of the product. The current structure results in there being no one who is considering the fairness aspects of the structure and of the transactions from the sole perspective of what is in the best interests of the investment fund and its investors.

be in the range of the \$250 - \$500 million. In the United States, the break-even point is estimated to be about US\$500 million.

<sup>&</sup>lt;sup>171</sup> The comments about investment funds relate in particular to open-end investment funds commonly known as "mutual funds". There are comparable concerns respecting other types of investment funds which should also be looked at and which hopefully will be the subject of further work that is done in this area.

<sup>&</sup>lt;sup>172</sup> It is unusual for there to be any independent counsel involved or for there to be any independent committee of the board involved.

<sup>&</sup>lt;sup>173</sup> I have recommended that an Investment Funds Statute be enacted for this purpose. See Section 18.02 of the 1995 Report. Also see the Remarks of Glorianne Stromberg to The Canadian Institute - OSC Reformulation Conference - The Mutual Fund Project Panel Discussion, The King Edward Hotel, Toronto, Canada, January 20 and 21, 1997.

<sup>&</sup>lt;sup>174</sup> See Footnote 172.

There are specific recommendations in the 1995 Report aimed at improving the governance provisions in respect of investment funds. The recommendations in the 1995 Report include those set out in Section 14 relating to the Management of Investment Funds, those contained in Section 18 relating to the Establishment and Governance of Investment Funds, those contained in Sections 19 - 22 relating to Conflicts of Interest, Execution of Portfolio Transactions, Soft Dollars and Fund of Funds, those contained in Sections 23 - 27 relating to Money Market Funds, Performance Information, Investment Objectives, Policies and Restrictions - National Policy No. 39, Custodianship of Securities - National Policy No. 39 and Miscellaneous Matters - National Policy No. 39.

Very few of these recommendations have been implemented although I understand that some fund groups have reviewed and modified their operations and procedures with these recommendations in mind. I also understand that there is a reluctance by some fund groups to take any action until it is mandated by regulators due to concerns about perhaps being at a competitive disadvantage or perhaps exposing themselves to liability if they should fail to meet self-imposed higher standards than what is the standard in the marketplace.

The need to implement the recommendations aimed at improving the governance provisions in respect of investment funds remains.

## 19.3. Trustees, Directors and Officers of Investment Funds

There are concerns that trustees, directors and officers of investment funds often lack the necessary competencies to enable them to carry out their role effectively.

There is a need to address these concerns and to identify the competencies and training needed by people at every level of the investment fund's operations in order for them to do their job. Knowledge and awareness is a critical tool to the achievement of optimal results.

Concerns have also been raised about the soundness of the trust arrangements and the fact that trustee(s) are not independent even in situations where they purport to be. The use of "bare trustees" (especially when a licensed trust company performs this function and leaves the public with the impression that there is substantive independence) is a situation of concern. Critics of the practice of using bare trustees question how sound the trust arrangements are when the real control and direction lies with the management companies or with one or two individuals.

They question the soundness of the custodial arrangements in the absence of effective internal systems, controls and procedures that provide the necessary checks and balances to ensure that the instructions given to the custodian are proper instructions. They question what controls are in place to ensure that the custodian has received all of the property and assets of the investment fund for safekeeping. They question how effective the arrangements are for establishing trust accounts. They question the adequacy of the controls and oversight for ensuring that timely and proper settlement of portfolio transactions occurs.

These questions all relate to matters that people assume are adequate until they find out that something more should have been be done. All of these questions are related to the issue of consumer protection. More work should be done in this area without delay to establish standards for prudent operating procedures that must be adhered to for all investment funds.

## 19.4. Adequacy of Internal Controls, Systems and Procedures

The adequacy of internal controls, systems, procedures and resources (human, technical and financial) are key factors in the governance review. The tendency to rely on outsourcing as a substitute for inhouse expertise is an area that requires special attention.<sup>175</sup> It is submitted that in the case of outsourcing, there at least needs to be sufficient in-house capability to monitor and oversee the adequacy of the services being provided and, in particular, to evaluate whether investment portfolios are being managed in accordance with their stated investment objectives and risk profiles.

It is also submitted that outsourced activities should be required to be performed to at least the same standards as would be required if the services were not being outsourced. An example of such a standard would be the requirement for third party service providers (including portfolio managers) to adhere to a code of conduct with respect to personal investing activities that is at least as comprehensive and stringent as would be applicable if the investment fund manager were providing the services directly. This is an area where the development of high international standards would help immensely.

## 19.5. Allocation of Costs and Expenses

One of the concerns that has been raised about the recommendations in the 1995 Report about fund governance is that they will add costs and that these costs will be passed on to the consumer/investor. In the current environment, this is probably true.

However, the fact that this is so raises the question of whether enough scrutiny is being given to the allocation of costs and expenses between the manager and its managed funds? There appears to be a tendency simply to pass on costs rather than to take a good hard look at operations to see if it is appropriate to do so or to see if there are not more efficient ways of doing things for less cost.

For example, when IFIC released its proposals for a model code of ethics for personal investing a variety of people observed that the manager's cost of compliance would be passed on to the unitholders of the funds. No one appears to have asked why the unitholders of the funds should have to pay extra to ensure that the manager of their funds is fulfilling its fiduciary responsibilities to them and that neither it nor its personnel is appropriating for their own account investment opportunities that belong to the funds.

No one appears to be asking how much of the manager's overhead expenses should be allowed to be passed on to the unitholders of the funds. No one appears to be asking how much of the manager's costs of distribution should be allowed to be passed on to the unitholders of the funds. No one appears to be measuring the impact of portfolio transactions and their costs on the investment funds. No one appears to be measuring whether the overall risk profile of the investment fund has been increased to boost performance to compensate for higher costs and expenses. No one appears to be reviewing valuations (except in the most extraordinary situations) to ensure that the securities owned by the investment fund could actually be sold for their carrying cost. Behind all of these questions are other questions.

\_\_\_

<sup>&</sup>lt;sup>175</sup> A related area that requires attention is when fund management groups who lack the requisite internal expertise purport to rely on the expertise of third party experts for portfolio management but in fact make decisions that affect the fund's portfolio.

<sup>&</sup>lt;sup>176</sup> It is generally acknowledged in the industry that at least 1% of the management expense ratio relates to distribution costs.

There are many experienced, reputable fund managers who do have procedures in place which they are continually monitoring and updating to deal with these issues. But there are no industry standards in place and competitive pressures sometimes result in blurred judgment being brought to bear on the issues.

All of these direct and indirect costs that are borne by the investment funds are reflected in their management expense ratios and their net asset values. These costs reduce the amount of the investor's return. Compounded over a period of years, these amounts substantially erode the amount that an investor will have to meet retirement and other needs. Therefore, these questions are not frivolous ones to raise. The answers have an impact on the ability of consumer/investors to provide for their well-being and retirement income. They have an impact on the shortfall that governments may ultimately be called upon to provide.

#### 20. INVESTMENT ALTERNATIVES

#### 20.1. Investment Alternatives

The environment of the last several years has focused attention on, and fostered the development of, a variety of investment alternatives. These investment alternatives seek to combine the advantages of pooling money for investment purposes with other features such as guarantees of capital and/or income, insurance, death benefits, protection from creditors, exemptions from probate fees, market-linked retirement income streams, relief from limits on foreign investments that apply to retirement income plans and relief from fund-of-fund investment restrictions under applicable securities laws.

The array of investment alternatives that are discussed in this Section 20 includes conventional mutual funds, closed-end investment funds, segregated funds offered by insurance companies, segregated funds offered by mutual fund organizations with an insurance wrapper, protected or guaranteed mutual funds, 177 index-linked or market-linked universal life policies, other types of variable investment contracts and annuities, index-linked guaranteed investment certificates, index-linked, commodity-linked or mutual fund-linked debt obligations, pooled funds that are offered on a private placement basis, pooled funds that are offered pursuant to exemption orders by securities regulatory authorities, certain types of pension and other retirement plans and certain types of wrap accounts offered by various intermediaries.

There are some who view the development of these investment alternatives as being very positive for consumer/investors. There are others who see the development as less than positive.

## The Positive Perspective

Those who view the development as positive for consumer/investors observe that never before have consumer/investors had such a variety of investment alternatives from which to choose to meet their needs.

## The Negative Perspective

Those who view the development of these investment alternatives as being less than positive for consumer/investors say that these arrangements are very complex. They are hard to understand. They

\_

<sup>&</sup>lt;sup>177</sup> In Canada, the term "protected or guaranteed mutual funds" refers to mutual funds that are offered in combination with stipulated insurance benefits but which are not "segregated funds" within the meaning of insurance legislation.

are instruments for self-destruction. Their very complexity leads to abuse. The more complex the arrangements are, the more games that can be played. There is no assurance that they will work as they are intended to work. They are aggressively marketed. Inadequate due diligence has been conducted to determine whether the arrangements are inherently suitable to achieve their purpose. They are simply a way for end suppliers to make more money. The "pro" and "con" lists that are given to consumer/investors are usually prepared by people who are offering alternative investment arrangements in competition with other investment arrangements, and this calls into question the objectivity of the comparisons.

## My Approach

Whether one views these investment alternatives positively or negatively is no longer the issue. The reality is that these investment alternatives exist. Substantial amounts of money have been and are continuing to be invested in them. The issue, as I see it, is to recognize that there are substantive differences among these investment alternatives, to identify those differences that give rise to substantive concerns and to address these concerns as best as one can. This is what I try to do in the remaining subsections of this Section 20.

The substantive differences that give rise to substantive concerns relate primarily to:

- (i) the suitability of the particular products to meet the specific needs of individual consumer/investors; and
- (ii) the level of prudential and regulatory supervision and oversight over the structuring of these products and the offering of the same to consumer/investors.

#### 20.2. Substantive Concerns

#### Lack of Common Standards

There is a lot of concern about the lack of common standards relating to the offering of the various investment alternatives to consumer/investors. The disclosure requirements, the registration requirements, the know-your-client/suitability obligations, the sales and business practice standards, the education and proficiency requirements, and the degree of regulatory oversight and supervision all vary depending on the type of investment alternative, the issuer of the investment alternative and who is offering the investment alternative for sale.

Of prime concern is the fact that some investment alternatives (such as segregated funds, index-linked or market-linked universal life policies and other variable annuity contracts) are excluded from the definition of "securities" under securities legislation. These investment alternatives are securities within the plain meaning of the term. However, they are not regulated under securities legislation. Nevertheless these "excluded" investment products are offered to consumer/investors as being the functional equivalents of investments that are "securities" within the meaning of securities legislation and are subject to the prospectus, registration and continuous disclosure requirements of securities legislation. These "excluded" investment alternatives, do not have the protection that the securities regulatory regime is designed to offer to consumer/investors who invest in securities.

Similarly, index-linked guaranteed investment certificates are exempt securities under securities legislation. These securities are complex yet they are offered to the public without the benefit of the

\_

<sup>&</sup>lt;sup>178</sup> To the extent that they are regulated, they are regulated under insurance legislation.

prospectus and registration requirements of securities legislation that apply when securities are offered to the public.

The lack of a common regulatory regime, together with the lack of common standards for investments and investment alternatives, increases the risk of consumer/investors being offered investments and other products that may not be suitable, or be the best match, for their respective needs.

The lack of a common regulatory regime and the lack of common standards respecting disclosure requirements, registration requirements, know-your-client/suitability obligations, sales and business practice standards, education and proficiency requirements, and regulatory oversight and supervision for investments and investment alternatives places some segments of the financial services industry at a competitive disadvantage. This is as a result of some of these investments and investment alternatives and some of the firms offering them being subject to more regulation than other investments, investment alternatives and firms are.

## Failure to Appreciate Substantive Differences

Given the variety and complexity of the investments and investment alternatives that are available in the marketplace, it is unlikely that consumer/investors (and perhaps even some of their advisers) appreciate the substantive differences that exist among the different investments and their alternatives. It is also unlikely that they appreciate the legal implications that flow from these differences.

For example, there is a fundamental legal difference between:

- (i) having a direct ownership interest in mutual fund securities, and
- (ii) having a contractual right to certain payments, the amount of which is determined according to the value of certain securities<sup>179</sup> that are owned by the contracting insurance company rather than the consumer/investor.

In the latter case, the payments to which the consumer/investor is entitled depend on the ability of the insurance company to meet its contractual obligations as they become due. This dependence on the credit of the insurance company introduces a credit risk factor relating to the insurance company that may not be readily apparent to the consumer/investor and that most consumer/investors have no means of monitoring.

In considering the ability of the insurance company to meet its contractual obligations as they become due, there are several factors that impact on this. These factors are less than obvious to consumer/investors and their advisers. They include

- (i) the sufficiency of the actuarial reserves;
- (ii) whether the insurance contract is a "policy of insurance" as defined in applicable insurance legislation which would be covered by the insurance compensation fund;
- (iii) whether the insurance contract is one that requires assets actually to be set aside and segregated from the general assets of the insurance company;
- (iv) the sufficiency of the segregation of assets to match the individual insurance contracts; and

<sup>&</sup>lt;sup>179</sup> These securities may be, or include, mutual fund securities.

(v) the sufficiency of reinsurance arrangements.

Another factor that may not be readily apparent is that if consumer/investors decide that there is reason to be concerned about the ability of an insurance company to meet its contractual obligations as they become due, there are usually substantial surrender fees that they are required to pay if they should decide to terminate their insurance contracts prior to the maturity date.

#### Restrictions and Preferences

Many of the investment products that have been developed in recent years have been designed to overcome certain investment restrictions or to take advantage of certain preferences.

The desire to convert "foreign property" (as such term is defined in income tax legislation) into Canadian property has been behind the development of many innovative investment products.

An example is the issue of mutual fund-linked debt obligations of Canadian banks. The investment returns of these debt obligations are linked to the performance of mutual funds that are "foreign property" for tax purposes.

Another example is the issue of the segregated fund products known as "guaranteed investment funds" by insurance companies where it was intended that despite the inclusion of underlying mutual funds that are "foreign property" for tax purposes, the segregated fund would not be considered to be "foreign property" under the Income Tax Act (Canada)..

These innovative products usually come with costs for consumer/investors that are not readily apparent to them. Sometimes these products also come with added risks that are not readily apparent to consumer/investors (or their advisers). This is what happens when a segregated fund contract is bought as opposed to investing directly in the mutual fund on which the segregated fund contract is based. When the costs, risks and alternatives are not readily apparent to consumer/investors, they do not address the question of whether the benefits of the investment product outweigh the costs and the added risks of the investment product having regard to their own particular situations.

The advantages conferred on certain insurance products under the Bankruptcy and Insolvency Act (Canada) is one of the factors that has led to the increased popularity of segregated funds. The "creditor-proofing" that a segregated fund offers provides a substantial competitive advantage to insurance companies over other investment products that are functional equivalents of segregated funds. However, the desire of consumer/investors for "creditor-proofing" has resulted in consumer/investors choosing an investment product that brings with it higher costs and risks that may not be apparent.

Another advantage that segregated funds and market-linked universal life polices enjoy that mutual funds do not relates to exemptions from probate requirements under provincial laws. This matter is discussed in Section 23 of this Review.

## Confusion Between Mutual Fund Offerings and Segregated Fund Offerings

The introduction by mutual fund organizations of segregated fund offerings of their mutual funds with an insurance wrapper (which offerings are regarded as insurance contracts under applicable insurance legislation) has the potential for confusing consumer/investors.

The names of the funds are the same and the advertisements often do not clearly distinguish whether they relate to a mutual fund product or to a segregated fund product. The increased use of brand advertising has increased the opportunity for confusion.

Consumer/investors (who according to some surveys often do not even know what mutual fund they bought) may not realize that there is a difference between buying the mutual funds directly and "buying" the same mutual funds in a segregated fund. They may not become aware of the differences in their rights until later on when it may be too late to do anything about it.

There are similar opportunities for consumer/investors to be confused in the case of segregated fund offerings by insurance companies that use existing "brand name" mutual funds or clones thereof to form the segregated fund pool to which the insurance contract is linked.

#### Sales Practices and Incentives

Industry participants have observed that some industry participants have adopted the view that the sales practices rule<sup>180</sup> does not cover segregated funds and that they are free to adopt sales practices and offer sales incentives in respect of segregated fund transactions that they would be prohibited from doing if the transactions were mutual fund transactions. As a result, some people are selling segregated funds in preference to mutual funds so as to avail themselves of added sales incentives and benefits.

There is a similar problem with some pooled investment funds that are offered pursuant to a statutory exemption under securities legislation or an exemption order issued by securities regulatory authorities. Consumer/investors invest in these funds through so-called "group retirement savings plans" or through the selection of investment options under defined contribution pension funds. I understand that in some cases, substantial finders' fees are paid to intermediaries and/or their representatives in addition to other sales incentives and that in some cases consumer/investors are not aware of these payments.

#### Other Differences

Some investment fund alternatives may involve more risk than is apparent. Some of these may add costs, some of which may be excessive for the services or assurances provided. Some investment fund alternatives may not offer the investment flexibility that consumer/investors need if their investment goals and objectives or their time horizons change. Some investment fund alternatives have different tax consequences.

Often, consumer/investors do not realize that there are other investments, either alone or in combination with other investments or products, that provide suitable investment alternatives at a more reasonable cost. They often do not realize that if they were to invest in these investments and other products they would increase their total return on their investments and the result would be a significant difference in their ability to meet their investment goals and objectives.

It is beyond the scope of this Review to analyze the substantial differences that exist between the different investment fund alternatives that are available in the marketplace or to try to address their suitability for the differing needs of consumer/investors. I recognize that the foregoing discussion barely touches on the issues. However, it should be sufficient to highlight the need for some changes to be made, for some more work to be done and for consumer/investors to gain a better understanding of issues that they should be considering in implementing their goals and objectives.

Subsection 20.3 of this Review outlines recommendations for addressing the substantive concerns that have been discussed.

-

<sup>&</sup>lt;sup>180</sup> See subsections 3.4 and 16.8 of this Review.

## 20.3. Recommendations to Address Concerns

# Common Regime for Money Management and Integrated Regulatory Structure

The early part of this Review discusses the need for a common regime for money management and for changes in the regulatory structure designed to implement the regime. The implementation of these recommendations would be of significant benefit to consumer/investors.

These recommendations would also "level the playing field" and remove the competitive disadvantages that flow from the fact that the various investment funds and investment fund alternatives are not subject to comparable regulation with respect to the structuring, offering and oversight thereof.

The implementation of the recommendations for a common regime for money management and for changes in the regulatory structure designed to implement the regime would result in all investment fund alternatives being subject to common requirements relating to their structuring and offering and would eliminate the different treatment (and lack of standards) for so-called "exempt" securities or offerings. 182

## Increasing Knowledge and Awareness

A heavy emphasis needs to be put on educating consumer/investors and their advisers about the substantive differences among the different investment fund alternatives and their suitability. Suggestions for enhancing the knowledge and awareness of consumer/investors and their advisers are contained in Sections 14 and 15 of this Review.

#### Effective Disclosure

Common standards should be established aimed at providing consumer/investors with full, true and plain disclosure of the material facts about the respective investment products. The recommendations respecting disclosure in the earlier parts of this Review should be taken into account in setting common standards.<sup>183</sup>

The intention is not to add regulatory costs that will ultimately be borne by the consumer/investor but to find a means to provide consumer/investors with effective, meaningful disclosure that will enable them to make reasoned decisions about how to use the different types of investment alternatives to implement their investment objectives and goals. One needs to look at whether there are alternatives to prospectus filing requirements for all investment fund products that would work equally well or better.

\_

<sup>&</sup>lt;sup>181</sup> See subsection 12.2 of this Review.

Consumer/investors who invest in pooled funds that are offered pursuant to a statutory exemption under securities legislation or an exemption order issued by securities regulatory authorities would benefit in particular if these pooled funds were required to meet common standards for the structuring of investment funds and for disclosure of information about them. See the comments under the subheading, "Effective Disclosure."

<sup>&</sup>lt;sup>183</sup> See Sections 17 and 18 of this Review.

In the case of segregated funds and index-linked or market-linked universal life policies, there is a special need to ensure that the risk management policies with respect to both the death and maturity guarantees, including the crystallization of gains, are adequately priced for and disclosed.<sup>184</sup>

In the case of these insurance contracts, the ability to amend the contracts unilaterally in a way that would negatively affect the policies should be prohibited. Alternatively, consumer/investors should be given the opportunity to cancel the policies without penalty and with the maturity date of the policy being accelerated to coincide with the date of cancellation so that surrender fees or termination fees will not be payable.

One of the common standards should relate to the disclosure of how income and realized capital gains and losses are taxed. Some consumer/investors do not realize, for example, that the gains they receive in respect of an index-linked or market-linked guaranteed investment certificate are regarded as a receipt of income for tax purposes and not as a capital gain.

#### Prudential Enhancements

Insurance legislation should be reviewed and changed where necessary to ensure that assets that back the obligations to make market-linked payments are segregated in trust to secure those payments. I am told by industry participants that this is of particular concern in the case of market-linked universal life policies.

The sufficiency of capital requirements, actuarial reserves and related matters needs to be dealt with without delay. I understand that the industry and its regulators have been aware of the concerns in this area for some time. Consumer/investors might well question the prudence of permitting additional segregated fund and market-linked universal life polices to be sold without these matters first being addressed.

There may well be other matters of a prudential nature that require changes to be made. A task force with the necessary expertise should be assembled without delay to review the matter.

#### Removal of Certain Restrictions and Preferences

There have been many submissions to the Government of Canada advocating the removal of the "foreign property" restrictions in income tax legislation or, alternatively, increasing the foreign property limits. The "foreign property" investment limits appear to have served whatever purpose they had. They are now readily able to be avoided and are being avoided. However, doing so adds costs for consumer/investors and exposes them to additional risks. It would seem to be time to remove these restrictions and to let the market regulate the level of foreign content in the portfolios of consumer/investors.

There have also been various attempts made to extend "creditor-proofing" to all retirement income plans rather than just to those that are insurance contracts. It would seem to be time for the federal and provincial governments to do this. This extension of "creditor-proofing" to all retirement income plans would serve as a key enabling mechanism under the governments' retirement policy for Canadians as it would assist consumer/investors in setting aside sufficient "creditor-proofed" retirement funds to provide for their retirement income.

<sup>-</sup>

<sup>&</sup>lt;sup>184</sup> See PriceWaterhouseCoopers, Insurance Review, July, 1998, Segregated Funds, for a useful discussion concerning financial disclosure recommendations, compliance issues, system requirements, tax considerations and risk management challenges related to segregated funds.

Recommendations relating to probate requirements are set out in Section 23 of this Review.

#### 21. ONGOING RELATIONSHIPS BETWEEN INTERMEDIARIES AND CONSUMER/INVESTORS

## 21.1. Switching and Churning

Industry participants tell me that there is substantial switching and churning of accounts going on.<sup>185</sup> Two situations in particular appear to be giving rise to this. Both situations relate to mutual fund securities that have been sold on a deferred sales commission basis.

The first situation relates to the permitted ten per cent per annum redemption of mutual fund securities that may be redeemed free of any deferred sales commission or redemption fee.

I understand that some sales representatives are advising their clients (or using powers of attorney that they have obtained from their clients) to redeem these "free shares" and to reinvest the proceeds either:

- (i) in the same fund on a zero front-end sales commission basis; or
- (ii) in another fund on a deferred sales charge basis.

Investing the redemption proceeds on a zero front-end sales commission basis results in the doubling of the ongoing trailing commission that the intermediary receives.

Investing the redemption proceeds on a deferred sales commission basis results in:

- (i) a new deferred sales commission period starting with respect to the mutual fund securities that are acquired with the redemption proceeds; and
- (ii) the intermediary receiving an immediate lump sum sales commission that is paid by the fund's manager plus an ongoing trailing commission.

When these transactions occur in a registered plan account such as a retirement savings plan or a retirement income fund, there are no income tax consequences to the client. However, if the transactions occur in a non-registered plan account, the transactions are a "disposition" for income tax purposes and give rise to a realized gain or loss that the client must take into account for tax purposes.

I have been told that in some cases clients are not even aware that the transactions have occurred. If this is so, it creates additional problems for clients whose securities are held in a non-registered plan account.

The second situation relates to the expiration of the period during which redemptions of mutual fund securities would give rise to a deferred sales commission. Industry participants tell me that as in the first situation, some sales representatives are advising their clients (or using powers of attorney that they have obtained from their clients) to redeem these mutual fund securities and to reinvest the proceeds either in the same fund on a zero front-end sales commission basis or in another fund on a

123

<sup>&</sup>lt;sup>185</sup> The discussion in Section 21 of this Review is confined to situations dealing with mutual funds. Industry participants have spoken with me about the switching and churning which they say goes on in the case of certain market-linked universal life policies. I have not, in the time constraints applicable to this Review, been able to follow-up on this to gain a better understanding of what is involved or the magnitude of any problems that exist.

deferred sales commission basis. The consequences of this action are the same as those described above in respect of the first situation.

Another consequence of the switching and churning transactions is the added record-keeping and administrative work that mutual fund management organizations perform for distributors to keep track of deferred sales commission transactions and the payment of ongoing trailing commissions. This work adds costs and is another contributing factor to the increasing costs of investing that are borne by consumer/investors.

Although individuals associated with the "manufacturers", the "distributors" and their respective service providers know what is going on, they remain silent. This silence and these situations are examples of how the competitive pressure on "manufacturers" to increase sales and access to distribution, and the competitive pressure on "distributors" to retain top performers have an impact on what happens to consumer/investors.

It is situations like these that undermine the confidence and trust that consumer/investors are encouraged to place in the integrity of the investment fund industry. Here the work of the relatively few undermines the work of the many in an industry that strives hard to merit the confidence and trust of consumer/investors.

## 21.2. Recommendations to Address Switching and Churning

There is no simple way to distinguish bona fide changes in investments from those that are not. There is also no simple way to prevent the abuse of trust that results when switching and churning occurs. I have set out some suggestions below that might help address the problems by keeping them from occurring in the first place.

## Systemic Changes - Education and Regulatory Structure

The recommendations made earlier in this Review relating to enhancing the knowledge and awareness of consumer/investors and industry participants should help.<sup>186</sup> In the case of consumer/investors increased knowledge and awareness should result in a more aware, questioning client. In the case of industry participants the enhanced knowledge and awareness of their fiduciary obligations and the consequences of breaching them should help to reduce the incidences of abuse of trust.

A regulatory system and structure that was aligned with advisory-based relationships as opposed to transaction-based relationships should also help.<sup>187</sup>

## Systemic Changes - Compensation

Commission-based compensation is at the root of the problems relating to switching and churning. It is also often at the root of most of the problems relating to suitability issues. This is so because as long as there is a product-based differential in the compensation that a sales representative or adviser will earn, the question will always arise as to whether the product that has been chosen is the best one for the client or the best one for the intermediary. Differences in sales commissions and the opportunity to

.

<sup>&</sup>lt;sup>186</sup> See Sections 14 and 15 of this Review.

<sup>&</sup>lt;sup>187</sup> See Sections 12, 13, 14, 15 and 16 of this Review.

generate new commission payments have long been responsible for many investment changes that are made <sup>188</sup>

The recommendations made earlier in this Review relating to the need to align the compensation structure so that it is more appropriate for an advisory-based relationship than for a transaction-based relationship should serve to remove or reduce any incentive or advantage to switching and churning.

In this respect, some firms have adopted the practice of not charging commissions on switches. Instead, a low flat administrative fee is payable. The internal compensation plans of these firms are also designed to discourage switches.

#### Switch Disclosure Document

Industry participants have suggested that the best method to deal with switching and churning is to require that a switch disclosure document be completed. They note that this type of document is used in the life insurance industry when policyholders surrender one policy and acquire another.<sup>189</sup>

The mandatory use of a switch disclosure document should help to focus the attention of consumer/investors on the proposed changes in their investments. This document should clearly set forth:

- (i) the change in investments that is being made;
- (ii) the reason for the change;
- (iii) the costs of the change; 190
- (iv) the income tax consequences of the change with an estimate of the amount of income tax that will be payable; and
- (v) any other information that is needed in order to illustrate the impact that the proposed changes will have on the consumer/investor.

The switch disclosure document should be required to be signed by the client in question, the sales representative and by the appropriate supervising officer(s) of the intermediary with whom the sales

<sup>&</sup>lt;sup>188</sup> Industry participants have told me about attending "road shows" to launch new investment funds and sales meetings where the people attending are urged to review their client records to identify how much "free room" there is to switch investments without incurring sales charges. The opportunity to earn extra commission and to be eligible for additional cooperative marketing and educational conference support is used as a motivating force to persuade sales representatives to redirect client investments.

<sup>&</sup>lt;sup>189</sup> I have been told that one of the reasons contributing to the effectiveness of the insurance industry's switch disclosure document to reduce unsuitable transactions is that it is a "chore" to prepare the document and when the client sees the impact of the changes, the client often decides not to proceed with the transaction.

<sup>&</sup>lt;sup>190</sup> These costs would include sales commissions or charges that are:

<sup>(</sup>i) payable directly or indirectly by the consumer/investors:

<sup>(</sup>ii) payable either immediately or at a later time; and

<sup>(</sup>iii) payable either on a lump sum basis or a continuing trailing basis.

The costs would also include any transaction or account fees and any other fees and charges that are payable directly or indirectly by the consumer/investor in respect of the transactions. The required information would relate to both the redemption and the reinvestment of the redemption proceeds.

representative is associated before the transactions are processed. Powers of attorney granted to sales representatives or to the intermediary should not be permitted to be used to eliminate the requirement for the client's signature. The client should receive a fully executed copy of the switch disclosure document.

In determining the situations where a switch disclosure document is required to be used, some thought needs to be given to ensuring that the requirements include transactions that may be designed to avoid the requirements. An example of where this might happen is if there is a delay between the redemption and the reinvestment of the redemption proceeds. There are other practices that may need to be addressed such as "pre-signed" approvals by supervising personnel.

The proposals for requirements for a switch disclosure document are designed to discourage inappropriate advice being given to consumer/investors. They are not designed to discourage bona fide investment changes.

The switch disclosure document should be a key compliance tool to assist all intermediaries to exercise oversight and supervision over the persons associated with the firms. To this end, there is no reason why intermediaries could not begin using such a document immediately. The use of the document need not await the promulgation of regulatory or self-regulatory requirements. However, ideally, the minimum requirements for the switch disclosure document would be mandated as a condition of registration and/or membership in a self-regulatory organization.

In the interim, consumer/investors should ask for a switch disclosure document that sets out the above information before agreeing to a switch of the nature described above.

## **Enhanced Supervision and Oversight Procedures**

Enhanced supervision and oversight procedures would assist in identifying and dealing with switching and churning. The switch disclosure document referred to above is one tool. There are other review mechanisms that could be used. What is needed is an impetus to motivate intermediaries to use these mechanisms.

While the best impetus may be the demands of consumer/investors, the reality is that most consumer/investors are not aware that they may be receiving inappropriate advice. Therefore the motivation needs to come from the regulators and the self-regulators. They should take a pro-active role in setting standards and in monitoring compliance with the standards.

They need to recognize that high standards that are implemented on a voluntary basis can place intermediaries at a competitive disadvantage and result in sales representatives leaving to join less-principled firms. The role therefore of the regulators and self-regulators is to ensure that all players in the industry are required to have, and to meet on an ongoing basis, the same high standards.

## 21.3. Registration of Securities and Powers of Attorney

An area of vulnerability for consumer/investors relates to how the securities that they own are recorded in the records of the intermediary with whom they deal and in the records of the investment funds that they own. The vulnerability arises primarily in the event of the insolvency of the intermediary.

Securities owned by clients of an intermediary are generally recorded either in the client's name or in the name of the intermediary. Securities that are recorded in the client's name are commonly referred to as "client-name securities". Securities that are recorded in the name of the intermediary are commonly referred to as "nominee-name securities".

Although there is no hard and fast rule:

- Firms that are full-service investment dealers generally record the securities owned by their respective clients as nominee-name securities. Not all of these firms are members of a selfregulatory organization (such as the IDA) and of the CIPF.
- Firms such as mutual fund dealers usually record the securities owned by their respective clients as client-name securities. These firms are not a member of a recognized self-regulatory organization and are not a member of the CIPF.
- Some full-service investment dealers allow client-name securities registrations as it facilitates keeping track of all transactions for a client's account "on-book" and some mutual fund dealers have nominee-name registrations because this gives them control and direction over their clients' accounts.

Where nominee-name registration is used, it is of the utmost importance to consumer/investors that the firms in question have the appropriate controls and procedures in place to protect their clients against fraud and the firm's insolvency. It is important that there be strong, effective internal and external oversight over operations. These factors are often missing in the case of firms that are not members of a self-regulatory organization and of the CIPF.

Where securities are recorded in the client's name, a practice seems to have developed of obtaining a power of attorney from the client in favor of the sales representative or the intermediary that enables transactions involving client-name securities to be effected for the client's account without requiring the client's signature. Depending upon the terms of the power of attorney, the client's express authorization for the transaction may not be required to be obtained prior to effecting the transaction.

The potential for the misuse of powers of attorney is great. Of particular concern is the potential use to redeem securities from clients' accounts and to use the proceeds to make fraudulent payments. Another potential for misuse is described above in Section 21.1 and involves the use of powers of attorney to effect switch and churn transactions.

The practice of obtaining powers of attorney creates another less obvious problem for consumer-investors in the event of the bankruptcy or insolvency of the intermediary with whom they deal.

This problem results by reason of the fact that the existence of powers of attorney that enable client-name securities to be dealt with by a dealer or its representative may be considered to "invade" the client-name securities in the event of the insolvency of the dealer. As a result there is concern that the client-name securities covered by the powers of attorney will not be considered to be "client-name securities" and, as such, available for immediate distribution to the clients entitled thereto. Instead it is believed that these securities would likely form part of the "customer pool securities" under the Bankruptcy and Insolvency Act (Canada). In this case there is likely to be considerable delay before clients will be able to take possession of and deal with the securities in their accounts. Another consequence is that in the event of shortfall, clients whose securities form part of the "customer pool securities" may not receive "one hundred cents on their dollar". <sup>191</sup>

\_

<sup>&</sup>lt;sup>191</sup> The provisions of the Bankruptcy and Insolvency Act (Canada) relating to the insolvency of securities firms are new additions to the Bankruptcy and Insolvency Act and they are complex. To date, there is little judicial guidance as to how the provisions will be interpreted. It is beyond the scope of this Review to try to deal with the ramifications of the provisions of the Bankruptcy and Insolvency Act other than to highlight certain problems that the provisions may create depending upon how

There can also be problems for clients in the event of the bankruptcy of the intermediary in whose name the securities are recorded where the clients' securities are held as nominee-name securities.<sup>192</sup> These problems are usually minimal if the intermediary is a member of a self regulatory organization, such as the IDA, that requires the intermediary to participate in the CIPF.<sup>193</sup>

As noted above, there are also problems in the event of the bankruptcy of the intermediary even when the clients' securities are held as client-name securities if the intermediary is able to negotiate these client-name securities pursuant to a power of attorney or other means.

Those who favor nominee-name registrations argue that in the event of the insolvency or bankruptcy of an intermediary, clients are better off being in the "customer pool" under the Bankruptcy and Insolvency Act (Canada) because if fraud is involved and there is a shortfall, the impact will be spread across the broader pool of client accounts rather than borne by specific individual client accounts and that losses on a per client basis are likely to be less.<sup>194</sup>

Apart from this arguable benefit in the event of the intermediary's insolvency or bankruptcy, the reasons given for advocating nominee-name registration relate primarily to the ease of:

- (i) effecting delivery in the event clients want to sell the securities;
- (ii) providing collateral for margin accounts or other loans; and
- (iii) safekeeping.

Clients are often discouraged from taking delivery of certificates evidencing their ownership of securities. To do so is expensive, particularly when so many securities are held in a book-based system. The systems used by full-service dealers are usually designed around their clients' securities being held as nominee-name securities. It would require substantial investments by full-service investment dealers to upgrade their systems to accommodate client-name registrations even though that type of registration may be preferred and systems now exist to support such registration.

client securities are registered and whether the granting of powers of attorney results in the client-name securities being considered to be in negotiable form.

<sup>&</sup>lt;sup>192</sup> A case in point relates to the insolvency in the Spring of 1998 of Vantage Securities Inc. ("Vantage"). This insolvency and the impact on clients has been the subject of numerous press articles. Vantage is not a member of the IDA and it is not a member of the CIPF. Vantage's insolvency has raised many questions about basic industry practices and the impact of these practices on clients' property and rights in the event of the insolvency or bankruptcy of a dealer. One of the areas of concern relates to the administration of self-directed registered plans sponsored by dealers and whether the administrative practices respecting these plans negatively impact clients' property and rights in the event of the dealer's insolvency or bankruptcy.

<sup>&</sup>lt;sup>193</sup> The reason that the problems are usually minimal is a practical one. The reason relates to the fact that the CIPF has substantial assets and the ability to obtain additional funding, if required, from its members. As a result, the CIPF is in a position to indemnify the bankrupt firm's receiver or trustee who then proceeds to distribute the client account securities to the clients in question once it is satisfied that the client is not indebted to the firm and is entitled to the securities.

<sup>&</sup>lt;sup>194</sup> While the scheme of distribution under the Bankruptcy and Insolvency Act (Canada) may strike those unfamiliar with this complex area of law as arbitrary and not "right", a review of the problems that existed prior to the enactment of the current provisions illustrates that the current provisions probably improve the ability of clients to reclaim their assets. It is, however, beyond the scope of this Review to make any determination in this respect.

Another reason that some intermediaries give for favoring nominee-name registration is because of the control that this gives the intermediary over clients and over their accounts. The fact that their securities are registered in nominee name makes it difficult for clients to move their accounts without the involvement and cooperation of the intermediary. Registration in nominee name also removes the ability of the issuer of the securities to communicate directly with clients and to know who their securityholders are.<sup>195</sup>

Industry participants tell me that this desire for control over clients and their accounts is responsible for an increasing number of mutual fund dealers (whose systems are usually designed to support client-name registrations) adopting the practice of registering their clients' securities in the name of the mutual fund dealer instead of the respective names of their clients.

Industry participants also tell me that another practice is developing whereby some mutual fund dealers are registering their clients' securities in the names of the mutual fund dealer and the client jointly. The result of this type of registration is that the dealer's consent is required for every transaction that the client may want to effect.

This joint registration means that all disclosure material that is sent by the mutual fund or other issuer to its securityholders of record is sent to the dealer as it is the first named of the joint registrants. I have been told that the explanation given to clients as to the need for joint registration is that the know-your-client/suitability requirements under applicable securities legislation require this.<sup>196</sup>

The practice of joint registration is one that is fraught with problems for consumer/investors. In addition to the obvious problems, there are tax implications and estate ramifications in the event of the client's incapacity or death.

### 21.4. Recommendations - Registration of Securities and Powers of Attorney

As can be seen from the foregoing, the matter of how securities are registered in the names of consumer/investors and the use of power of attorneys in connection therewith is of prime importance to consumer/investors. It is unfortunate that the current securities regulatory regime, combined with the current provisions of bankruptcy and insolvency legislation and property law, inadvertently operate in a manner that makes consumer/investors vulnerable.

One contributing reason is the fact that in Canada not all intermediaries who sell securities (including mutual fund securities) to the public are required to belong to a self-regulatory organization, the members of which are also required to be a member of the CIPF.

I understand that the current provisions in the Bankruptcy and Insolvency Act (Canada) relating to the bankruptcy or insolvency of securities dealers are modeled on comparable provisions of United States law. I also understand that the American provisions were adopted in Canada because the provisions had been proven to work effectively in the United States. It was therefore thought that they would also work effectively in Canada and that, by adopting them, there would be an added advantage that would result from the harmonization of Canadian and American laws.

<sup>&</sup>lt;sup>195</sup> Rule changes are currently being proposed by the CSA to improve the ability of issuers to communicate directly with their clients. See the Request for Comments published in the July 17, 1998 issue of the OSC Bulletin at (1998) 21 OSCB 1388.

<sup>&</sup>lt;sup>196</sup> I am not aware of any legal basis in law for this view.

It appears that in taking this practical approach, an important difference between Canadian and American law was overlooked. This difference in law relates to the fact that in the United States all persons who distribute securities to the public (including mutual fund securities) are required to be a member of the National Association of Securities Dealers and to be a member of the Securities Investor Protection Fund. Unfortunately, in Canada we do not yet have a comparable requirement in place.

The bankruptcy and insolvency provisions of both the Canadian and American legislation are designed to work when they are backstopped by an investor protection fund with deep pockets. This backstop does not exist in Canada for firms that are not members of a self-regulatory organization and members of the CIPF. This is a serious gap as is illustrated by the case of Vantage Securities Inc.

The CSA are in the process of filling this gap through the introduction of regulatory initiatives that will ultimately require all dealers to be a member of a self-regulatory organization and to participate in an adequately-funded customer protection fund.

However, this is a slow process. The fact that the process is slow makes it all the more imperative that we examine whether the process that is being worked on is sufficient for its purpose. This is particularly so because it is unlikely that small, fragmented self-regulatory organizations that focus on single products (like mutual funds) which are offered by a limited number (i.e. mutual fund dealers) of the intermediaries who offer such products, are likely to build up an adequately funded customer protection fund for several years (or perhaps ever).

The following are some suggestions that are aimed at helping to deal with the challenges presented by the current situation.

## Regulatory Structure

The unsatisfactory situation for consumer/investors that results from continuing the current fragmented regulatory structure reinforces the need for an integrated regulatory and supervisory structure and, in particular, the need for a single, strong, effective self-regulatory organization that will operate on a national basis and in which membership will be compulsory for all who provide investment advisory and financial planning services to the public. Specific recommendations for this are contained in Sections 12 and 16 of this Review.

#### **Customer Protection Funds**

The funding provisions for the CIPF (and also for all of the other customer protection funds) need to be reviewed. 197 It is probably time to go back to first principles and to re-examine their soundness and relevancy in the face of today's reality.

It is time to review what losses are covered by the customer protection funds and what losses are not covered. It is time to make sure that this information is clearly communicated to consumer/investors. A common misapprehension is that losses resulting from fraud are covered. This is true only if the firm involved is also insolvent or bankrupt.

It is important that firms that are non-members of a self-regulatory organization the members of which are also required to be a member of the CIPF ("non-member firms") begin now to provide funding to

-

<sup>&</sup>lt;sup>197</sup> I am told by industry participants that the actuarial assumptions that underlie the current funding provisions of the CIPF are insufficient to deal with the magnitude of the potential claims against it. I have no way of assessing the merit of these assertions except to note that the people making the assertions are responsible people.

cover losses that arise on the insolvency or bankruptcy of these firms. These firms should not wait until they are required to become members of a self-regulatory organization. It is well-acknowledged that the current provincial customer protection funds are inadequate to deal with client losses in the event of the insolvency or bankruptcy of a participating firm. Requirements to participate in interim funding arrangements should be a condition of ongoing registration.

### Insurance Coverage

The use of insurance to cover client losses resulting from errors and omissions or breach of contractual or fiduciary obligations (whether or not the firm that is involved is insolvent or bankrupt) should be looked into. The coverage requirements in this respect should be designed to complement and fill gaps in customer protection fund coverage.

Not all firms have this type of insurance coverage. Some non-member firms may not have the requisite internal systems, controls and procedures in place that would be required by an insurer underwriting errors and omissions and fidelity risks. The conditions of registration in this respect need to be reviewed and addressed.

#### Interim Measures

It should be recognized that an industry or regulatory delay in reviewing the conditions of registration relating to customer protection fund participation and insurance coverage for non-member firms exposes these non-member firms and their clients to added risks.

These non-member firms may well be at a competitive disadvantage as a result of their clients moving their accounts to member firms. Therefore, it is in the interests of the non-member firms to address without delay the problems resulting from inadequate customer protection funds and inadequate insurance coverage.

Given the potential problems that clients of non-member firms could experience because of the lack of an adequate customer protection fund to backstop client losses on the insolvency or bankruptcy of such non-member firms, the wisdom of permitting new registrations of non-member firms is questionable. Either a moratorium should be placed on such new registrations or appropriate conditions of registration should be attached to such registrations that would address the risk to clients.

#### Task Force

A knowledgeable task force should be assembled to review the complex interaction of property law, bankruptcy and insolvency law, personal property security law, corporate law and securities law matters that affects the interests of consumer/investors as they seek the most effective way to make and implement the fundamental decisions that will affect their ability to meet their lifetime goals and objectives.

This recommendation really belongs as part of the broader systemic recommendations dealt with in the earlier part of this Review. It is part of a fundamental need to review basic laws to ensure that they work effectively together. The discussion in subsection 21.3 of this Review of the interaction of property laws, securities laws, bankruptcy and insolvency laws, and tax laws highlights an area where laws do not work as effectively as they were intended to.

\_

<sup>&</sup>lt;sup>198</sup> See the 1995 Report - Section 9.02 - Contingency Fund Participation.

There is also a need to review basic laws to ensure that they work effectively for the electronic world. Most laws were developed with the paper world in mind - even some of those that ostensibly were designed for the electronic world. There is a need to review the industry practices and systems that have developed to see how they accord with current laws.

#### Matters such as:

- (i) the registration of securities;
- (ii) the interaction of administrative systems with book-based systems where there is a central depository of issuers' securities and with book-based system where there is no such depository;
- (iii) the impact of these issues and of tax laws on custodial and safekeeping requirements; and
- (iv) the methods for consumer/investors to deal easily with and communicate their instructions regarding their property interests;

need to be reviewed to determine whether adjustments in laws and/or procedures are necessary or desirable. Alternative means of communicating and authenticating client instructions need to be developed. Again, technological advances should be making this possible. Developments in the ability to use and verify the authenticity of electronic signatures should be an enabling facility in this respect for consumer/investors.

## Alerting Consumer/Investors

In the meantime, consumer/investors need to be alerted to the problems that they may encounter as a result of:

- (i) the manner in which their securities are recorded on the books of the intermediary with which they deal:
- (ii) the granting of powers of attorney;
- (iii) the membership status of the intermediary in a self-regulatory organization that is required to be a member of the Canadian Investor Protection Fund; and
- (iv) the lack of comprehensive insurance coverage for errors and omissions and fidelity obligations.

In this context, the importance of consumer/investors using due care in the selection of an intermediary should be emphasized.

Ideally, this consumer/investor alert would be accompanied by some practical advice for consumer/investors including the importance of not granting general powers of attorney to

intermediaries or their representatives and of alternative means of communicating their instructions. 199

# 21.5. Leverage

The current climate of low interest rates, a strong and seemingly continuous upward trend in stock market returns and the recognition by consumer/investors of the need to invest more aggressively to provide for their retirement income have combined to encourage people to commit a major portion of their assets to investment funds.

It is not clear what proportion of people's investments have been acquired with borrowed money. This is because loans for investment purposes are provided by third party lenders such as banks, trust companies, insurance companies and credit unions and not by the investment funds. Loans are also obtained from other sources including margin account loans from intermediaries (such as full-service investment dealers), insurance policy loans, and home mortgage loans.

Another reason that it is not clear how leveraged people's investments are is because there is no requirement for consumer/investors or intermediaries to disclose to the investment funds whether the money that is being invested represent borrowed money.<sup>201</sup> Presumably, those who lend the money have this information but they are not required to provide this information publicly. There also does not appear to be any entity that gathers and compiles this information.<sup>202</sup>

The euphoria of the last several years has served as a springboard for some intermediaries and seminar speakers to urge people to maximize their investment opportunities by using borrowed money to invest. They have seized the opportunity that has resulted from the merging of the prime drivers that are said to motivate consumer/investor behavior, namely hope, fear and greed. Accordingly, these intermediaries and seminar speakers have been aggressively advocating various strategies aimed at minimizing taxes and avoiding the impact of tax claw backs and senior benefit cut-back provisions. Three-to-one leverage programs<sup>203</sup> relating to segregated fund investing have also become the flavor of the day.

<sup>&</sup>lt;sup>199</sup> Section 11 of the 1995 Report describes the potential for the misuse of powers of attorney and recommends that business practice standards be developed to ensure that powers of attorney granted by clients are not misused. Although some work has been done in the area, I am not aware of any basic standards having emerged. Given the problems that powers of attorney may present in the event of the insolvency or bankruptcy of an intermediary, the matter of whether they should be used at all, except in limited, specific circumstances, should be re-examined. I believe that the practice of an intermediary taking a *continuing* power of attorney from clients is a relatively recent one.

<sup>&</sup>lt;sup>200</sup> In the current regulatory regime, the prime responsibility for issuing a consumer/investor alert of the nature described above could be expected to fall under the auspices of the CSA.

<sup>&</sup>lt;sup>201</sup> Mutual fund management organizations have some information about this matter as a result of administering systematic withdrawal plans for investors in their sponsored funds. These systematic withdrawal plans are often used to set up automatic payments to pay the interest and principal payments as they become due under a loan. Mutual fund organizations therefore are aware of payments that are directed to be made to third parties such as financial institutions even if they do not have actual knowledge that the payments that are being made are in respect of a loan provided to the investor. Mutual fund management organizations are not generally required to disclose any information that they do have.

<sup>&</sup>lt;sup>202</sup> In the case of loans made by federal financial institutions, this information may be part of the information that such institutions are required to provide to OSFI.

<sup>&</sup>lt;sup>203</sup> These programs generate four times the amount of the sales commissions that the intermediary would otherwise receive on the sale of the insurance and the mutual funds and four times the amount of the trailing commissions that would otherwise be payable. The mutual fund management organizations receive four times the assets under administration that

The desire to increase assets under administration has led some investment fund organizations to sponsor seminars and media presentations that have involved some very aggressive people and tactics. Some (including the Ontario Securities Commission) have referred to these people as "schills".

When reputable investment fund organizations sponsor people of this nature, such sponsorship brings with it in the eyes of consumer/investors an implicit endorsement of the speakers and the tactics. Such sponsorship involves an unwarranted transfer of the trust that consumer/investors have in the integrity of the investment fund managers to the speakers. The various leveraging techniques are promoted at retirement planning or investing seminars, in newsletters, newspapers and magazines and websites. They are also promoted through the use of radio and TV spots that leave the listener or viewer with the impression that some sort of legitimacy has been conferred on the speaker by reason of the speaker's association with the radio or TV station in question.

The public is left with the impression that "everyone is doing it", "it's easy", "you're foolish if you don't", "only the tax people will gain if you don't", "it's not risky", "you will gain" and so on. There is no disclosure of the substantial direct and indirect interests that the persons advocating these strategies have in maximizing the amount of the loans and the investments in question.

Virtually all of the advertising is focused on the upside. The systematic withdrawal plan material provided by mutual fund management organizations includes illustrations that are based only on *increases* in the annual growth rate of the sponsored investment funds. The annual growth rates used often exceed the historical compounded annual rate of return for the periods shown in the illustrations.

There are few requirements with respect to the disclosure of the impact of leverage. Several years ago, the OSC issued a disclosure statement that was required to be delivered to investors who consider borrowing money to buy investment funds. The purpose of the disclosure statement was to make investors aware of the risks involved in borrowing to invest. The adequacy of the OSC statement for its purpose has been questioned by many. In July of 1998, IFIC released a Fact Sheet on leveraging. While the Fact Sheet does provide some information for consumer/investors, it does not appear that its primary purpose is to help consumer/investors fully appreciate and assess the impact of their decision to borrow money to invest.

When leveraging strategies work, they are a win-win situation for all the participants who are involved in the process including the consumer/investor, the lender, the intermediary, the sales representative, the seminar speaker and the investment fund manager. Everyone stands to make money which is the objective.

When leveraging strategies do not work, it is still a win-win situation for everyone except the consumer/investor and, if the consumer/investor's collateral and other assets are insufficient to repay the loan, the lenders.

The substantial fees, charges, expenses, sales commissions (including trailing commissions), mortgage expenses, legal charges and interest payments that have been paid, directly or indirectly, by the consumer/investor are not refundable.

they would otherwise have on which their management fees are calculated so there is an incentive for them to sponsor or support through cooperative marketing payments, speakers and seminars of the nature described above.

\_\_\_

<sup>&</sup>lt;sup>204</sup> See Re Dino P. DeLellis (1998) 21 OSCB 305.

When loans are called and the investment funds are liquidated to repay the loan, consumer/investors may find that their redemption proceeds are substantially reduced by deferred sales charges, redemption fees and surrender charges. If there is a shortfall in the amount of the redemption proceeds, consumer/investors will be forced to liquidate other assets (such as a home or business) in order to raise the funds to repay the loans.

### 21.6. Recommendations Respecting Leverage

## Compliance and Enforcement Actions

Regulatory and self-regulatory authorities need to focus on the issue of suitability of leveraged transactions for the consumer/investor in question and the adequacy of the disclosure documents relating to the transactions.

The implementation by regulators and self-regulators of a strong compliance-oriented program, supported by hearings to consider the fitness for continued registration where the transactions are found not to be suitable or that the disclosure that has been made is inadequate, would curb a lot of inappropriate transactions.

## Refund of Fees, Charges and Expenses

The consumer/investor should be entitled to a refund of the fees, charges and expenses that the consumer/investor has incurred if:

- (i) the disclosure relating to the "loan and own" transactions is inadequate; and
- (ii) the consumer/investor's loan has been called and the proceeds on the redemption of the investment fund securities are insufficient to repay the loan.

This refund would include all sales commissions, trailing commissions and deferred sales commissions. The amount of the refund would be applied to reduce the balance of the loan. Any excess would belong to the consumer/investor.

## **Business Practice Standards**

Focusing on the fiduciary obligations of intermediaries would help. This could be done in a variety of ways. A starting point would be to develop minimum mandatory business practice standards to address suitability issues relating to the loan and own transactions.

Suitability issues would deal with matters such as:

- (i) the overall suitability of using leverage to achieve the consumer/investor's investment objectives;
- (ii) the amount of the loan together with the debt to equity ratio that the loan represents;
- (iii) the terms of the loan;
- (iv) the nature of the investments being acquired with the loan proceeds;
- (v) the suitability of these investments for the investment needs of the consumer/investor; and
- (vi) the other assets that are available to repay the loan.

In addition to addressing suitability issues, the business practice standards would address disclosure issues and issues that relate to conflicts of interest.<sup>205</sup>

# **Enhanced Oversight and Supervision**

Enhanced oversight and supervision by intermediaries respecting the suitability of proposed loan and own transactions should help. The supervisory standards in this respect would require sales representatives to disclose the "loan and own" transactions to the intermediary.<sup>206</sup> Information about such transactions would be part of the know-your-client/suitability provisions. Proper systems, controls and procedures should be in place which may not be bypassed.

## Increasing Knowledge and Awareness

Increasing the knowledge and awareness of consumer/investors about the sensitivity of the "loan and own" proposals that they receive to the assumptions about the investment returns and interest rates that have been used to prepare the proposals should help consumer/investors make more informed decisions. The initiatives recommended in Section 14 of this Review should assist consumer/investors in this respect.

Various tools could be developed to help consumer/investors decide whether they should borrow money to invest. These tools could include:

- (i) a basic consumer/investor awareness document prepared by an independent party who is not promoting the benefits of leveraging investments;
- (ii) a specific "loan and own" disclosure document;
- (iii) a "calculator" that allows consumer/investors to simulate the results of different interest rate and growth rate assumptions including various negative growth rate assumptions.

#### Loan and Own Disclosure Document

A disclosure document respecting the "loan and own" transaction should be required to be delivered to the consumer/investor before the loan is made. The purpose of this document would be to outline in plain language the terms of the proposed loan and of the proposed use of proceeds.

It is beyond the scope of this Review to identify all of the information that should be contained in a disclosure document of this nature. A working group (which should include independent parties who are not promoting the benefits of leveraging investments) should be assembled for this purpose. Current requirements relating to consumer loan transactions and cost of credit disclosure may provide a sound starting point.

I suggest that as a minimum, the document should contain several sensitivity scenarios with time horizons that relate to the term of the loan. The basic assumptions should be consistent with actual

.

<sup>&</sup>lt;sup>205</sup> See the recommendations in the 1995 Report - Section 11.04 - Leverage.

<sup>&</sup>lt;sup>206</sup> Frequently, sales representatives simply steer clients to a lending institution and the intermediaries who are responsible for exercising oversight and supervision over the sales representatives are not aware that the funds to pay for the investments are borrowed. Consequently the intermediary's review does not focus on the suitability of the transaction in the broader context of the use of borrowed money.

experience.<sup>207</sup> The sensitivity scenarios should include scenarios where the value of the investment declines and interest rates increase. The information in this document should be complemented by the information in the confirmation/point-of-sale document referred to in subsection 17.8 of this Review. The two documents could be combined if the information is clearly presented and it is clear to the consumer/investor what costs and risks are involved. If payments are made or received by related parties<sup>208</sup> this information should be disclosed together with the amounts of the payments.

## Systematic Withdrawal Plans

Schedules outlining systematic withdrawal plan payments should include sensitivity scenarios that conform with the parameters described above under the heading "Loan and Own Disclosure Document". The appropriateness of using securities that have been acquired on a deferred sales commission basis in a systematic withdrawal plan that is or may be used to pay the principal and interest on a loan and own transaction is questionable. This is a suitability issue that should be reviewed as part of the enhanced oversight and supervision referred to above.<sup>209</sup>

## Seminar Speakers/Implicit Media Endorsements

A code of standards and ethics should be developed for investor education and retirement planning seminars and for other initiatives including newsletters, websites, radio, TV and print media used to induce consumer/investors to invest and to use various investment strategies. Persons conducting such seminars or other initiatives should be expected to adhere to this code of standards and ethics and to be registered as "securities advisers" with the Single SRO referred to in subsection 16.3 of this Review.<sup>210</sup>

The Non-Partisan Standards Council referred to in subsection 15.10 of this Review could be responsible for developing and administering the code. Failure to adhere to the code could result in the suspension or termination of registration as a securities adviser. Through a combination of:

- (i) rules of fair practice of the Single SRO referred to in subsection 16.3 of this Review;
- (ii) the amendment of the sales practices rule referred to in subsection 16.8 of this Review; and
- (iii) voluntary action by the media, sponsors of websites, and by industry participants:

speakers and authors who did not conform to the standards and whose registration has been suspended or terminated would not be able to be retained to speak at seminars sponsored by industry

<sup>207</sup> For example assuming 20 years of continuous growth rates at 20 per cent per year for an investment fund is simply not

borne out by the actual performance of most investment funds.

208 "Related parties" would include the sales representative, the sales representative's firm, their respective associates or

<sup>&</sup>quot;Related parties" would include the sales representative, the sales representative's firm, their respective associates or affiliates, the investment funds in which the investment is being made and the manager or principal distributor of such investment funds and their respective associates and affiliates.

<sup>&</sup>lt;sup>209</sup> If margin calls are made under these loan and own plans, it comes as an added shock to consumer/investors that the units that are required to be redeemed to meet the margin calls are subject to deferred sales commissions being deducted from the redemption proceeds otherwise available to meet the margin call. This will be the case if the units being redeemed exceed the number of units that the investment fund in question permits as a "free redemption."

<sup>&</sup>lt;sup>210</sup> The Securities Act (Ontario) contains provisions for the registration of "securities advisers". See Section 99(4) of the Regulation made under this Act.

participants or to use media facilities (including their sponsored websites) to publish their views. In addition, investment fund organizations would not be permitted to provide such persons with cooperative marketing payments to subsidize their promotional activities of such persons.

In addition to ensuring that securities advisers comply with the code of standards and ethics and to the disclosure of who is sponsoring and picking up the costs related to the event, it is important that there be up-front disclosure that the speaker is being paid to speak, accompanied by information as to how the speaker is being compensated, who is responsible for compensating the speaker and what future compensation or participation in profits the speaker will receive.

This disclosure should provide sufficient detail to enable consumer/investors to understand how any transactions they enter into will affect the speaker's current or future compensation. The disclosure should include disclosure of any actual or potential ownership interest that a speaker has in any entity that is involved in or with the speaker's presentation or that the speaker recommends as an investment or otherwise.<sup>211</sup>

#### 21.7. Transfer of Accounts

Probably the most frequently voiced complaint by industry participants relates to the length of time it takes to transfer accounts from one intermediary to another, particularly in the case of registered plan accounts. This frustration is shared by consumer/investors who do not understand why it takes so long to transfer their accounts.

Delays in transferring accounts have been a problem for a long time. The minimum time period to transfer accounts is now about four weeks for a cash account and six to eight weeks for a registered plan account. Often the time periods are longer. During this period, the investments are in limbo. Consumer/investors are unable to effect transactions in their accounts or to exercise any voting or other rights. They are exposed to market fluctuations and they may not receive any interest on cash balances or dividends or other distributions made on investments held in their accounts because of funds being held in suspense accounts pending payment.<sup>212</sup>

According to industry participants the reason for the delay in transferring accounts is that there is no impetus for the relinquishing intermediary<sup>213</sup> to process the transfer transaction quickly. There is nothing that the relinquishing intermediary will gain by doing so. It has already lost its customer so the motivation to keep its customer happy is not there. Often the relinquishing intermediary is losing assets under administration. This will affect its revenues and the amount that its sales representatives are entitled to receive. It may therefore be in the relinquishing intermediary's interests to delay the transfers until after the end of a month or the end of a quarter in order to maintain their entitlements in respect of the period in question. A delay in completing the transfer also provides the sales representative with the

-

<sup>&</sup>lt;sup>211</sup> Sometimes these speakers include examples of investments that they "like." Sometimes these investments turn out to be investments in which they have a financial interest. Too often consumer/investors are influenced by the speaker's comments and end up losing a lot of money. Hopefully, by raising the level of knowledge and awareness of consumer/investors through initiatives outlined in Section 14 of this Review, consumer/investors will be able to protect themselves better. In the meantime, there is also a need to try to deal with the problems by developing some standards as noted above.

<sup>&</sup>lt;sup>212</sup> Although they are entitled to receive such distributions, if consumer/investors do not ask for them it is unlikely that anyone will take the time to make sure they receive their entitlement.

<sup>&</sup>lt;sup>213</sup> The adjective "relinquishing" is used by the industry to describe the intermediary who is giving up or transferring the account. The adjective "receiving" is used to describe the intermediary who is receiving the account as a result of the transfer.

opportunity to contact the client to see if the client will reconsider. Another factor that contributes to the delay is that the relinquishing intermediary, as might be expected, places priority on completing work that serves the needs of continuing and new clients.

The requirements of Revenue Canada with respect to the transfer of registered plans are often used as an excuse for the delay. However, Revenue Canada has simplified its requirements. It no longer requires a prescribed form to be used in order to effect transfers from one registered plan to another. Also its requirements regarding registered plan transfers do not affect non-registered plan transfers.

# Inter-Industry Working Group

An inter-industry working group was established in late 1997 to develop guidelines for the timely processing of registered account transfers including the creation of a standard model form for use by the financial services industry for a registered account transfer request. This model form includes much of the same information that was previously required by the Revenue Canada form.

The Registered Account Transfer Group's Report was issued in 1998 and has been distributed to the participants in all sectors of the financial services industry. I understand that its timely processing guidelines are being used by the participants in the mutual fund industry but that they are not being generally used by the participants in the other sectors of the financial services industry despite their participation in the development of the guidelines. The guidelines are voluntary and there are no sanctions for not adhering to them.

The Registered Account Transfer Group's Report outlined the common problems that are encountered in the transfer of registered plan accounts. Some of these problems are complex. The problems highlight that in some cases there is a lack of understanding on the part of consumer/investors and their advisers as to what is involved in effecting the transfer and the financial penalties that may be incurred in doing so.

Some problems are caused by inadequate instructions being given to the relinquishing intermediary concerning whether the assets are to be transferred in kind or liquidated. In some cases, instructions are given to transfer assets in kind when it is not possible to make such transfers. These problems contribute to the delays.

Another contributing factor to the delays is the fact that not all participants in the various sectors of the financial services industry have automated their processes to the same extent. A lot of the processes that are involved are still manual processes. Even where processes have been automated, delays occur due to the lack of established standards for electronic processing that apply across the various sectors of the financial services industry.

Understanding the problems is only a first step. Now that the problems have been clearly identified in the Registered Account Transfer Group's Report, they need to be dealt with and dealt with promptly. Eliminating the delays in the transfer of accounts would benefit consumer/investors immensely. Consumer/investors are exposed to unacceptable market risks by reason of the delays.

# 21.8. Recommendations Relating to the Transfer of Accounts

### Establishment of Cross-Industry Standards

Cross-industry standards should be established for the processing of all transfer requests relating to non-registered plans as well as registered plans. A lot of work has been done by the Registered Plan Transfer Group. This work should be completed and satisfactory standards should be issued.

In developing the standards, it is important that the time delays be kept to the minimum time required to complete the transaction. It is difficult to understand, for example, why the Registered Plan Transfer Group's Report proposed that five business days be allowed to liquidate money market instruments when the settlement date for money market transactions is on a same-day basis. There is no reason why a duly completed transfer request cannot be processed in the same timeframe that any other order for the sale of securities is processed. There is no reason why it should be treated differently from any other sell order.

While it is important that the time delays prescribed in the transfer request standards be realistic, it is also important to recognize that some of the delays result from inadequacies in industry participants' systems, controls and procedures and the failure to adopt common cross-industry electronic processing standards. Consumer/investors should not have to suffer from this inefficiency.

Therefore cross-industry electronic processing standards also need to be developed.<sup>214</sup> It should be a condition of registration or licensing and membership in the Single SRO that all financial institutions, registrants and members of self-regulatory organizations together with their respective service providers have the ability to adhere to the established cross-industry electronic processing standards and the account transfer processing standards.

This is an area where an integrated regulatory and self-regulatory approach would help. This is an area that is unlikely to move forward unless a concerted effort is made by all of the current regulators and self-regulators to make it happen. Industry and regulatory fragmentation and operational inefficiencies should not be allowed to negatively affect consumer/investors.

## Increasing Knowledge and Awareness

This having been said, it is important to recognize that many delays are caused by consumer/investors and their advisers giving inadequate instructions to carry out the requested transfer and to not understanding what is actually required to effect the transfer. Not all assets can be transferred in kind. Not all assets can be instantly liquidated. Some assets can only be liquidated if a substantial penalty payment is made or deferred sales commissions or surrender fees are paid. Where estates are involved, further documentation is usually required. It is important that consumer/investors and their advisers recognize this **before** they give instructions to transfer the account. It is important that if they decide to proceed that clear and sufficient instructions accompany the transfer request so that the relinquishing intermediary can proceed to process the transaction.

To deal with this problem, it is important that the education and proficiency requirements for advisers cover this subject. There would be two benefits to having a more knowledgeable adviser. Firstly, the adviser would be better able to explain to clients what is involved in the transfer process. Secondly, the advisers would be better able to complete or assist the client in completing the account transfer instruction forms.

Programs aimed at increasing the knowledge and awareness of consumer/investors should also cover the subject of account transfers so that consumer/investors are aware of how the account transfer

-

<sup>&</sup>lt;sup>214</sup> FundSERV has done substantial work in this area. FundSERV is a Canadian enterprise that has spearheaded the development of what it characterizes as being the most efficient investment fund transaction processing system in the world. Its goals include (i) providing the foundation for a fully automated investment fund environment, (ii) helping its members to maximize the benefits of their investment in technology, (iii) promoting efficiencies at every stage of the administrative process, (iv) reducing its members' dependence on temporary staff to handle seasonal increases in order volume, and (v) providing support and assistance in the development of industry standards to realize these goals. More information about FundSERV is posted on its website at <a href="http://www.fundserv.com">http://www.fundserv.com</a>.

process works, what is involved and what constitutes a realistic time frame for completing transfers. The basic education document referred to in subsection 17.7 of this Review should include some of the basic information in this respect.

# Know-Your-Client/Suitability Procedures

As part of the know-your-client/suitability procedures,<sup>215</sup> the receiving intermediary should review with clients who decide to transfer their accounts how the account transfer process works, what is involved and what a realistic time frame is for completing the transfer.

The fees, charges and expenses (including any financial penalties for early redemptions, surrender fees and deferred sales commissions) should be discussed with the client and the client's consent obtained.

This acknowledgment and consent, where appropriate, should accompany the transfer request as it will minimize delays caused by the relinquishing intermediary contacting the client to ensure that the client understands the consequences of completing the requested transfer.

A copy of the acknowledgment and consent should be given to the client together with a written description of what is involved in the transfer and what the expected timeline is.

These requirements should be supplemental to the switch disclosure document referred to in subsection 21.2 and could be part of such document.

#### Penalties for Failure to Adhere to Standards

Industry participants suggested that there should be penalties for delays in completing transfers that exceed the delays set forth in the account transfer standards. These penalties would be based on the client's loss of opportunity and would be payable to the client.

#### Regulatory Action

**Tax Aspects of Investment Funds** 

Coordinated, concerted and cross-sector regulatory and self-regulatory action needs to be taken to make sure that account transfer standards are adopted and adhered to without further delay. Simply asking the industry to come up with proposals has produced some results but more is needed to make them operate in an effective and timely manner.

# 22. TAX MATTERS

One of the least understood aspects of investing in investment funds relates to tax matters. There is

general agreement that it would make a big difference to consumer/investors if the tax aspects of investing in investment funds were simplified.

-

<sup>&</sup>lt;sup>215</sup> The "transfers-in" should be subject to special audit for suitability, particularly when the "transfers-in" are being invested in the dealer's proprietary funds or wrap accounts. Industry participants tell me that there have been huge outflows of money to the "captive fund sector" and refer to there being "major conflicts of interest" created by the desire to increase the assets in these captive funds.

Various industry groups have made submissions to governments respecting the complex, technical issues that are involved. They have also made submissions about the need to level the playing field by eliminating tax advantages that flow to products offered by one sector of the financial services industry over products which are their functional equivalents but are offered by other sectors of the financial services industry. Many of the tax issues are being studied by governments. Hopefully the result of these studies will be to simplify and rationalize a tax structure that was designed (at least in part) for an era when product, function and advice had not merged.

The major areas of concern that were raised by industry participants are listed below. Changes in these areas would benefit consumer/investors.

# Foreign Property Limits

The limits on the investment by registered plans in foreign property were at the top of everyone's list of tax matters that they would like to see changed.

Apart from the substantive investment issues that the foreign property investment restrictions raise, people's concerns centered on the costs of compliance with the restrictions. These costs include:

- (i) the administrative costs involved in monitoring foreign content;
- (ii) the trading costs and the lost opportunity costs of taking action to bring portfolios on side;
- (iii) the substantial penalties to which registered plans and consumer/investors are subject if the foreign property content in the investment funds or the registered plan accounts exceeds permitted limits; and
- (iv) litigation costs where liability is disputed.

It was pointed out that despite the efforts made by investment funds and their advisers to monitor foreign property content, mistakes do occur. The consequences for an investment fund and its securityholders can be financially devastating unless relief is granted by tax authorities.

In the case of consumer/investors part of the difficulty in monitoring foreign content is that many people do not have accurate records of the adjusted cost base of their respective investments.

Often this is because people do not understand the concept of "adjusted cost base". They do not recognize what information they need in order to calculate their adjusted cost base. They do not know how to calculate their adjusted cost base and they do not understand the circumstances in which they will need to use the information.

Many people rely on their advisers to provide this information and to monitor their foreign content. However, the records maintained by their advisers may not be sufficient to enable an accurate calculation to be made. While in some cases this may be a systems problem, it often is due to the fact that the client has not provided the intermediary with accurate historical information that predates the person becoming a client of the intermediary. Gathering this information is very labor intensive for those who attempt to do it.

Many submissions have been made to the federal government urging that either:

(i) the foreign property limit on investments be raised to the level which most experts say would be the "natural" level for foreign investments if there were no restrictions on foreign property investments; or

(ii) the foreign property investment restrictions be repealed.

In determining what course of action to follow, one of the factors that should be considered is the enormous administrative and other costs that are involved in monitoring foreign content. These costs are ultimately are borne by consumer/investors.

Another factor that should be considered relates to the potential liabilities that problems in this area create for industry participants and consumer/investors alike. These costs and liabilities affect the amount that consumer/investors will have to provide for their retirement income and will have an impact on the amount otherwise needed to be provided by governments as part of their social and retirement programs to make up shortfalls.

A third factor that should be considered is that it has become very easy, through a variety of investment vehicles that have been created for the purpose, to avoid the foreign property restrictions on investments. Doing so usually comes at a higher cost for consumer/investors which will have an impact on the amount that they will have to provide for their retirement income and on the amount otherwise needed to be provided by governments as part of the social and retirement programs for Canadians.

These three factors seem to argue in favor of repealing the restrictions. However, if the decision is made to retain the foreign property investment restrictions, it would benefit everyone involved if the calculation of foreign property content could be simplified. Hopefully, the administrative cost savings would be passed on to the consumer/investor.

## Registered Retirement Savings Plans

The deadline for contributions to registered retirement savings plans has created a sales frenzy for investment fund products that does not well serve the needs of consumer/investors. This sales frenzy is transaction-oriented. It is inconsistent with the expressed desire of consumer/investors to make better decisions to provide for their well-being.

In addition, the large cash inflows into investment funds present investment challenges for fund managers as they seek to invest the money they have received. Often too much money is chasing too few investments given the current restrictions on investing in foreign property. The large volume of orders that is received during the registered retirement savings plan season stresses all of the systems and procedures including the know-your-client/suitability procedures.

To remedy the problem, it is suggested that the deadline for contributions to registered retirement savings plan be changed to coincide with the individual's birth date.

### Amendments to Registered Plan Requirements

Industry participants note the high costs that are involved in amending registered plans to conform with changes in governmental requirements for such plans. These costs are ultimately borne by consumer/investors. Industry participants would like to find a means to deal with these changes without an amendment to the plans being required.

One way to do this would be by not requiring that the substantive provisions that are contained in the income tax legislation and that are common to all registered plans be stated in the application and related declaration of trust form. Instead, there would be a requirement simply to notify planholders of any substantive change within a certain period of time following the effective date of the change. This notification could accompany, or be contained in, an interim report or an account statement to eliminate the costs of an added mailing.

This would avoid the need to amend registered plan documents every time a change in the applicable legislation is made but would still provide planholders with notice of the change. Planholders should benefit from a reduction in administrative costs and governmental resources would be freed for other activities.

# **Capital Gains Distributions**

There has been a growing concern over the last several years about the capital gains distributions that are made by investment funds to unitholders who are not tax exempt. This concern has led to a growing number of media writers and intermediaries advising people to redeem their units prior to the year-end record date for distributions or to defer their purchase of units until after the year-end record date for distributions.

Sometimes this desire to avoid receiving a taxable distribution overrides sound investment decisions or results in consumer/investors being taxed on realized gains on redemption that exceed the taxable capital gains distributions that they would otherwise have received.

The fact that the capital gains distributions increase the adjusted cost base of the units held by consumer/investors and that this increased adjusted cost base will reduce the amount of the capital gains (or increase the amount of the capital loss) when the consumer/investor's units are redeemed is often overlooked. However, even when people do take this fact into account, there are still concerns about the timing gap.

The concerns lead to a skewing of investment decisions made by investment fund managers who try to manage portfolios to eliminate capital gains and by consumer/investors as they endeavor to avoid receiving taxable distributions.

Industry participants would like to see changes in how capital gains distributions are calculated to enable the amount of any capital gains distribution to be credited to unitholders more frequently than once a year. <sup>216</sup> Ideally they would like to see the Income Tax Act (Canada) amended in a way that would enable them to credit realized capital gains and losses on a periodic basis throughout the year so as not to accumulate unallocated taxable capital gains that would have to paid out.

Another area of concern with respect to capital gains tax matters relates to mutual fund corporations with multiple classes of securities. The separate classes of securities are basically separate funds. People switch from one fund to another without paying capital gains tax. Tax liability on capital gains is the responsibility of the remaining securityholders. The potential liability of such securityholders for capital gains tax is masked until the fund stops growing. This is a complex area that could present consumer/investors with unexpected tax liabilities. Beyond the importance of disclosure, the issues seem to relate to the question of the suitability of the investment in the first place and fairness issues in structuring the investment.

## Payment of Income and Capital Gains Distributions

Industry participants are concerned about the amount of paperwork that is involved in investment fund management matters. The paperwork involving income and capital gains distributions is one of the areas of complaint. They would like to see it reduced.

-

<sup>&</sup>lt;sup>216</sup> The provisions of the Income Tax Act (Canada) relating to calculating the amount of the capital gains distribution that must be made in order for the investment fund not to be subject to tax are complex. The result is that until year-end, investment funds do not have all of the necessary information to make the calculation.

They would like to eliminate the requirement to pay out distributions and to reinvest the distributions.

This could be done by enabling the amount of any distributions that are payable to be capitalized as is permitted to be done in investment funds where all of the investors are tax exempt. The ability to "capitalize" distributions would have the added benefit of simplifying the calculation of the unitholder's adjusted cost base for foreign property tax purposes. However, other modifications in income tax legislation would have to be made either to eliminate tax on the capitalized amount of the capital gains distributions in the year in question or by adjusting the amount of the realized capital gain (or loss) on the disposition of the unitholder's securities.

Industry participants point out that one of the differences between the treatment of mutual funds and segregated funds under the Income Tax Act (Canada) relates to the requirement to pay out and then reinvest the distributions. Segregated funds do not have to do this. They just issue tax information statements. Mutual funds have the administrative burden (and related costs) of the pay-outs and reinvestments which adds to their competitive disadvantage.

## Information for Income Tax Purposes

Some industry participants have indicated that there are problems meeting the deadlines for providing consumer/investors with the information that they require for income tax purposes.

These industry participants say that they do not receive all of the information that they need to prepare the tax information statements for consumer/investors until the last day of February. This makes it difficult to meet the March 31 deadline for issuing the tax information statements that consumer/investors need for income tax purposes. They also say that consumer/investors expect to have all of the information that they require for income tax purposes by the last day in February. This erroneous expectation results in many consumer/investors having filed their income tax returns before they have received the tax information slips that they require respecting their investment fund income.

This is an area that should be looked at to see if better technology and better systems would enable basic information to be provided earlier. This seems to be a business process issue that should be addressed by the investment fund organization or other intermediary before it offers investment funds to the public or records securities owned by its clients in the firm's name.<sup>217</sup>

It would also seem that better communication with consumer/investors would help alert consumer/investors to what they should expect to receive and when. Some investment funds are beginning to include this information in interim reports and with year-end account statements. In addition, some investment funds are beginning to include explanations about the capital gains distributions that they anticipate making.

### **Uniform Provisions for the Taxation of Investment Funds**

Concern was expressed about how differences in tax treatment for products like segregated funds and other investment funds such as mutual funds provide a competitive advantage for one sector of the financial services industry over another. Concern was expressed about how these differences may

<sup>&</sup>lt;sup>217</sup> It should be noted that some industry participants have administrative and record-keeping systems in place that track throughout the year all of the information that they need to prepare the tax information statements that they send to their unitholders. They are not reliant on the T-5 slips that they receive from the issuers of the portfolio securities that they held during the tax year in question in order to prepare the tax information statements that their unitholders require.

skew the advice given to consumer/investors and may distort the investment decisions that are made by consumer/investors.

This is an area that should be reviewed as part of a coordinated cooperative effort to level the playing field.<sup>218</sup>

#### 23. PROPERTY LAWS

# 23.1. Modernizing and Harmonizing Property Laws

Another area that would help consumer/investors achieve their goals and objectives relates to modernizing and harmonizing basic laws relating to the ownership of property. This is a complex area that crosses many other areas of the law including corporate law, securities law, insurance law, trust law, pension law, estate law and tax law.

Some of these laws are provincial laws which vary from province to province. Some of these laws are federal laws. Some areas of these laws are covered by both provincial and federal laws. There is needless complexity.

This is a crucial area where coordinated, cooperative and concerted action on the part of all levels of government to modernize and harmonize laws would benefit consumer/investors by ensuring that people's basic property rights are not dependent on where they live in Canada. A lot of the work has been done in this area by various task forces, Uniform Law Commissions and other organizations. It is time for Canadians to reap the benefit of this.

It would also be an opportune time to review our basic laws to ensure that they work effectively for the electronic world as well as for the paper world and that they are harmonized internationally.

Some of the areas that most directly affect consumer/investors in relation to investment funds are noted below.<sup>219</sup> All of these areas create substantial direct and indirect costs for consumer/investors when they come up against them.

### Eligibility of Investment Funds as Trustee Investments

There are questions about the eligibility of the various types of investment funds as investments for trustees. The laws regarding this matter vary from province to province. These laws vary depending on the type of investment fund vehicle and on the type of issuer of the investment fund.

Some jurisdictions have attempted to deal with the question of eligibility by adopting "prudent person" provisions. There are questions as to whether the "prudent person" provisions sufficiently address all of the questions relating to eligibility that arise under trust law. These questions relate to whether the "prudent person" provisions overcome the principles of trust law that provide in effect that a trustee may not delegate discretionary authority and that a trustee must maintain an even hand between the interests of the capital and income beneficiaries. There is also a related question as to whether these principles of trust law override an express provision in a will that permits a trustee to retain and/or to invest in investment funds.

.

<sup>&</sup>lt;sup>218</sup> Section 20 of this Review discusses these concerns in more detail.

<sup>&</sup>lt;sup>219</sup> In order to keep this Review to a manageable length, I have not attempted to discuss the issues in any depth.

Examples of the practical areas of concern that exist by reason of there being questions relating to the eligibility of investment funds as investments for trustees include:

- (i) the ability of executors and trustees under a will or an inter-vivos trust to invest in mutual funds or other types of investment funds;
- (ii) the ability of executors and trustees under a will to retain or to add to any investment in mutual funds or other types of investment funds owned by the deceased person at the time of his or her death;
- (iii) the ability of a trustee under an "in-trust account" to invest in mutual funds or other types of investment funds;
- (iv) the ability of a trustee under a registered plan (such as a registered retirement savings plan, a registered retirement income plan, a registered education savings plan and a pension plan that invests in pooled funds) to invest in mutual funds or other types of investment funds; and
- (v) the ability of a trustee of an investment fund to invest in other investment funds (e.g. when fundof-fund investments are made).

## Designation of Beneficiaries

The provisions relating to designation of beneficiary requirements vary from province to province and from product to product. The provisions relating to changing the designation of a beneficiary also vary. There are also variances in the provisions of the laws that relate to whether the designation of a beneficiary in a will prevails over a designation of beneficiary that is made outside of the will.

# **Probate Requirements**

The provisions relating to the circumstances in which probate is required to be obtained vary from province to province and according to the type of investment fund vehicle that is involved. These variances give rise to two concerns.

The first concern relates to the fact that investment decisions are sometimes distorted by the issue of whether the investment fund vehicle is exempt from probate or not.<sup>220</sup>

The second concern is that some people are apparently making some very improvident arrangements to avoid paying probate fees. This is an area where a better understanding by consumer/investors and their advisers of basic property laws and the consequences of actions that they may take would assist.

In addition to addressing the two concerns noted above, it would benefit consumer/investors if the laws of all jurisdictions were harmonized to raise the exemption levels for requiring executors to obtain probate to a realistic level. Currently, consumer/investors incur substantial costs in seeking waivers of probate from the various investment fund organizations. Investment fund organizations also incur substantial costs in dealing with these requests as well as other estate-related matters.

<sup>&</sup>lt;sup>220</sup> Segregated funds (being insurance contracts) are generally exempt from probate requirements whereas mutual funds are not. The two investment products are functional equivalents of each other.

### **Creditor-Proofing**

The provisions relating to the circumstances in which investment funds and annuity payments are exempt from seizure by creditors under provincial and federal laws vary from jurisdiction to jurisdiction and from product to product. These variances distort investment decisions and provide some sectors of the financial services industry with a competitive advantage over other sectors of the financial services industry.

The variances also add to the complexity of the products and to their costs. These added costs are borne by consumer/investors who seek to avail themselves of the advantages of "creditor-proofing" their investments or the annuity payments that they receive. Consumer/investors who seek to avail themselves of this protection from the claims of creditors may ultimately discover that they have done so at a considerable cost to their capital and at an increased exposure to credit risk.<sup>221</sup>

# **Powers of Attorney**

The provisions relating to powers of attorney vary from province to province. One of the problem areas for consumer/investors relates to continuing powers of attorney and the circumstances in which the provisions of a continuing power of attorney override or are overridden by specific powers of attorney. These questions frequently arise as a result of financial institutions requiring their own form of power of attorney to be executed to deal with the transactions effected at or through the financial institution.

This is an area where harmonization of laws would help. However, if consumer/investors and their advisers had a better understanding of the issues involved in granting a power of attorney this would help even more. This subject matter is one that should be covered in the initiatives that are undertaken to enhance the knowledge and awareness of consumer/investors and their advisers.<sup>222</sup>

#### In-Trust Accounts

The problems with in-trust accounts relate primarily to the fact that sometimes people have not thought through all of the issues that need to be addressed when a trust is created. These issues go beyond the issue of how the assets of the in-trust account will be taxed.<sup>223</sup>

Other problems<sup>224</sup> that arise with respect to in-trust accounts relate to inconsistencies in some standard form documents that some intermediaries have prepared for use by their clients and the fact that sometimes the standard form document does not address all of the issues that need to be addressed.

Again, it would benefit consumer/investors and their advisers if they had a better understanding of the issues that are involved in the use of in-trust accounts. The subject matter is one that should be covered in the initiatives that are undertaken to enhance their knowledge and awareness.<sup>225</sup>

<sup>222</sup> See Sections 14 and 15 of this Review.

<sup>&</sup>lt;sup>221</sup> See Section 20 of this Review.

One issue regarding in-trust accounts, as noted under the subheading, *Eligibility of Investment Funds as Trustee Investments*, is whether investment funds are eligible investments for an in-trust account.

Other issues regarding in-trust accounts are outlined in an article by Jonathan Chevreau in The Financial Post dated June 27-29, 1998 entitled "Child trusts require caution".

<sup>&</sup>lt;sup>225</sup> See Sections 14 and 15 of this Review.

It was also suggested that it would benefit consumer/investors if standard forms to establish in-trust accounts were developed. Given the problems referred to above concerning "standard form documents" considerable care would have to be taken to ensure that these documents do not create more problems than they solve. Technology might provide some assistance through the ability to generate a customized "standard form" that is based on answers to specific questions that are designed to elicit the information that is needed to produce an effective in-trust document that reflects the consumer/investor's wishes.

The problems that arise with respect to the sufficiency of establishing seemingly simple in-trust accounts highlight the importance to consumer/investors of the competency of the financial, legal and tax advisers that they rely on. The problems also highlight the need for the integrated educational and proficiency requirements that are recommended in Section 15 of this Review.

## Standard Execution Requirements

Another area where harmonization of the laws of the respective jurisdictions would simplify matters for consumer/investors, their advisers and all industry participants relates to standardizing the requirements for executing documents. These requirements include the necessity (or lack thereof) for signatures being witnessed, notarized and guaranteed.

The standards for the execution of documents also need to provide for and address the standards for electronic signatures and the recognition of them.

The laws of evidence of the respective jurisdictions need to be reviewed, modernized and harmonized in this respect.

#### 24. CONSUMER REDRESS MECHANISMS

#### 24.1. Consumer Redress Mechanisms

The subject of consumer/investor redress mechanisms requires more work to be done than what I have been able to do in the context of this Review. My primary focus in this Review and in the 1995 Report has been on initiatives that lessen the need for consumer redress mechanisms by trying to keep the problems from occurring in the first place. However, there is still a big gap between the "what is" and the "what should be". Consumer redress mechanisms therefore need to become one of the strategies to close this gap.

## 24.2. Types of Consumer Redress Mechanisms

The common types of consumer redress mechanisms include:

- (i) the right, in certain circumstances, to withdraw from or rescind a transaction and to receive a refund of any money that has been paid;
- (ii) the right, in certain circumstances, to receive damages:
- (iii) the right, in certain circumstances, to make claims against customer protection funds that are maintained by or on behalf of certain sectors of the financial services industry to protect customers, depositholders and policyholders;
- (iv) dispute resolution procedures, including complaint procedures and voluntary or mandatory mediation and arbitration procedures;

- (v) compliance and enforcement proceedings;
- (vi) class action proceedings;
- (vii) private litigation proceedings;
- (viii) recourse to industry ombudsmen; and
- (ix) depending on the nature of the claim, recourse to government-appointed ombudsmen.

#### 24.3. Variances in Consumer Redress Mechanisms

Currently, the redress mechanisms that are available to consumer/investors vary according to:

- (i) the investment product;
- (ii) the institution issuing the investment product; and
- (iii) the intermediary that is involved.

These differences and their ramifications insofar as their redress rights are concerned are not readily apparent to consumer/investors.

#### 24.4. Recommendations

Although I obtained various perspectives on consumer redress mechanisms during the course of my discussions with industry participants and I have reviewed some of the work that has been done in this area, I am not currently in a position to make specific recommendations relating to consumer redress mechanisms beyond some of the more obvious ones. These recommendations relate to the following needs.

#### **Customer Protection Funds**

There is a need to make clear disclosure to consumer/investors as to whether their investments are covered by a customer protection fund, which customer protection fund(s) provide coverage, the limits on the amount of the coverage, and the circumstances in which coverage may be claimed.<sup>226</sup>

As noted in subsection 21.4 of this Review, there is a need to review the scope and adequacy of the customer protection funds for their purpose.

#### Complaint Handling Standard

There is a need to develop a common standard for the handling of consumer/investor complaints that will apply across all the sectors of the financial services industry and on a Canada-wide basis, with the procedures for consumer/investors to follow being clearly and simply laid out. This standard would address at least the following elements:

(i) the need for each registrant to establish a procedure for handling complaints which would include requirements to keep records of all complaints received and of how and when they were

<sup>&</sup>lt;sup>226</sup> Subsection 17.8 of this Review recommends that this disclosure be made in the confirmation/point-of-sale statement.

resolved and to make this information available to regulators (including self-regulators); the procedures would also address the circumstances under which the information *must* be reported to regulators/self-regulators;<sup>227</sup>

- (ii) the need to provide consumer/investors with clear information about what to do if they have a problem; ideally this information would be provided at the time of opening an account and would include the persons (identified by positions) to whom complaints should be addressed; the information would alert consumer/investors to the fact that enforcement proceedings brought by regulators and self regulators are not likely to result in consumer/investors receiving restitution and would alert them to the procedures that they may take in order to receive restitution and/or be awarded damages;
- (iii) the need for there to be a central registry (on-line) to record disciplinary proceedings against intermediaries and their representatives.

This is an area where regulators, self-regulators and consumer representatives might effectively work together in the development of a common standard for the handling of consumer complaints that will apply across all the sectors of the financial services industry and on a Canada-wide basis.

#### Access to the Legal System

There is a need to develop more effective ways for consumer/investors, at an affordable cost, to pursue remedies to recover losses they incur caused by negligence or resulting from a breach of a contractual or a fiduciary obligation.<sup>228</sup>

#### Withdrawal and Rescission Rights

There is a need to identify clearly what the rights of withdrawal and rescission are in respect of any investment product.<sup>229</sup>

#### 24.5. Ombudsmen

The use of industry-appointed ombudsmen and/or government-appointed ombudsmen has come to be regarded as a consumer redress measure.

Before such measures are proceeded with, it is important that more thought be given to this. Firstly, what is it that the ombudsman is supposed to do? What powers is the ombudsman to have?

\_

<sup>&</sup>lt;sup>227</sup> See subsection 16.7 of this Review.

<sup>&</sup>lt;sup>228</sup> Currently, class action proceedings and voluntary and mandatory arbitration or mediation proceedings do not adequately address this need. Other initiatives need to be explored to build on what is now available. Perhaps law school sponsored programs could be developed to assist consumer/investors in framing and pursuing their complaints. Perhaps additional case assessment and case management procedures could be developed to facilitate actions proceeding through the courts. Most consumer/investors simply do not have the resources to pursue their legal remedies especially when the proceedings are protracted by one interlocutory proceeding after another. Some claims are too small to be heard in the Supreme Courts but they involve principles of law to which sometimes small claims court judges are not able to do justice for a variety of reasons, including the volume of work.

<sup>&</sup>lt;sup>229</sup> Section 17.05 - *Withdrawal and Rescission Rights* - of the 1995 Report identified problems with the current provisions of securities legislation relating to how the withdrawal and rescission rights affect investment funds. An IFIC working group has done considerable work in this area in the intervening years but the matter has not yet received in-depth attention from securities regulators.

Without the power to enforce decisions and to compel action to be taken, the process simply adds another layer of "authority" that lacks the power to order restitution and to award damages. Restitution and damages are the remedies that most consumer/investors really want and are looking for when they lodge their complaints with the industry regulators and self-regulators.

If the ombudsman has the power to enforce decisions and to compel action to be taken, the infrastructure needed to support such powers will require the setting up of a parallel legal system. Is this appropriate or necessary? If it is, will it not rapidly run into the same problems that people currently have in obtaining easy, affordable access to justice?

I am not saying that there is no role for ombudsmen. What I am saying is that there is a need to identify clearly what the expected role is of ombudsmen.

It may be that ombudsmen do have a role to play in addressing issues such as coercive tied selling, breach of privacy rights and other arbitrary actions if there is no other body able to do this. However, I do not think ombudsmen should be used as an alternative to the legal system without a lot more thought being given to the matter.

As noted in subsection 24.4, we need to explore ways of improving easy, affordable access to justice for consumer/investors. Improving the knowledge and awareness of consumer/investors, developing a Canada-wide, cross-sector complaint handling standard for the financial services industry, and developing programs and procedures to facilitate actions proceeding through the courts (as noted above under the subheading, "Access to the Legal System") should go a long way toward meeting the needs of consumer/investors.

#### 25. OTHER INVESTMENT FUND MATTERS

#### 25.1. Other Investment Fund Matters

Several other matters relating to investment funds were raised by, or discussed with, industry participants during the course of the work in connection with this Review. They relate primarily to the need for additional prudential oversight and to the need to adopt operating standards and procedures. Substantially more time and resources are needed to deal properly with these matters. For the purposes of this Review, I can only highlight some of the areas that require attention.

### Liquidity and Leverage

Liquidity and leverage are two issues that need to be looked at, particularly in the case of investment funds that are redeemable on demand.

It should be remembered that the present restrictions on investment and practices of investment funds that are redeemable on demand were developed in a simpler investment environment, where no individual owner of these investment fund securities held a significant portion of the outstanding securities of the investment fund.

The impact on liquidity of large blocks of securities issued by an investment fund being owned by (or subject to the control and direction of) a single person or persons acting in a collective manner in the event of redemption, needs to be reviewed. Examples of such holdings include fund-of-fund arrangements, asset allocation services, holdings under segregated funds or universal life policies, holdings acquired to hedge or otherwise secure obligations under debt-linked obligations, and holdings by clients of financial intermediaries who decide that their clients should switch out of a particular investment fund or funds.

The impact on liquidity of large redemptions under systematic withdrawal plans and under registered plans (such as defined contribution pension plans and registered retirement income funds) needs to be reviewed having regard to the aging population and the anticipated increase in the redemptions from these plans as this element of the population begins to draw on its savings to meet its retirement needs.

The impact on liquidity of large redemptions pursuant to margin calls on unitholders of the investment fund or the policyholder of an insurance contract that is linked to the securities issued by the investment fund needs to be reviewed.

The potential impact of imperfect hedging strategies, whether in the "cash" market or in the "derivatives" market, needs to be reviewed. As recent experience has shown, sometimes reality does not match mathematical theory.

The potential impact of default under securities lending and repurchase agreements needs to be reviewed before investment restrictions are changed that remove or liberalize these restrictions and before further exemption orders are issued. There is a need to consider at what point securities lending activities change the nature of an investment fund and make it more comparable to a lending institution that leverages its deposit base (such as a bank, trust company or insurance company) and which is subject to prudential oversight and supervision aimed at ensuring the solvency of the institution.

The impact on an investment fund of the inherent leverage in its investment portfolio needs to be reviewed in light of the fundamental principle that an investment fund whose securities are redeemable on demand is not permitted to leverage its investment portfolio. The inherent leverage in an investment fund's portfolio adds volatility to the investment fund's return stream. An example of inherent leverage exists with securities where the purchase price is payable in installments. Other examples include the inherent leverage in investments such as zero coupon bonds, securities issued by companies that are heavy borrowers (such as Utilities) and high beta securities (such as those of brokerage firms).

Other areas that have an impact on the liquidity of an investment fund (i. e. on its ability to meet, within the settlement period, redemption requests on demand) relate to:

- (i) the valuation of securities i.e. whether the securities can be sold at their carrying cost without delay and whether such trades will settle; underlying issues related to this include how the investment portfolio is priced, where the prices come from and whether they really are market prices or simply prices supplied by sub-advisers whose objectivity in providing the prices may be questioned;
- (ii) the potential inability to settle securities transactions which can be problematical in some emerging markets and otherwise;
- (iii) the concentration of ownership of securities which can have a negative impact if the securities need to be liquidated or if there are problems that wipe out the value of the securities;
- (iv) the potential for default under securities lending arrangements or repurchase agreements; these agreements usually permit re-lending of securities; it is sometimes difficult to sort out the competing claims on such securities and to identify counterparty risk (which risk includes the exposure of one's counterparty to its counterparties);
- (v) the potential for massive switches from equity funds to fixed income funds, money market funds, and to cash or cash equivalents;
- (vi) the potential inability to borrow up to the 5% ceiling provided for in the current investment restrictions applicable to open-end investment funds in order to meet redemption requests in

the absence of stand-by lines of credit or in the event that commitments under such lines of credit are not honored

Fund governance measures, operating standards, risk management measures, independent oversight and review procedures all need to focus on these issues. They are all tools that would be expected to play a key role in prudential regulatory oversight.

## Systematic Withdrawal Plans

Another issue of a prudential nature that needs to be looked into is whether the oversight of systematic withdrawal plans is sufficient. These plans are a form of pay-out annuity. Their use and significance will increase greatly as the baby boom generation emerges from the wealth accumulation stage to the wealth distribution stage.

Currently, pay-out annuities which are not linked to life insurance, are not subject to any regulation or prudential oversight. Now is the time to be considering whether any regulation or prudential oversight is needed for these pay-out annuities and, if so, what form it should take. Issues respecting disclosure of the terms of the pay-out annuity, the unilateral ability to amend the terms of the annuity and the imposition of penalties that effectively lock-in the consumer/investor need to be addressed.

Another area that needs to be addressed is the impact that industry consolidation may have on these plans as investment funds and/or their managers merge.

## The Need to Develop Standards for Operating Procedures

There is a need to develop written industry standards for operating procedures for the operations of investment funds and for the operations of the distributors of investment funds. These standards should apply throughout Canada and on a cross-sector basis.

These standards should provide for the electronic interface of all participants in the financial services industry.

I am told that standards are being developed but they are being developed by "technology-driven people" on an ad hoc basis and largely without input from the people charged with prudential oversight and control. They are also being developed without regulatory oversight and input.

Some industry participants have observed that there is a need to bring a compliance perspective to bear on some of the systems and procedures that are being put in place. Industry participants tell me that one example of this need relates to the FundSERV system which has been built so that it permits dealers to enter payment orders in the names of alternate payees. The ability to designate alternate payees increases the risk of fraud.

Some industry participants also consider that FundSERV's system of handling switch transactions is problematic. This system apparently provides for redemption proceeds being reinvested before the redemption proceeds are actually paid out by the redeeming fund. If the redeeming fund has a liquidity problem (which could occur if the redeeming fund suspends redemptions or is unable to fund the payment of the redemption proceeds) there is a question as to who bears the added costs or the losses.

Some industry participants see this as a theoretical concern rather than a practical one. Whether the concern is theoretical or practical does not seem to be the issue. The practice seems to be an unsound

one and there is no need for it. If it is going to prevail, there is a need to expedite moving to same day settlement procedures.<sup>230</sup>

I therefore agree with those industry participants who see a need for a compliance audit to be conducted of FundSERV's standards and business processes. This can only serve to enhance the very significant contribution that FundSERV has made, and is continuing to make, to the investment fund industry and to the development of industry standards and operating procedures.

## Dealer Operating Standards and Controls

There are several areas that need to be addressed with respect to dealer operating standards and controls. Hopefully all of these will be addressed by the business practices and standards of the Single SRO. The areas that need particular attention relate to the need:

- (i) to have a "portfolio system" of record-keeping in place to keep track "on-book" of what a customer holds and that provides for and permits client name registrations;
- (ii) to have rigorous systems, controls and oversight procedures in place if nominee registrations are used; and
- (iii) to address fundamental business process/system control procedures aimed at reducing the risk of fraud; an example of this is the need for the independent preparation and distribution of account statements to clients.

#### **Error Correction Procedures**

There are no standard operating procedures for correcting errors whether made by fund managers or by dealers. Errors that occur within registered plans are particularly difficult to deal with.

I am told that there is a lot of pressure brought to bear on fund managers to agree to the backdating of orders in situations that could be prejudicial to the other unitholders in the investment fund. The willingness or unwillingness of fund managers to accede to these requests often affects whether the dealer will continue to sell the fund manager's sponsored funds. Ethical fund managers end up absorbing the costs of error corrections that they did not cause in order to keep dealers happy and not prejudice the unitholders of their funds. However, this can be costly and there is no independent monitoring of what happens.

Developing sound operating procedures for error correction that do not prejudice the unitholders of the investment fund need to be developed and universally adopted.

### Registrar and Transfer Agency Operations

Industry participants have expressed concern about the fact that registrar and transfer agency operations for investment funds are unregulated and that there are no specified standards for what is required to be done. A related factor is that the increasing use of nominee registrations has resulted in the transfer to dealers of most of the record-keeping work that was once performed by the respective registrars and transfer agents of investment funds.

.

<sup>&</sup>lt;sup>230</sup> However, the feasibility of doing this before all trades are settled on a same-day basis is questionable.

As a result, the dealer's records are often the only record of an investor's ownership interest in an investment fund. Although I understand that some dealers have invested significant resources in improving their record-keeping ability, others have not. I am told that record-keeping errors are a "real problem". This is an area of vulnerability for investors.

It is an area where there is a need for some prudential oversight and direction as well as for operating standards and independent oversight thereof.

## Self-Directed Registered Plans

Another area of vulnerability for consumer/investors relates to assets held under registered plans in the custody of unregulated, or inadequately regulated, entities. Again, this is an area where there is a need for prudential oversight and for operating standards. Personnel working in this area require a high degree of training and proficiency in custodial and other basic legal requirements relating to the transferability of property interests.

### Comparability of Performance Information

There are several issues related to the lack of comparability of performance and other information. Some of these issues have been dealt with in earlier Sections of this Review as well as in the 1995 Report. An area that has not been discussed in either place relates to the fact that marketing people are using the stated management expense ratio of an investment fund as the "sales price" of the investment fund. People are encouraged to compare investment funds on the basis of this "sales price" - the management expense ratio. There are two reasons why it is not appropriate to make this comparison.

The first relates to the fact that the management expense ratio is not a "fixed price". It fluctuates depending on future performance and other costs that are comprised in the calculation of the ratio.

The second reason that the stated management expense ratio is not a comparative measure is because there is no consistent basis for disclosing what is or is not included in the calculation of the management expense ratio. These ratios do not reflect the impact of items such as management fee waivers, management expense subsidizations or the use of directed commissions to pay operating expenses. Management fee reductions and management fee rebates for certain investors also skew the usefulness of the management expense ratio for comparison purposes.

One way to address this issue would be to require that all the subsidies, waived amounts, operating expenses that have been paid for with directed commissions, management fee reductions and management fee rebates be added back for the purpose of stating the management expense ratio.

This seems to be a "disclosure" matter and, as in all disclosure matters, there is a need to ensure that the information that is provided is not inherently misleading.

# Standardizing Terminology

Industry participants point to the need to standardize industry terminology dealing with sales commissions. They point to the need to be sure that the terminology that is used corresponds with the ordinary plain meaning that consumer/investors would expect the words to have. An example is the use of the term "no load" which should mean just that - as opposed also to being used to refer to a fund where the management fee includes a level load sales commission that is paid by the manager to the dealer.

A related issue is the use of the term "transaction fee" and the point at which a transaction fee becomes a disguised sales commission.

Words such as "primarily" and "substantially" also need to be given some generally accepted definition.

# **Business Recovery Programs**

Industry participants highlight the need for regulators to pay more attention to the adequacy of business recovery programs. They recommend that industry participants (including investment funds, intermediaries and their respective service providers) should periodically disclose what their procedures are and that audits should cover the adequacy of such programs.

The importance of these programs takes on added significance given the Year 2000 concerns and the vulnerability that results from the property interests of consumer/investors being evidenced only by book entries in the records of intermediaries.

\* \* \*

The various matters discussed in this Review have a direct or indirect effect on the ability of consumer/investors to achieve their objectives of lifetime self-sufficiency for themselves and their families. They reinforce the need not only for better regulation but also for this regulation to be coordinated, cooperative and harmonized. They reinforce the need to deal with the many issues that have been recognized for some time as needing to be dealt with. They reinforce the need to address the new challenges that are increasingly being presented.

### 26. CONCLUSION

The well-being of consumer/investors depends on governments and regulators alike rising to the challenges that face Canadians as they seek to become better decision-makers and in doing so to provide for their financial security.

This well-being is also dependent on consumer/investors rising to the challenges that they face. They have to do their part. They have to help themselves. They cannot abdicate their responsibility to act prudently and with full knowledge of the facts.

To further the well-being of consumer/investors (which include all Canadians), I have, in various places throughout this Review, suggested the need to establish a number of bodies including a Regulator, a Single SRO, certain task forces and working groups, all of which would have specific functions or tasks.

The reality is that nothing will happen unless it is made to happen by a leadership role being assumed by those at the highest levels of the federal and provincial governments and industry.

Collectively, we need to find a way to give meaning to the often overworked words of "coordination, cooperation and harmonization". This is a community problem. No one sector or regulatory group acting alone has the ability or power to do this. We need to act together. We need to remember that it is people who make things happen and that it is people who prevent things from happening. There is a positive role for everyone to play. Our well-being depends upon it.

Glorianne Stromberg Toronto, Canada October 3, 1998